### SECURITIES AND EXCHANGE COMMISSION

# **FORM 10-Q**

Quarterly report pursuant to sections 13 or 15(d)

Filing Date: **1998-08-14** | Period of Report: **1998-06-30** SEC Accession No. 0000950130-98-004044

(HTML Version on secdatabase.com)

### **FILER**

### **HEALTH MANAGEMENT ASSOCIATES INC**

CIK:792985| IRS No.: 610963645 | State of Incorp.:DE | Fiscal Year End: 0930

Type: 10-Q | Act: 34 | File No.: 001-11141 | Film No.: 98687772

SIC: 8062 General medical & surgical hospitals, nec

Mailing Address 5811 PELICAN BAY BLVD SUITE 500 NAPLES FL 33963 Business Address 5811 PELICAN BAY BLVD SUITE 500 NAPLES FL 33963 9415983131

### UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q		
[X] QUARTERLY REPORT PURSUANT TO SECTION OF THE SECURITIES EXCHANGE ACT		
For the Quarterly Period Ended Jun	ne 30, 1998	
OR		
[ ] TRANSITION REPORT PURSUANT TO SECTION OF THE SECURITIES EXCHANGE ACT		
For the transaction period from _	to	
Commission File Number 000-	18799	
HEALTH MANAGEMENT ASSOCIATES,	INC.	
(Exact name of Registrant as specified :	in its charter)	
DELAWARE	61-0963	
(State or other jurisdiction of incorporation or organization)	(I.R.S. Emplored Identification	oyer
5811 Pelican Bay Boulevard, Suite 500, Naples,	Florida	34108-2710
(Address of principal executive offices)	(Zi	p Code)
(941)598-3131		
(Registrant's telephone number, includ	ing area code)	

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months, and (2) has been subject to such filing requirements for the past 90 days.

Yes X No

At July 31, 1998, the following shares of the Registrant were outstanding:

Class A Common Stock 251,161,097 shares

of

# HEALTH MANAGEMENT ASSOCIATES, INC. FORM 10-Q FOR THE QUARTERLY PERIOD ENDED JUNE 30, 1998

### INDEX

PART	I. FINANCIAL INFORMATION		P	age
ITEM	1. FINANCIAL STATEMENTS			
	Consolidated Statements of Incom Three Months Ended June 30, 199	-		. 3
	Consolidated Statements of Incom Nine months ended June 30, 1998			. 4
	Consolidated Balance Sheets June 30, 1998 and September 30,	1997		. 5
	Consolidated Statements of Cash Nine months ended June 30, 1998			. 6
	Notes to Interim Condensed Conso	lidated Financ	ial Statements	. 7-8
ITEM	2. MANAGEMENT'S DISCUSSION AND CONDITION AND RESULTS OF OP			. 9-13
PART	II. OTHER INFORMATION			. 14
SIGN	ATURES			. 15
INDE	X TO EXHIBITS			. 16-17
		2		
	<ul><li>I. FINANCIAL INFORMATION</li><li>1. Financial Statements</li></ul>			
	HEALTH MANAGEM	ENT ASSOCIATES	, INC.	
	CONSOLIDATED S	TATEMENTS OF I AUDITED)	NCOME	
<tab:< td=""><td>LE&gt; FION&gt;</td><td></td><td></td><td></td></tab:<>	LE> FION>			
			ths ended a 30,	
		1998	1997	
<s></s>		<c></c>	<c></c>	

Net patient service revenue	\$302,150,000	\$232,691,000
Costs and expenses: Salaries and benefits Supplies and expenses Provision for doubtful accounts. Depreciation and amortization Rent expense Interest, net	103,641,000 88,759,000 24,661,000 13,080,000 7,139,000 1,242,000	78,879,000 69,672,000 17,765,000 9,347,000 5,001,000 997,000
Total costs and expenses	238,522,000	181,661,000
Income before income taxes  Provision for income taxes	63,628,000 24,973,000	51,030,000 20,029,000
Net income	\$ 38,655,000 ======	\$ 31,001,000 ======
Net income per share:  Basic  Diluted	\$ .15 \$ .15 =======	\$ .13 \$ .12 =======
Weighted average number of shares outstanding: BasicDiluted	250,871,000 257,882,000 ======	242,823,000 248,703,000 ======

See accompanying notes.

3

HEALTH MANAGEMENT ASSOCIATES, INC.

CONSOLIDATED STATEMENTS OF INCOME (UNAUDITED)

<TABLE> <CAPTION>

<S>

Net patient service revenue	\$839,642,000	\$669,645,000
Costs and expenses: Salaries and benefits Supplies and expenses Provision for doubtful accounts Depreciation and amortization Rent expense Interest, net	292,356,000 247,071,000 70,380,000 35,816,000 19,528,000 3,315,000	231,827,000 200,447,000 56,610,000 26,687,000 14,262,000 3,483,000
Total costs and expenses	668,466,000	533,316,000
<pre>Income before income taxes</pre>	171,176,000	136,329,000
Provision for income taxes	67,186,000	53,510,000
Net income	\$103,990,000 ======	\$ 82,819,000 ======
Net income per share:  Basic  Diluted	\$ .42 \$ .41 ======	\$ .33
Weighted average number of shares outstanding: BasicDiluted	248,268,000 254,865,000	240,646,000 247,362,000

  |  |See accompanying notes.

4

HEALTH MANAGEMENT ASSOCIATES, INC.

CONSOLIDATED BALANCE SHEETS

ASSETS

\_\_\_\_\_

<TABLE> <CAPTION>

June 30, September 30,
1998 1997
-----(Unaudited)
<C> <C>

<S>

Current assets: Cash and cash equivalents Receivablesnet Supplies, prepaids and other assets Funds held by trustee	\$ 15,675,000 214,002,000 33,370,000 1,782,000	\$ 67,381,000 132,896,000 21,589,000 1,225,000
Income taxes - receivable and deferred	13,039,000	13,039,000
Total current assets	277,868,000	236,130,000
Property, plant and equipment	901,062,000 173,678,000	613,752,000 141,033,000
nee proporoj, prano ana oqurpmener	727,384,000	472,719,000
Other assets: Funds held by trustee	3,949,000	944,000
Deferred charges and other assets	61,969,000	17,768,000
Total	65,918,000	18,712,000
	\$1,071,170,000	\$727 <b>,</b> 561 <b>,</b> 000
<caption></caption>		
LIABILITIES AND STOCKHOLDERS' EQUITY		
Current liabilities:		
<s></s>	<c></c>	<c></c>
Accounts payable	\$ 43,774,000 45,051,000	\$ 33,943,000 37,453,000
Accrued expenses and other liabilities  Current maturities of long-term debt	9,242,000	8,263,000
Income taxescurrently payable and deferred	34,140,000	3,221,000
income carret carrette and accorrection	01,110,000	
Total current liabilities	132,207,000	82,880,000
Deferred income taxes	19,817,000	82,880,000 18,699,000
Deferred income taxes Other long-term liabilities	19,817,000 16,600,000	82,880,000 18,699,000 16,112,000
Deferred income taxes	19,817,000	82,880,000 18,699,000
Deferred income taxes Other long-term liabilities Long-term debt Stockholders' equity: Preferred stock, \$.01 par value, 5,000,000	19,817,000 16,600,000	82,880,000 18,699,000 16,112,000
Deferred income taxes Other long-term liabilities Long-term debt  Stockholders' equity: Preferred stock, \$.01 par value, 5,000,000 shares authorized Common stock, Class A, \$.01 par value, 300,000,000 shares authorized, 251,048,000 and 244,057,000 shares issued and outstanding at	19,817,000 16,600,000	82,880,000 18,699,000 16,112,000
Deferred income taxes Other long-term liabilities Long-term debt  Stockholders' equity: Preferred stock, \$.01 par value, 5,000,000 shares authorized Common stock, Class A, \$.01 par value, 300,000,000 shares authorized, 251,048,000 and 244,057,000 shares issued and outstanding at June 30, 1998 and September 30, 1997,	19,817,000 16,600,000 150,724,000	82,880,000 18,699,000 16,112,000 49,650,000
Deferred income taxes Other long-term liabilities Long-term debt  Stockholders' equity: Preferred stock, \$.01 par value, 5,000,000 shares authorized Common stock, Class A, \$.01 par value, 300,000,000 shares authorized, 251,048,000 and 244,057,000 shares issued and outstanding at	19,817,000 16,600,000	82,880,000 18,699,000 16,112,000 49,650,000
Deferred income taxes Other long-term liabilities Long-term debt  Stockholders' equity: Preferred stock, \$.01 par value, 5,000,000 shares authorized Common stock, Class A, \$.01 par value, 300,000,000 shares authorized, 251,048,000 and 244,057,000 shares issued and outstanding at June 30, 1998 and September 30, 1997, respectively	19,817,000 16,600,000 150,724,000 -	82,880,000 18,699,000 16,112,000 49,650,000 - 2,441,000 181,966,000 375,813,000
Deferred income taxes Other long-term liabilities Long-term debt  Stockholders' equity: Preferred stock, \$.01 par value, 5,000,000 shares authorized  Common stock, Class A, \$.01 par value, 300,000,000 shares authorized, 251,048,000 and 244,057,000 shares issued and outstanding at June 30, 1998 and September 30, 1997, respectively  Additional paid-in capital	19,817,000 16,600,000 150,724,000 - 2,510,000 269,510,000 479,802,000	82,880,000 18,699,000 16,112,000 49,650,000 
Deferred income taxes. Other long-term liabilities. Long-term debt.  Stockholders' equity: Preferred stock, \$.01 par value, 5,000,000 shares authorized. Common stock, Class A, \$.01 par value, 300,000,000 shares authorized, 251,048,000 and 244,057,000 shares issued and outstanding at June 30, 1998 and September 30, 1997, respectively. Additional paid-in capital. Retained earnings.	19,817,000 16,600,000 150,724,000 	82,880,000 18,699,000 16,112,000 49,650,000 

5

### HEALTH MANAGEMENT ASSOCIATES, INC.

### CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

<TABLE> <CAPTION>

	Nine months ended June 30,		
		1997	
<\$>	<c></c>		
Cash flows from operating activities:			
Net income	\$ 103,990,000	\$ 82,819,000	
Depreciation and amortization	35,816,000	26,687,000	
(Gain) loss on sale of fixed assets	(42,000)	(69,000)	
Changes in assets and liabilities:			
Receivablesnet	(44,352,000)		
Other current assets	(4,200,000)	(705 <b>,</b> 000)	
Deferred charges and other assets	(3,988,000)		
Accounts payable	3,020,000	2,308,000	
Accrued expenses and other liabilities Income taxes	(3,109,000)	1,682,000	
currently payable and deferred	8,322,000	11,990,000	
Other long term liabilities	487,000	1,756,000	
Net cash provided by operating activities .	95,944,000	112,582,000	
Cash flows from investing activities:			
Acquisition of facilities, net of cash acquired	(168,213,000)	(51,467,000)	
Additions to property, plant and equipment	(42,025,000)	(47,227,000)	
Proceeds from sale of equipment	188,000	•	
Net cash used in investing activities	(210,050,000)	(98,396,000)	
Cash flows from financing activities:			
Proceeds from long-term borrowings	80,535,000	384,000	
Principal payments on debt	(22, 187, 000)		
Increase in funds held by trustee	(3,562,000)		
Issuance of common stock, net of costs	7,614,000		
Net cash provided by financing activities	62,400,000	, ,	

</TABLE>

See accompanying notes.

6

### HEALTH MANAGEMENT ASSOCIATES, INC. NOTES TO INTERIM CONSOLIDATED FINANCIAL STATEMENTS

### 1. Basis of Presentation

The consolidated balance sheet as of September 30, 1997 has been derived from the audited consolidated financial statements included in Health Management Associates, Inc.'s (the Company's) 1997 Annual Report. The interim consolidated financial statements at June 30, 1998 and for the three and nine month periods ended June 30, 1998 and 1997 are unaudited; however, such interim statements reflect all adjustments (consisting only of a normal recurring nature) which are, in the opinion of management, necessary for a fair presentation of the financial position and results of operations for the interim periods presented. The results of operations for the interim periods presented are not necessarily indicative of the results to be expected for the full year. The interim financial statements should be read in conjunction with the audited consolidated financial statements of the Company included in its 1997 Annual Report.

### 2. Earnings Per Share

In 1997 the Financial Accounting Standards Board issued Statement of Financial Accounting Standards No. 128, Earnings per share. Statement 128 replaced the previously reported primary and fully diluted earnings per share with basic and diluted earnings per share. Unlike primary earnings per share, basic earnings per share excludes any dilutive effects of options, warrants, and convertible securities. Diluted earnings per share is very similar to the previously reported fully diluted earnings per share. All earnings per share amounts for all periods have been presented, and where necessary restated to conform to the Statement 128 requirements.

The following table sets forth the computation of basic and diluted earnings per share (in thousands, except per share data):

<TABLE> <CAPTION>

Three months ended Nine months ended

		June 30,		June 30,		
		1998	1997	1998	1997	
<s></s>		<c></c>	<c></c>	<c></c>	<c></c>	
	Numerator:					
	Net income	\$ 38,65	5 \$ 31,001	\$103,990	\$ 82,819	
		======	= ======	======	======	
	Denominator:					
	Denominator for basic earnings					
	Per share-weighted average shares	250,87	1 242,823	248,268	240,646	
	Effect of dilutive securities-					
	employee stock options	7,01	1 5,880	6,597	6,716	
	Denominator for diluted					
	earnings per share	257 <b>,</b> 88	2 248,703	254,865	247,362	
		======	= ======	=======	=======	
	Basic earnings per share	\$ .1	5 \$ .13	\$ \$ .42	\$ .34	
		======	= ======	=======	=======	
	Diluted earnings per share	\$ .1	5 \$ .12	\$ .41	\$ .33	
		======	= =======		=======	

</TABLE>

During January 1998 the Company completed an acquisition through a merger agreement and tax-free stock exchange transaction (see Note 3). As a result, the Company issued approximately 4,950,000 shares of the Company's Class A common stock.

7

### HEALTH MANAGEMENT ASSOCIATES, INC. NOTES TO INTERIM CONSOLIDATED FINANCIAL STATEMENTS

### 3. Acquisitions

-----

In November 1997 the Company acquired a 125-bed acute care hospital for consideration totalling approximately \$20.9\$ million, including \$20.2\$ million in cash and the assumption of \$700,000\$ in debt.

In January 1998 the Company acquired certain assets of a 180-bed general acute care hospital pursuant to the terms on an asset purchase agreement. Also in January the Company acquired a two hospital, 221-bed general acute care system via a merger agreement and a tax-free stock exchange transaction. The consideration involved to complete these two transactions totalled approximately \$190.5 million, which included approximately \$80 million in Company stock, \$73.5 million in cash, and the assumption of \$37 million in debt.

In June 1998 the Company acquired a 166-bed health system, comprised of two acute care hospitals and one ambulatory surgical center, for consideration totalling approximately \$75.9 million, including \$71.9 million in cash and the assumption of \$4.0 million in debt.

The operating results of the foregoing hospitals have been included in the accompanying consolidated statements of income from the respective dates of acquisition. The following unaudited pro forma combined summary of operations of the Company for the nine months ended June 30, 1998 and 1997 gives effect to the operations of the hospitals purchased during this fiscal year as if the hospital acquisitions had occurred as of October 1, 1996 (in thousands, except per share data):

	Nine months ended June 30,			
	19	98		1997
			-	
Net patient service revenue	\$ 91	6,217	\$	842,172
Net income	\$ 10	5,081	\$	86,216
Net income per share:				
Basic	\$	.42	\$	.36
Diluted	\$	.41	\$	.35

### 4. Stock split

On June 22, 1998 the Company's Board of Directors approved a three-for-two stock split on the Company's Common Stock in the form of a 50% stock dividend to shareholders of record on June 30, 1998, which was distributed on July 17, 1998. All share and per share data in the accompanying consolidated financial statements and footnotes have been restated for all periods presented to reflect the effect of the stock split.

8

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Results of Operations
----Three months ended June 30, 1998 compared
to three months ended June 30, 1997

Net patient service revenue for the three months ended June 30, 1998 ("1998 Period") was \$302,150,000, as compared to \$232,691,000 for the three months ended June 30, 1997 ("1997 Period"). This represented an increase in net patient service revenue of \$69,459,000, or 29.9%. Hospitals in operation for the entire 1998 Period and 1997 Period ("same hospitals") provided \$21,452,000 of the increase in net patient service revenue, which resulted primarily from inpatient and outpatient volume increases. The remaining increase of \$48,007,000 included \$48,089,000 of net patient service revenue from the acquisition of the 125-bed Southwest Hospital effective November 1, 1997, the 221-bed River Oaks Health System effective January 1, 1998, the 180-bed Riley Memorial Hospital effective January 2, 1998, and the 166-bed Regional Healthcare, Inc., health system effective June 1, 1998, offset by a decrease of \$82,000 in Corporate and miscellaneous revenue.

During the 1998 Period the Company's hospitals generated total patient days of service and an occupancy rate of 152,880 and 45.5%, respectively, versus

123,326 and 44.0%, respectively, for the 1997 Period. Same hospital patient days and occupancy for the 1998 Period were 129,820 and 45.9%, respectively, versus 123,326 and 44.0%, respectively for the 1997 Period. Same hospital admissions for the Company during the 1998 Period were 26,238, up 4.6% from the 25,096 admissions during the 1997 Period.

The Company's operating expenses (salaries and benefits, supplies and expenses, provision for doubtful accounts and rent expense) for the 1998 Period were \$224,200,000 or 74.2% of net patient service revenue as compared to \$171,317,000 or 73.6% of net patient service revenue for the 1997 Period. Of the total \$52,883,000 increase, approximately \$17,079,000 related to same hospitals, which was largely attributable to the increased patient volumes. Another \$35,422,000 of increased operating expense related to the acquisitions mentioned previously. The remaining \$382,000 represented an increase in Corporate and miscellaneous other operating expenses.

The Company's earnings before depreciation and amortization, interest and income taxes (EBITDA) were \$77,950,000 for the 1998 Period as compared to \$61,374,000 for the 1997 Period, an increase of \$16,576,000 or 27.0%. The EBITDA margin was 25.8% for the 1998 Period compared to 26.4% for the 1997 Period.

The Company's depreciation and amortization costs increased by \$3,733,000 and interest expense increased by \$245,000. The increase in depreciation and amortization resulted primarily from the acquisitions mentioned previously. The increase in interest expense was due largely from acquisition related debt, as well as lower investment income in the 1998 Period (as a result of cash used for the acquisitions previously mentioned), which is netted against interest expense.

9

## Item 2. Management's discussion and Analysis of Financial Condition and Results of Operations (continued)

The Company's income before income taxes was \$63,628,000 for the 1998 Period as compared to \$51,030,000 for the 1997 Period, an increase of \$12,598,000 or 24.7%. The increase resulted primarily from same hospital volume increases and the acquisitions mentioned previously. The Company's provision for income taxes was \$24,973,000 for the 1998 Period as compared to \$20,029,000 for the 1997 Period. These provisions reflect effective income tax rates of 39.25% for both periods. As a result of the foregoing, the Company's net income was \$38,655,000 for the 1998 Period as compared to \$31,001,000 for the 1997 Period.

Results of Operations
----Nine months ended June 30, 1998 compared
to nine months ended June 30, 1997

Net patient service revenue for the nine months ended June 30, 1998 ("1998 Nine Month Period") was \$839,642,000, as compared to \$669,645,000 for the nine months ended June 30, 1997 ("1997 Nine Month Period"). This represented an increase in net patient service revenue of \$169,997,000, or 25.4%. Same hospitals provided \$62,644,000 of the increase in net patient service revenue,

which resulted primarily from inpatient and outpatient volume increases. The remaining increase of \$107,353,000 included \$108,195,000 of net patient service revenue from the acquisitions, offset by a decrease of \$842,000 of Corporate and miscellaneous revenue.

During the 1998 Nine Month Period the Company's hospitals generated 461,999 total patient days of service and an occupancy rate of 48.3%, respectively, versus 376,359 and 46.0%, respectively, for the 1997 Nine Month Period. Same hospital patient days and occupancy for the 1998 Nine Month Period were 386,501 and 49.7%, respectively, versus 359,894 and 46.4%, respectively, for the 1997 Nine Month Period. Same hospital admissions for the Company during the 1998 Nine Month Period were 76,634, up 5.1% from the 72,895 admissions during the 1997 Nine Month Period.

The Company's operating expenses for the 1998 Nine Month Period were \$629,335,000 or 75.0% of net patient service revenue as compared to \$503,146,000 or 75.1% of net patient service revenue for the 1997 Nine Month Period. Of the total \$126,189,000 increase, approximately \$42,302,000 related to same hospitals, which was largely attributable to increased patient volumes. Another \$82,475,000 of increased operating expense related to the hospital acquisitions mentioned previously. The remaining increase of \$1,412,000 represented an increase in Corporate and miscellaneous other operating expenses.

10

# Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations (Continued)

The Company's earnings before depreciation and amortization, interest and income taxes were \$210,307,000 for the 1998 Nine Month Period as compared to \$166,499,000 for the 1997 Nine Month Period, an increase of \$43,808,000 or 26.3%. The Company's EBITDA margin increased slightly to 25.0% for the 1998 Nine Month Period, as compared to 24.9% for the 1997 Nine Month Period.

The Company's depreciation and amortization costs increased by \$9,129,000 and interest expense decreased by \$168,000. The increase in depreciation and amortization resulted primarily from the acquisitions previously mentioned. The decrease in interest expense reflects higher investment income in the 1998 Nine Month Period (which is netted against interest expense), partially offset by interest expense on acquisition related debt.

The Company's income before income taxes was \$171,176,000 for the 1998 Nine Month Period as compared to \$136,329,000 for the 1997 Nine Month Period, an increase of \$34,847,000, or 25.6%. The increase resulted primarily from same hospital volume increases and the acquisitions mentioned previously. The Company's provision for income taxes was \$67,186,000 for the 1998 Nine Month Period as compared to \$53,510,000 for the 1997 Nine Month Period. These provisions reflect effective income tax rates of 39.25% for both periods. As a result of the foregoing, the Company's net income was \$103,990,000 for the 1998 Nine Month Period as compared to \$82,819,000 for the 1997 Nine Month Period.

Liquidity and Capital Resources

The Company's operating cash flows totaled \$95,944,000 for the 1998 Nine Month Period as compared to \$112,582,000 for the 1997 Nine Month Period. The positive cash flows resulted from the Company's increased profitability and management of its working capital. The Company's investing activities used \$210,050,000 and \$98,396,000 for the 1998 Nine Month Period and 1997 Nine Month Period, respectively. Acquisitions and ongoing capital expenditure requirements accounted for substantially all of the funds used in investing activities. Financing activities provided net cash of \$62,400,000 for the 1998 Nine Month Period and \$8,576,000 during the 1997 Nine Month Period. The \$53,284,000 increase resulted primarily from borrowings used to finance the previously mentioned June 1, 1998 acquisition. See the Condensed Consolidated Statements of Cash Flows for the nine months ended June 30, 1998 and 1997 at page 6 of this Report.

11

## Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations (Continued)

The Company had approximately \$25,500,000 of available cash on hand at July 31, 1998. In addition, the Company has a total of \$255 million of credit available under its two unsecured lines of credit.

The Company's credit agreements contain certain covenants which, without prior consent of the banks, limit certain activities of the Company and its subsidiaries, including those relating to merger, consolidation and the Company's ability to secure indebtedness, make guarantees, and grant security interests. The Company must also maintain minimum levels of consolidated tangible net worth, debt service coverage, and debt to cash flow and net worth.

At the present time, the Company anticipates that cash on hand, internally generated funds and funds available under its lines of credit will be sufficient to satisfy the Company's requirements for capital expenditures, future acquisitions and working capital.

Year 2000 Computer Issues

The Year 2000 Computer Issue is the result of most computer programs using two digits rather than four to identify a year in a data field. These programs were designed and developed without considering the impact of the upcoming change in the century. If not corrected, many computer applications could fail or create erroneous results.

The Company expects to be substantially complete with modifications to its existing computer software by December 31, 1998, and does not believe the Year 2000 Issue will propose material operational problems for its computers. Relating to certain computer aided medical equipment, the Company has been contacting third-party vendors regarding the compliance status of their products. However, the Company has no assurance that the systems of the Federal and State governments, other payors or other companies with which the Company's systems interface on or which they rely, will be upgraded on a timely basis.

The Company is utilizing both internal and external resources to complete the Year 2000 modifications. The majority of the costs relating to the Year 2000 Issue will be expensed as incurred. Management of the Company does not

believe such costs will have a material adverse impact to the Company's future results of operations. However, there can be no guarantee that actual results could differ materially from those anticipated. Factors that might cause material differences include, but are not limited to, the availability and cost of trained personnel and the ability to locate and correct all relevant computer coding and all medical equipment affected.

12

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations (Continued)

Forward-Looking Statements

Certain statements contained in this Report, including, without limitation, statements containing the words "believes," "anticipates," "intends," "expects" and words of similar import, constitute "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Such forwardlooking statements involve known and unknown risks, uncertainties and other factors that may cause the actual results, performance or achievements of the Company or industry results to be materially different from any future results, performance or achievements expressed or implied by such forward-looking statements. Such factors include, among others, the following: general economic and business conditions, both nationally and in the regions in which the Company operates; industry capacity; demographic changes; existing government regulations and changes in, or the failure to comply with, governmental regulations; legislative proposals for health care reform; the ability to enter into managed care provider arrangements on acceptable terms; changes in Medicare and Medicaid payment levels; liability and other claims asserted against the Company; competition; the loss of any significant ability to attract and retain qualified personnel, including physicians; the availability and terms of capital to fund additional acquisitions or replacement facilities. Given these uncertainties, prospective investors are cautioned not to place undue reliance on such forward-looking statements. The Company disclaims any obligation to update any such factors or to publicly announce the results of any revision to any of the forward-looking statements contained herein to reflect future events or developments.

13

### PART II - OTHER INFORMATION

Item 1. Legal Proceedings.

None.

Item 2. Changes in Securities.

None.

Item 3. Defaults upon Senior Securities.

None.

Item 4. Submission of Matters to a Vote of Security Holders

None

Item 5. Other Information.

\_\_\_\_\_

None.

Item 6. Exhibits and Reports on Form 8-K.

\_\_\_\_\_

a. Exhibits:

\_\_\_\_\_

See Index to Exhibits located on page 16.

b. Reports on Form 8-K:

\_\_\_\_\_

None

14

### SIGNATURES

Pursuant to the requirements of the securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned, thereunto duly authorized.

HEALTH MANAGEMENT ASSOCIATES, INC.

DATE: August 7, 1998 BY: /s/ Stephen M. Ray

-----

Stephen M. Ray

Senior Vice President-Finance (Duly authorized officer and Principal Financial Officer)

15

### INDEX TO EXHIBITS

(2) Plan of acquisition, reorganization, arrangement, liquidation or succession.

Not applicable.

- (3) (I) Articles of Incorporation
  - 3.1 The Fifth Restated Certificate of Incorporation, previously filed and included as Exhibit 3.1 to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 1995, is incorporated herein by reference.
  - (ii) By-laws

The By-laws, as amended, previously filed and included as Exhibit 3.2 to the Company's Quarterly Report on Form 10-Q for the quarter ended December 31, 1995, is incorporated herein by reference.

(4) Instruments defining the rights of security holders, including indentures. The Exhibits referenced under (3) of this Index to Exhibits are incorporated herein by reference.

Fourth Amended and Restated Credit and Reimbursement Agreement among the Company and NationsBank of Florida National Association and the Banks named therein, dated December 1, 1994, previously filed and included as Exhibit 4.12 to the Company's Annual Report on Form 10-K for the year ended September 30, 1994, is incorporated herein by reference.

Credit Agreement dated May 6, 1996 between First Union National Bank of Florida and the Company, pertaining to a \$10 million working capital and cash management line of credit, previously filed and included as Exhibit 4.3 to the Company's Annual Report on Form 10-K for the fiscal year ended September 30, 1996, is incorporated herein by reference.

Amendment Agreement No. 1 to Fourth Amended and Restated Revolving Credit and Reimbursement Agreement, made as of September 30, 1996, previously filed and included as Exhibit 4.1 to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 1997, is incorporated herein by reference.

(10) Material contracts

Definitive Agreement, dated March 12, 1998, and Amendment to Definitive Agreement, both signed between Regional Healthcare, Inc. and Health Management Associates, Inc., is included herein as Exhibit 10.1 at page 18 of this Report.

Lease Agreement, made as of June 1, 1998, between Hernando County, Florida and Hernando HMA, Inc., is included herein as Exhibit 10.2 at page 78 of this Report.

16

#### INDEX TO EXHIBITS (Continued)

(11) Statement re computation of per share earnings.

Not applicable.

(15) Letter re unaudited interim financial information. Not applicable. (18) Letter re change in accounting principles. Not applicable. (19) Report furnished to security holders. Not applicable. (22) Published report regarding matters submitted to vote of security holders. Not applicable (23) Consents of experts and counsel. Not applicable. (24) Power of attorney. Not applicable. (27) Financial Data Schedule. Financial Data Schedule is included herein as Exhibit 27.1 at page of

this Report.

(99) Additional exhibits.

Not applicable.

### DEFINITIVE AGREEMENT

by and among

REGIONAL HEALTHCARE, INC.
HERNANDO HEALTHCARE, INC.
SPRING HILL REGIONAL HOSPITAL, INC.

and

HERNANDO COUNTY, FLORIDA a political subdivision of the State of Florida ("County")

and

HERNANDO HMA, INC. HEALTH MANAGEMENT ASSOCIATES, INC.

DEFINITIVE AGREEMENT

#### TABLE OF CONTENTS

<TABLE> <CAPTION>

		PAGE	
<s></s>		<c></c>	
<5/		<c></c>	
ARTICLE	. DEFINITIONS	. 2	)
	.1 Definitions	. 2	)
	.2 Interpretation	. 6	)
ARTICLE	BASIC TRANSACTIONS	. 7	1
	.1 Lease of Leased Assets	. 7	1
	.2 Purchase of Purchased Assets	. 7	1
	.3 Assets Excluded from Transactions	. 8	3
	.4 Assumption of Certain Liabilities	. 8	3
	.5 Purchase Price; Rent	. 8	3
	.6 Post-Closing Adjustment	. 8	3
	.7 Transfer Taxes	. 9	)
	.8 Title Insurance and Survey	. 9	)
ARTICLE	. CLOSING	. 10	)
	.1 Closing	. 10	)
	.2 Actions of Sub and HMA at Closing	. 10	)
	.3 Actions of Regional at Closing	. 11	
ARTICLE	. REPRESENTATIONS AND WARRANTIES OF REGIONAL	. 12	)
	.1 Organization and Authorization	. 13	3
	.2 Assets; Title		3
	.3 Consents; Absence of Conflicts With Other		
	Agreements, Etc	. 13	3
	.4 Binding Obligations	. 14	Ĺ
	.5 Financial Statements	. 14	ŀ

	4.6 Licenses	
	4.7 Medicare Participation; Accreditation	
	4.8 Regulatory Compliance	
	4.9 Contracts	
	4.10 Equipment	
	4.11 Insurance	
	4.12 Employee Benefit Plans	
	4.13 Employee Relations	
	4.14 Litigation or Proceedings	
	4.15 Special Funds	
	4.16 Medical Staff Matters	
	4.17 Subsequent Results	
	4.18 Environmental Matters	
	4.19 Taxes	
	4.20 No Broker's Fees	
	4.21 Disclaimer	. 18
ARTICLE	5. REPRESENTATIONS AND WARRANTIES OF SUB AND HMA	
	5.1 Organization and Authorization of HMA	
	5.2 Organization and Authorization of Sub	. 19
<td>&gt;</td> <td></td>	>	
	-i-	
(M3DID)		
<table></table>		
<captio< td=""><td>N&gt;</td><td></td></captio<>	N>	
		DACE
		PAGE
<s></s>		<c></c>
<5>		<0>
	5.3 Consents; Absence of Conflicts With Other	
	Agreements, Etc	. 19
	5.4 Binding Obligations	
	5.5 Regulatory Compliance	
	5.6 Litigation or Proceedings	
	5.7 No Broker's Fees	
	NO BIORCE S ICCS	. 20
ARTICLE	6. PRE-CLOSING ACTIONS BY REGIONAL	. 20
инттопп	6.1 Transition; Access	
	6.2 Operations	
	6.3 Regulatory Approvals	
	6.4 Additional Financial Information	
	6.5 Additional Requirements	
	6.6 Closing Conditions	
	0.0 Clobing Conditions	• 22
ARTICLE	7. PRE-CLOSING ACTIONS BY SUB AND HMA	. 22
	THE CHOCKNO MOTIONS BY COD MAD MAD MAD CONTROL OF CONTR	
	7 1 Regulatory Approvals	
	7.1 Regulatory Approvals	. 22
	7.1 Regulatory Approvals	. 22
ARTICLE	7.2 Closing Conditions	. 22 . 22
ARTICLE	7.2 Closing Conditions	<ul><li>22</li><li>22</li><li>22</li></ul>
ARTICLE	7.2 Closing Conditions	. 22 . 22 . 22
ARTICLE	7.2 Closing Conditions	. 22 . 22 . 22 . 22 . 22
ARTICLE	7.2 Closing Conditions  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations	. 22 . 22 . 22 . 22 . 22 . 23
ARTICLE	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.	. 22 . 22 . 22 . 22 . 22 . 23 . 23
ARTICLE	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.	. 22 . 22 . 22 . 22 . 23 . 23 . 23
ARTICLE	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.  8.6 Survey and Leasehold Title Insurance Policy.	. 22 . 22 . 22 . 22 . 23 . 23 . 23 . 23
ARTICLE	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.  8.6 Survey and Leasehold Title Insurance Policy.  8.7 No Adverse Change.	. 22 . 22 . 22 . 22 . 23 . 23 . 23 . 23
ARTICLE	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.  8.6 Survey and Leasehold Title Insurance Policy.  8.7 No Adverse Change.  8.8 Instruments from County.	. 22 . 22 . 22 . 22 . 23 . 23 . 23 . 23
ARTICLE	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.  8.6 Survey and Leasehold Title Insurance Policy.  8.7 No Adverse Change.	. 22 . 22 . 22 . 22 . 23 . 23 . 23 . 23
	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.  8.6 Survey and Leasehold Title Insurance Policy.  8.7 No Adverse Change.  8.8 Instruments from County.  8.9 Other Instruments and Documents.	. 22 . 22 . 22 . 22 . 23 . 23 . 23 . 23
	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.  8.6 Survey and Leasehold Title Insurance Policy.  8.7 No Adverse Change.  8.8 Instruments from County.  8.9 Other Instruments and Documents.  9. CONDITIONS PRECEDENT TO REGIONAL'S OBLIGATIONS.	. 22 . 22 . 22 . 22 . 23 . 23 . 23 . 23
	7.2 Closing Conditions.  8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS.  8.1 Representations, Warranties and Covenants.  8.2 Opinion of Regional's Counsel.  8.3 Pre-Closing Confirmations.  8.4 Hart-Scott Rodino.  8.5 Actions or Proceedings.  8.6 Survey and Leasehold Title Insurance Policy.  8.7 No Adverse Change.  8.8 Instruments from County.  8.9 Other Instruments and Documents.	. 22 . 22 . 22 . 22 . 23 . 23 . 23 . 23

	9.3	Pre-Closing Confirmations	24
	9.4	Hart-Scott Rodino	24
	9.5	Actions or Proceedings	24
	9.6	Other Instruments and Documents	25
	9.7	Instruments From County	25
ARTICLE	10.	INDEMNIFICATION	25
	10.1	Indemnification by HMA	25
	10.2	Indemnification by Regional	25
	10.3	Survival	26
	10.4	Limitations	26
	10.5	Indemnification Procedure	29
ARTICLE	11.	TERMINATION	30
<td>&gt;</td> <td></td> <td></td>	>		
		-ii-	
<table></table>			
CAPTIO	N>		
			PAGE
_			
<s></s>			<c></c>
	11 1	Ontional Manmination	30
	11.1	- <u>-</u>	
	11.2	Actions on Termination; Effect of Termination	
	11.3	Termination Payments	31
ARTICLE	12	FURTHER COVENANTS	32
AITTCHE	12.1		
	12.2		
	12.3		
	12.4	-	
	12.4		
	12.6	Cooperation	
	12.7		
	12.7	special Post-Closing Covenants	34
ARTICLE	13	IN GENERAL	44
литтепь	13.1	Enforcement Expenses	44
	13.2	Notice	
	13.3	Schedules and Other Instruments	
	13.4	Choice of Law	
	13.5 13.6	Benefit Waivers and Consents	
		Severability	
	13.8	Inferences	45
	13.9	Entire Agreement	45
		0 Amendment	
		1 Counterparts	
	13.12	2 Guaranty by HMA	46

-iii-

</TABLE>

#### DEFINITIVE AGREEMENT

This Definitive Agreement (this "Agreement") is made and entered into on the 12th day of March, 1998, by and among Regional Healthcare, Inc., a Florida not-for profit corporation ("RHI"), Hernando Healthcare, Inc., a Florida not-for-profit corporation ("HHI") and Spring Hill Regional Hospital Inc. a Florida not-for-profit corporation ("Spring Hill") (hereinafter RHI, HHI and Spring Hill

13.13 Bankruptcy Court Jurisdiction.....

are collectively called "Regional"), and Hernando HMA, Inc., a Florida corporation ("Sub"), and Health Management Associates, Inc., a Delaware corporation ("HMA").

Whereas, Regional operates certain facilities located in Hernando County, Florida, consisting of two (2) general acute care hospitals licensed by the State of Florida for 166 beds known as Brooksville Regional Hospital and Spring Hill Regional Hospital (the "Hospital" or "Hospitals"), together with all businesses owned or leased by the Hospitals or by Regional in connection with the Hospitals, including PineBrook Regional Medical Center, an outpatient diagnostic and ambulatory surgery center and medical office building and all related services and offices, including all real property, whether developed or undeveloped, associated with any of the foregoing (all of which, including the Hospitals, being hereinafter sometimes referred to collectively as the "Hospital Facilities");

Whereas, certain of the assets comprising the Hospital Facilities are owned by Regional, and certain of the assets comprising the Hospital Facilities are owned by Hernando County (the "County") and leased to Regional pursuant to that certain Lease Agreement dated as of October 3, 1989, between the County and Regional (the "County Lease");

Whereas, Sub desires to purchase, and Regional desires to sell to Sub, certain assets of Regional pursuant to the terms contained in this Agreement;

Whereas, the parties desire that the County and Regional terminate the County Lease, that Regional transfer the remainder of the Hospital Facilities to the County, and that the County lease such remainder of the Hospital Facilities to Sub, and Sub desires to lease the remainder of the Hospital Facilities from the County, all as set forth in that certain Lease Agreement attached hereto as Exhibit A, which would be entered into at the Closing, as defined below, among the County, as "Lessor", Sub, as "Lessee" and HMA, as "Guarantor" (the "Lease"); and

Whereas, Sub is the wholly-owned subsidiary of HMA and HMA has agreed to all of the terms of this Agreement.

Now, therefore, for and in consideration of the premises, and the agreements, covenants, representations and warranties hereinafter set forth, and other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties agree as follows:

#### ARTICLE 1. DEFINITIONS

1.1 Definitions. In addition to the other definitions contained in the

heading paragraph and the Recitals of this Agreement and in Section 1.2, the following terms will, when used in this Agreement, have the following respective meanings:

"Affiliate" means an entity which, directly or indirectly, controls, is controlled by, or is under common control with, the referenced party.

"Assets" means, collectively, the Leased Assets and the Purchased Assets.

"Assumed Liabilities" means, collectively:

- a. all obligations of Regional or any entity comprising Regional arising out of or by reason of a transaction or event occurring, or payment or obligation accruing or arising, after the Effective Time under the Contracts;
- b. the following employee liabilities of Regional or any entity comprising Regional as of the Effective Time pertaining only to those

employees of Regional employed by Sub upon and after the Effective Time, including all employment and other taxes payable by the employer with respect to such liabilities (the "Vacation and Sick Time Liabilities"):

- (1) accrued vacation time as of the Effective Time as recorded by Regional on its books and records; and
- (2) 100% of the sick time liability, whether or not recorded by Regional on its books and records as of the Effective Time.

The Vacation and Sick Time Liabilities for the employees as of February 28, 1998, are described in "Schedule 1.1 -Vacation and Sick Time Liabilities."

- c. the accounts payable and accrued expenses of Regional as of the Effective Time including the amounts described in "Schedule 1.1 Accounts Payable and Accrued Expenses" that have not been paid as of the Closing Date in the ordinary course of business;
  - d. all of Regional's obligations and liabilities under the

-2-

Regional Healthcare, Inc. Pension Plan and Trust Agreement as amended and restated effective as of January 1, 1989; and

e. All of Regional's obligations and liabilities under Title VI of the Health Services Act of 1946, as amended from time to time, commonly known as the "Hill-Burton Act".

"Closing" has the meaning given it in Section 3.1.

"Closing Date" has the meaning given it in Section 3.1.

"Closing Statement" has the meaning given to it in Section 2.6.

"Confidential Information" has the meaning given it by Section 12.3.

"Contracts" means, collectively, those certain contracts, operating leases, capital leases, agreements and commitments of Regional or any of the entities comprising Regional that are set forth in Schedule 2.2(c), together with all other contracts, agreements and commitments added to Schedule 2.2(c) by mutual written agreement of the parties. The consents required for the assignment of specific Contracts shall be designated on Schedule 2.2(c).

"County" means Hernando County, Florida, a political subdivision of the State of Florida.

"Effective Time" has the meaning given it in Section 3.1.

"Enforceability Exceptions" has the meaning given it by Section 4.4.

"Environmental Laws" means, collectively, all federal, state and local statutes, regulations, ordinances, codes and orders (including all amendments thereto) pertaining to environmental matters (which includes air, water vapor, surface water, groundwater, soil or natural resources), including the Comprehensive Environmental Response, Compensation and Liability Act, the Medical Waste Tracking Act, the Resource Conservation and Recovery Act, the Clean Air Act, the Federal Water Pollution Control Act, the Safe Water Drinking Act and the Toxic Substance Control Act.

"Environmental Permits" means all applicable permits, licenses, registrations, orders, authorizations and approvals of any Governmental Entity required under any of the Environmental Laws.

"Estimated Purchase Price" has the meaning given to it by Section 2.5.

"Excluded Assets" means, collectively:

-3-

- a. Cash, cash deposits in banks and escrows, and all other cash equivalent items of Regional;
- b. Regional's corporate and fiscal records and other records pertaining to the operations of Regional which Regional is required by law to retain in its possession;
- c. Regional's assets not related (directly or indirectly) to the Hospitals, all to the extent specifically described on "Schedule 1.1 -Excluded Assets";
- d. the non-transferable licenses, permits, certifications and governmental approvals with respect to the Hospitals; and
- e. All refunds and reimbursements for periods prior to the Effective Time, including, but not limited to, those described in Section 12.1.

"Financial Statements" means: (a) the audited balance sheets of Regional as of September 30, 1996 and 1997, and the related audited statements of revenues and expenses, fund balances and cash flows for the years then ended, and (b) the unaudited balance sheet of Regional as of December 31, 1997, and the related unaudited statements of revenues and expenses, fund balances and cash flows for the three months then ended.

"First Party" has the respective meanings given it by Sections 11.3(a) and 12.3.

"GAAP" means generally accepted accounting principles.

"Good Title" means, in the case of personal property, good and assignable title, and, in the case of real property, title as shown by the Title Commitment, subject to the matters described in this Agreement, including Permitted Encumbrances.

"Governmental Entity" means any federal, state or local court, legislative body, governmental body, municipality, political subdivision, department, agency or authority.

"Hazardous Substances" means and includes: (a) any hazardous materials, hazardous wastes, hazardous substances, and toxic substances as those or similar terms are defined under any Environmental Law; (b) Medical Waste; (c) any asbestos or any material that contains any hydrated mineral silicate, including chrysolite, amosite, crocidolite, tremolite, anthophylite and/or actinolite, whether friable or non-friable; (d) any polychlorinated biphenyls or polychlorinated biphenyl-containing materials or fluids; (e) radon; (f) any other hazardous, radioactive, toxic or noxious substance, material, pollutant or solid, liquid or gaseous waste; (g) any pollutant or contaminant (including petroleum, petroleum hydrocarbons, petroleum products, crude oil or any

-4-

fractions thereof, natural gas or synthetic gas) that in its condition, concentration or area of release could have a significant effect on human health, the environment or natural resources; and (h) any substance that whether by its nature or its use, is subject to regulation under any Environmental Laws or with respect to which any Environmental Laws or Governmental Entity requires environmental investigation, monitoring or remediation.

"Indemnified Party" and "Indemnifying Party" have the respective meanings given them by Section 10.5(a).

"Leased Assets" means, collectively: (a) the Premises; and (b) all rights, privileges and interests appurtenant to the foregoing; provided, however, that the Leased Assets do not include any of the Purchased Assets or any of the Excluded Assets. The Leased Assets include but are not limited to the real property described in "Schedule 1.1. - Leased Assets."

"Leased Premises" has the meaning given it under the Lease.

"Licenses and Permits" means, collectively, the State of Florida Agency for Health Care Administration license, and other licenses, permits, certificates, registrations, consents, orders, authorizations and approvals of any Governmental Entity, including the County and the State of Florida, with respect to the Hospital Facilities or the Leased Assets.

"Losses" means any and all liabilities, claims, demands, suits, causes of action, judgments, obligations, damages and deficiencies, and all costs and expenses incident thereto (including reasonable attorneys' fees and all other reasonable expenses incurred in investigating and preparing or defending any litigation or proceeding, commenced or threatened, whether such litigation or proceeding is prosecuted by a party hereto or by a third party).

"Medical Waste" means any solid waste generated in the diagnosis, treatment or immunizations of human beings or animals or in research pertaining thereto. Medical Waste includes special waste from health care facilities or providers which if improperly treated or handled may serve to transmit infectious diseases and which is composed of animal waste, bulk blood and blood products, microbiological waste, pathological waste or sharps, but does not include garbage or refuse from offices, kitchens or other non-health care activities.

"Other Party" has the meaning given it by Section 12.3.

"Permitted Encumbrances" means: (a) the Lease; (b) encumbrances set forth on "Schedule 1.1 - Permitted Encumbrances"; (c) liens for taxes, not yet delinquent, for years subsequent to 1997; (d) the effect of any zoning ordinances currently enacted and affecting the Premises; (e) matters shown on the Surveys, and (f) such other liens

-5-

or encumbrances as HMA or Sub may expressly permit in writing subsequent to the Closing Date; provided, however, that liens or encumbrances arising out of any capital debt or other long-term liabilities of Regional or County (other than liens and encumbrances set forth on "Schedule 1.1 - Permitted Encumbrances") are not Permitted Encumbrances.

"Purchase Price" has the meaning given it by Section 2.5.

"Purchased Assets" has the meaning given it by Section 2.2.

"Release" or "Released" when used in reference to Hazardous Substances has the meaning given by 42 U.S.C, (S) 9601(22).

"Second Party" has the meaning given it by Section 11.3(a).

"Taxes" means, collectively, federal, state and local income, payroll, withholding, excise, sales, use, real and personal property, use and occupancy, business and occupation, mercantile, real estate, capital stock and franchise or other taxes, including penalties and interest thereon and estimated taxes.

"Survey" means the following ALTA surveys of the Premises prepared by Heidt

& Associates, Inc.: Drawing No. 222219-1 dated February 11, 1998, Drawing No. 342218-1 dated February 23, 1998, Drawing No. 342318-1 dated February 23, 1998, and Drawing No. 182220-1 dated February 23, 1998, and certified in accordance with generally accepted professional standards describing the boundaries, improvements, setbacks and all matters of record set forth in the Title Commitment.

"Title Commitment" means the Leasehold Title Insurance Commitment dated February 9, 1998, issued by Lawyer's Title Insurance Corporation. A copy of the Title Commitment is attached as "Schedule 1.1 - Title Commitment".

1.2 Interpretation. In this Agreement, unless the context otherwise

requires: (a) references to "Articles" and "Sections" are to the Articles or Sections of this Agreement and references to "Schedules" are to the Schedules annexed hereto; (b) references to any party to this Agreement include references to its respective successors and permitted assigns; (c) references to judgment include any order, writ, injunction, decree, determination or award of any court or tribunal; (d) references to a person or entity include references to any individual, company, body corporate, association, partnership, firm, joint venture, trust or Governmental Entity; (e) any of the terms defined herein may, unless the context requires otherwise, be used in the singular or the plural depending on the reference; (f) the masculine pronoun includes the feminine and the neuter, and vice versa, as appropriate in the context; (g) with respect to any matter or thing, the terms "including" or "include" mean including but not limited to such matter of thing; and (h) with

-6-

respect to any party, the term "knowledge" or "best of knowledge" of such party means the actual awareness of any executive officer of such party. The divisions of this Agreement into articles, sections and subsections and the use of captions and headings in connection therewith are solely for convenience and have no legal effect in construing the provisions of this Agreement.

#### ARTICLE 2. BASIC TRANSACTIONS

2.1 Lease of Leased Assets. At Closing, conditioned upon the requisite

actions being taken by the County, the County and Regional will terminate the County Lease, Regional will transfer and convey to the County the portion of the Leased Assets owned by Regional, and the County, Sub and HMA will execute and deliver the Lease.

2.2 Purchase of Purchased Assets. At Closing, Regional will sell, assign

and transfer to Sub all of Regional's right, title and interest in and to the following except as otherwise provided in the last sentence of this section (collectively, the "Purchased Assets"): (a) all furniture, furnishings, equipment, machinery, data processing hardware and software, vehicles and other tangible personal property owned on the Closing Date by any entity comprising Regional contained on the Premises or used in connection therewith or with the operation of the Hospital Facilities, including, but not limited to, the items described in Schedule 4.10, and all other rights, privileges and franchises owned or held on the Closing Date by any entity comprising Regional relating to the operation or development of, or associated with, the Hospital Facilities; (b) all businesses owned by Regional or any entity comprising Regional with respect to the Hospitals, including those businesses comprising the Hospital Facilities; (c) the Contracts, as described in Schedule 2.2(c); (d) Regional's accounts receivable on the Closing Date (including those related to Medicare and Medicaid) (the "Accounts Receivable"); (e) Regional's inventories of goods and supplies on the Closing Date used or maintained in connection with or located in the Hospital Facilities, including, food, cleaning materials, disposables, linens, consumables, office supplies, pharmaceuticals, drugs and medical

supplies, on the Closing Date (the "Inventories"); (f) to the extent transferable, all licenses, permits, registrations, certificates, consents, accreditations, approvals and franchises held on the Closing Date by any entity comprising Regional necessary to operate and conduct business at the Hospital Facilities; (g) all goodwill associated with the Hospitals and Hospital Facilities and all assignable warranties (expressed or implied) and assignable rights and claims assertable by Regional related to the operation of the Hospital Facilities; (h) all patents and patent applications associated with the Hospitals or the Hospital Facilities; (i) all trademarks or trade names and copyrights of Regional, and any proprietary manuals; and (j) Regional's prepaid expenses and deposits on the Closing Date (collectively, the "Prepaid Items and Deposits") as described on Schedule 2.2(j); provided, however, that the Purchased Assets do not include any of the Leased Assets or any of the Excluded Assets.

-7-

2.3 Assets Excluded from Transactions. The Excluded Assets, including the

refunds, reimbursements and obligations associated with Medicare and Medicaid and other third party payor programs for periods prior to the Closing Date including those described in Section 12.1, are not intended by the parties to be transferred to Sub or HMA, whether by lease or by purchase, at Closing and are excluded from the transactions contemplated by this Agreement and by the Lease.

2.4 Assumption of Certain Liabilities. As of the Effective Time, Sub will

assume all of the Assumed Liabilities and Sub and HMA, jointly and severally, will indemnify and hold Regional harmless with respect to the Assumed Liabilities. It is expressly understood and agreed that except for the Assumed Liabilities, neither Sub nor HMA will assume, nor will either of them be liable for, any liability, obligation, claim against or contract of Regional (or any of their Affiliates) or the County arising out of Regional's ownership or operation of the Hospital Facilities or Sub's lease and operation of the Hospital Facilities. As of the Closing Date, Sub shall become the plan sponsor of Regional's defined benefit pension plan and trust replacing Regional and, thereafter, Sub shall have all of the rights, duties and obligations of the plan sponsor of such pension plan.

2.5 Purchase Price; Rent. As consideration for the Leased Assets and

Purchased Assets, Sub will pay to Regional (a) a purchase price for the Purchased Assets in the amount of Fifty-eight Million and 00/100 Dollars (\$58,000,000.00) minus (i) an amount equal to fifty percent (50%) of the accrued sick leave, (ii) one hundred percent (100%) of the accrued vacation and holiday pay of Regional's employees that are hired by Sub as of the Closing Date, and (iii) one hundred percent (100%) of all of the assumed accounts payable and accrued expenses of Regional as of the Effective Time that exceed the sum of One Million One Hundred Thousand and 00/100 Dollars (\$1,100,000.00) (the "Purchase Price"), and (b) as agent for the County rent for the Leased Assets in the amount of Twenty Million and 00/100 Dollars (\$20,000,000.00) (the "Rent"). On or prior to the Closing Date, Regional will deliver to Sub Regional's estimate of the Purchase Price including, but not limited to, the items described in Section 2.5(a)(i), (ii) and (iii) in accordance with Regional's methods of determining such items (the "Estimated Purchase Price"). At the Closing, Sub will pay to Regional, and HMA will cause Sub to pay to Regional, by wire transfer of immediately available funds, the Estimated Purchase Price and the Rent. The Estimated Purchase Price will be subject to a post-Closing adjustment as provided by Section 2.6.

2.6 Post-Closing Adjustment.

Within sixty (60) days after the Closing Date, Regional will give to Sub a

(iii) in accordance with Regional's methods of determining such items. Sub will have the right to review the Closing Statement. Regional will permit Sub and its independent accountants access to review the working papers and schedules of Regional utilized or prepared in connection with the preparation of the Closing Statement. Within thirty (30) days after its receipt of the Closing Statement, Sub will, by written notice to Regional, either accept the Closing Statement or describe in reasonable detail any proposed adjustments to the Closing Statement as it pertains solely to the items described in Section 2.5(a)(i), (ii) and (iii) and the reasons therefore. If Regional does not receive such notice of proposed adjustments within the thirty (30) day period, then Sub will be deemed irrevocably to have accepted the Closing Statement. Regional and Sub will negotiate in good faith to resolve any disputes over any proposed adjustments to the Closing Statement. However, if any such dispute is not resolved within thirty (30) days following receipt by Regional of the proposed adjustments, Sub and Regional jointly will select an independent public accounting firm, which will determine the amounts of the items described in Section 2.5(a)(i), (ii) and (iii) in accordance with Regional's methods consistently applied in determining such items, which determination of the accounting firm will be final and binding on the parties. The fees and expenses of such accounting firm will be shared by Regional and Sub in proportion to the relative amounts of the dispute and amount determined by the accounting firm for the account of Regional and Sub, respectively. Subsequent to resolution an adjusted closing statement (the "Adjusted Closing Statement") will be signed by all parties and delivered in substitution for the Closing Statement and, if cash adjustments are to be made, then payment shall be made to the appropriate party by wire transfer of immediately available funds, within fifteen (15) days after the earliest of (a) Sub's actual or deemed acceptance of the Closing Statement, (b) the agreement of Regional and Sub in writing as to the Closing Statement or any adjustments thereto, and (c) the determination of the accounting firm described in this Section 2.6.

2.7 Transfer Taxes. All sales, transfer, documentary stamp and other

excise taxes on the transfer of the Purchased Assets to Sub, and all sales, rental, documentary stamp tax or similar taxes with respect to the Lease or on the payments under the Lease, shall be the sole obligation of Sub and HMA and Sub and HMA, jointly and severally, shall indemnify and hold Regional and the County harmless with respect thereto. All sales, transfer, documentary stamp and other taxes due on the transfer of the portion of the Leased Assets owned by Regional from Regional to the County shall be the sole obligation of Regional and Regional hereby agrees to indemnify and hold Sub and HMA harmless with respect thereto.

2.8 Title Insurance and Survey. The title insurance premium for the

Leasehold Title Insurance Policy to be issued pursuant to the terms of the Title Commitment obtained by Sub shall be paid by Sub. The cost of the Survey obtained by Regional shall be paid by Regional.

-9-

#### ARTICLE 3. CLOSING

3.1 Closing. Subject to the satisfaction or waiver by the appropriate ----party of all the conditions precedent to Closing specified in Articles 8 and 9, the closing of the transactions under this Agreement (the "Closing") will take

place at the offices of Annis, Mitchell, Cockey, Edwards & Roehn, P.A., Suite

2100, One Tampa City Center, Tampa, Florida 33601 (or such other place as the parties may mutually agree) on or before forty-five (45) days from the later of the date of signing of this Agreement by all parties or the adoption by the Hernando County Board of Commissioners of a resolution approving and authorizing the signing and delivery of the Lease to Sub, but in any event on or before May 15, 1998, except as otherwise provided below (the "Closing Date"). In the event that all of the conditions precedent under this Agreement have been satisfied or waived in writing, except that Hernando County has not approved and authorized the signing of the Lease to Sub and HMA on or prior to May 15, 1998, and Hernando County has not, on or prior to May 15, 1998, expressly disapproved Sub and HMA, the Closing Date shall be automatically extended in fifteen (15) day increments up until June 29, 1998. If Hernando County has expressly disapproved Sub and HMA then Sub and HMA shall have until May 15, 1998, to resolve the matter with Hernando County with no right on the part of HMA and Sub to extend the Closing Date beyond May 15, 1998, unless the extension of the Closing Date is granted by Regional as set forth below. In the event that any of the conditions precedent to closing set forth in this Agreement have not been satisfied or waived in writing, on or prior to May 15, 1998 then, at the sole discretion of Regional, the Closing Date may be extended in fifteen (15) day increments until such conditions are satisfied or waived in writing or the expiration of any fifteen (15) day extension occurs without an additional extension being granted by Regional. The parties agree that the transfer, sale and lease of the Leased Assets and the Purchased Assets shall be calculated and made effective as of 12:01 a.m., Eastern Time, on the Closing Date (the "Effective Time").

- (a) the Lease, for delivery to the County, duly executed by Sub and by HMA;
- (b) the Estimated Purchase Price and the Rent, delivered by wire transfer of immediately available funds;
- (c) copies of resolutions duly adopted by the Boards of Directors of Sub and of HMA, authorizing and approving their respective performance of the transactions contemplated hereby and by the Lease, and their respective execution and

-10-

delivery of this Agreement, the Lease and the other documents described herein, each certified as true, complete and in full force and effect as of the Closing Date by appropriate officers of Sub and HMA, respectively;

- (d) certificates of a duly authorized Vice President of Sub and HMA certifying that each covenant and agreement of Sub and HMA to be performed prior to or as of Closing pursuant to this Agreement has been performed in all material respects;
- (e) certificates of incumbency of the respective officers of Sub and HMA executing this Agreement, the Lease and the other documents described herein, each dated as of the Closing Date;
- (f) a certificate of existence and good standing of HMA from the State of Delaware, and a certificate of existence and active status of Sub from the State of Florida, each dated within 10 calendar days prior to the Closing Date;
- (g) the opinion of Sub's and HMA's counsel described in Section 9.2;

- (h) a certificate of a duly authorized Vice President of Sub and HMA certifying that the representations and warranties of Sub and HMA set forth in this Agreement are true and correct in all material respects as of the Closing Date; and
- (i) such other instruments and documents as are customary for the transactions contemplated by this Agreement and as Regional reasonably deems necessary to effect the terms contemplated hereby and by the Lease.
- - (a) exclusive possession of all of the Purchased Assets other than equipment leased from third parties;
  - (b) documents of transfer and assignment, duly executed by Regional, assigning to Sub Good Title to the Purchased Assets subject only to Permitted Encumbrances;
  - (c) copies of resolutions duly adopted by the Board of Directors of each of the corporations comprising Regional, authorizing and approving

-11-

Regional's performance of the transactions contemplated hereby and its execution and delivery of this Agreement and the other documents described herein, certified as true, complete and in full force and effect as of the Closing Date by an appropriate officer of Regional;

- (d) certificates of the appropriate officers of each of the corporations comprising Regional certifying that each covenant and agreement of Regional to be performed prior to or as of Closing pursuant to this Agreement has been performed in all material respects;
- (e) a certificate of incumbency of the officers of each of the corporations comprising Regional executing this Agreement and the other documents described herein, dated as of the Closing Date;
- (f) the opinion of Regional's counsel described in Section 8.2;
- (g) a certificate of the appropriate officer of each of the corporations comprising Regional certifying that the representations and warranties of Regional set forth in this Agreement are true and correct in all material respects as of the Closing Date;
- (h) the instruments required by the title company as part of the Title Commitment to be delivered by Regional, which items are set forth in Schedule B, Section 1, items 1, 2, and 3(c)-(k) of the Title Commitment and, to the extent acceptable to Regional, such other instruments as are sufficient to delete the standard exceptions set forth in Schedule B -Section 2 of the Title Commitment;
- (i) certified resolution of the County Commission authorizing the appropriate representative of the County to sign the Lease and all other documents required by the County to effect the

(j) such other instruments and documents as are customary for the transactions contemplated by this Agreement and as HMA reasonably deems necessary to effect the terms contemplated hereby and by the Lease.

#### ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF REGIONAL

Regional (solely with respect to the representations and

-12-

warranties pertaining to Regional or any corporation comprising Regional), represents and warrants to Sub and to HMA the following:

4.1 Organization and Authorization. Each of the corporations comprising

Regional is a not-for-profit corporation created under the laws of the State of Florida. Each of such corporations has the requisite corporate power and authority to enter into this Agreement, perform its obligations hereunder, and conduct its business as now being conducted.

### 4.2 Assets; Title.

-----

- (a) Except for any loss, destruction or damage caused by Sub or HMA or any of their respective Affiliates, officers, agents, employees or contractors, Regional will bear all risk of loss, destruction or damage to any of the Assets occurring prior to Closing, whether due to fire, accident or other casualty, willful act, condemnation, riot, act of God or otherwise.
- (b) Regional has no knowledge of any matters encumbering title to the Assets owned by Regional except the matters covered by the Title Commitment. Regional has Good Title to all of the Purchased Assets (other than leased equipment or real estate) owned by it and as of the Closing Date Regional will have Good Title to all of the Purchased Assets owned by it subject only to the Permitted Encumbrances.
- (c) At Closing, Regional will transfer to County Good Title to all of the Leased Assets owned by Regional subject only to the Permitted Encumbrances.
- (d) To the knowledge of Regional, there are no material violations by Regional of zoning laws or municipal ordinances regarding zoning matters affecting the Premises.
- 4.3 Consents; Absence of Conflicts With Other Agreements, Etc. Regional's

execution, delivery and performance of this Agreement and the other documents contemplated hereby to which Regional is a party, and the consummation of the transactions contemplated hereby: (a) are within the corporate power of each of the corporations comprising Regional, are not in contravention of law or of the terms of any governing instruments of the corporations comprising Regional and have been duly authorized by all appropriate corporate action; subject to the (i) signing of the consent by the County in the form appended to the Lease, (ii) signing by the United States Bankruptcy Court for the Middle District of Florida, Tampa Division ("Bankruptcy Court"), of an order ("Bankruptcy Order"), approving

the transactions under this Agreement and, containing, to the extent permitted by the Bankruptcy Court, the elements described in "Schedule 4.3 -Bankruptcy Order", (iii) expiration or termination of the waiting period under the Hart-Scott-Rodino Act, (iv) delivery of the consents described in Schedule 2.2(c), and (v) issuance of any required licenses, permits and certificates of need, it being understood, however, that except for subsections 4.3(a)(i)-(v) no approval or consent of, or filing with, any Governmental Entity is required in order for Regional to perform its obligations under this Agreement; and (b) provided that all required licenses, permits and certificates of need and third party consents are obtained as described above, will neither conflict with nor result in any material breach or contravention of, nor permit the acceleration of the maturity of a material portion of, the Assumed Liabilities, nor the creation of any lien, charge or encumbrance affecting any of the Assets.

4.4 Binding Obligations. This Agreement constitutes the valid and legally

binding obligation of Regional and is enforceable against Regional in accordance with the terms hereof, except as enforceability may be restricted, limited or delayed by applicable bankruptcy or other laws affecting creditors' rights generally and except as enforceability may be subject to general principles of equity (collectively, the "Enforceability Exceptions").

4.5 Financial Statements. Regional has heretofore delivered to HMA, and

there are annexed hereto as Schedule 4.5, true and complete copies of the Financial Statements. The Financial Statements have been prepared in accordance with GAAP, applied throughout the periods indicated on a consistent basis (except as noted in the notes thereto), and except as described in Schedule 4.5. Except as described in Schedule 4.5. the balance sheets comprising the Financial Statements present fairly in all material respects the financial condition of Regional or the Hospital Facilities, as the case may be, at the dates indicated thereon, and the statements of revenue and expenses comprising the Financial Statements present fairly in all material respects the results of operations of Regional or the Hospital Facilities, as the case may be, for the periods indicated thereon.

- 4.6 Licenses. The Hospitals are duly licensed by the State of Florida as
  -----a 91 bed (in the case of Brooksville Regional Hospital) and 75 bed (in the case
  of Spring Hill Regional Hospital) general acute care hospital. The ancillary
  departments and other operations thereof (included in the Hospital Facilities)
  that are required to be specifically licensed are duly licensed by the
- departments and other operations thereof (included in the Hospital Facilities) that are required to be specifically licensed are duly licensed by the appropriate state agencies. A list of all Licenses and Permits held by Regional is attached as Schedule 4.6.
  - 4.7 Medicare Participation; Accreditation. The Hospital is qualified for

participation in the Medicare and Medicaid programs by Healthcare Financing Administration ("HCFA"), has a current and valid provider agreement with the Medicare and Medicaid programs, and is currently accredited by the Joint Commission on Accreditation of Healthcare Organizations ("JCAHO"), except as set forth in

-14-

Schedule 4.7, and is in compliance in all material respects with the conditions of participation in such programs.

4.8 Regulatory Compliance. Except as set forth in Schedule 4.8, the

Hospital Facilities are in compliance in all material respects with all applicable statutes, rules, regulations and requirements of all Governmental Entities having jurisdiction over the Hospital Facilities and their operations,

including the Florida Agency For Health Care Administration, and Regional has timely filed all reports, data and other information required to be filed with such Governmental Entities where a failure to file timely would have a material adverse effect on the Hospital Facilities. Except as set forth in Schedule 4.8, since January 1, 1994, neither Regional nor the Hospital has received written notice of a violation of any applicable ordinance or other law, order, regulation or requirement, or notice of condemnation, lien, assessment relating to any part of the Assets or the operation of the Hospital Facilities.

4.9 Contracts. Regional has made available to HMA true and complete

copies of each of the Contracts. Each of the Contracts constitutes the valid and legally binding obligation of Regional and is enforceable against Regional in accordance with its terms except as enforcement may be limited by any or all of the Enforceability Exceptions. To the best of Regional's knowledge, each of the Contracts constitutes the valid and legally binding obligation of the other party thereto and is enforceable against such party in accordance with its terms except as enforcement may be limited by any or all of the Enforceability Exceptions. Each of the Contracts constitutes the entire agreement between the respective parties thereto relating to the subject matter thereof. Except as provided in Schedule 4.9, to the best of Regional's knowledge, all obligations required to have been performed under the terms of the Contracts have been performed in all material respects and no act or omission has occurred or failed to occur which, with the giving of notice, the lapse of time or both would constitute a default under any of the Contracts, and each of the Contracts is in full force and effect without default on the part of any party thereto. Where a consent to the assignment of any Contract is required as set forth in Schedule 2.2(c), Regional will use its reasonable efforts to obtain such consent prior to Closing.

- 4.10 Equipment. A depreciation schedule listing, as of December 31, 1997,
  ----the furniture, fixtures and equipment of Regional is attached as Schedule 4.10.
  - 4.11 Insurance. Annexed hereto as Schedule 4.11 is a list of the insurance

policies covering the ownership and operations of the Hospital Facilities and the Assets. All of such policies are now and will be until Closing in full force and effect (unless replaced with comparable coverages) with no premium arrearages.

4.12 Employee Benefit Plans. Except as set forth in Schedule 4.12,

Regional has no pension, profit sharing, compensation, deferred compensation or other employee pension, health or welfare benefit plan or arrangement relating to any one or more employees

-15-

at, or the operations of, the Hospital Facilities.

4.13 Employee Relations. Except as set forth in Schedule 4.13: (a) there

is no pending or, to the best of Regional's knowledge, threatened employee strike, union representation effort, work stoppage or labor dispute; (b) no collective bargaining agreement exists or is currently being negotiated by Regional; no demand has been made to Regional for recognition by a labor organization by or with respect to any employees at the Hospital Facilities; and (c) there are no pending or, to the best of Regional's knowledge, threatened unfair labor practice claims, equal employment opportunity claims, wage and hour claims, unemployment compensation claims, workers' compensation claims or the like with respect to the Hospital Facilities.

4.14 Litigation or Proceedings. Except to the extent set forth in

-----

Schedule 4.14, to the best of Regional's knowledge, as of the date of this Agreement, there are no material claims, actions, suits, proceedings or investigations pending or threatened against or affecting the Hospital Facilities or Regional at law or in equity, or before or by any Governmental Entity. Without limiting the generality of the foregoing, to the best of Regional's knowledge, as of the date of this Agreement, no action or proceeding before any Governmental Entity has been instituted against Regional or threatened against Regional which seeks to restrain or prohibit the transactions contemplated hereby or by the Lease. For the purpose of this Section 4.14, the term "material" shall mean any matter having a potential claim in excess of One Hundred Thousand Dollars (\$100,000.00).

4.15 Special Funds. Except as set forth in Schedule 4.15, none of the

Assets is subject to any restrictions as a result of amounts received by Regional for the purchase or improvement of the Assets or any part thereof under restricted or conditioned grants or donations.

4.16 Medical Staff Matters. Regional has heretofore provided to HMA true,

correct and complete copies of the bylaws and rules and regulations (if any) of the medical staffs of the Hospitals. Except as set forth in Schedule 4.16, there are no pending or, to the best of Regional's knowledge, threatened disputes with applicants, staff members or health professional affiliates, and all appeal periods in respect of any medical staff member or applicant against whom an adverse action has been taken have expired.

4.17 Subsequent Results. Since December 31, 1997, and except as disclosed

in the Financial Statements or any other Schedule, there has not been: (a) any material adverse change, as defined in Section 8.7; (b) any material damage, destruction or loss (not covered by insurance) adversely affecting the Hospital Facilities; or (c) any sale, assignment, transfer or disposition of any item of plant,

-16-

property or equipment of the Hospital Facilities having a net book value in excess of \$100,000 (other than supplies), except in the ordinary course of its business or where replaced by a comparable or better item.

4.18 Environmental Matters. Regional acknowledges that HMA has conducted a

Phase I environmental assessment of the Premises (the "Environmental Assessment") and a copy of the Environmental Assessment has been provided to Regional and its counsel. Except as disclosed in Schedule 4.18 or the Environmental Assessment, to the best of Regional's knowledge, the Hospital Facilities are not in material violation of any Environmental Laws. Except as disclosed in Schedule 4.18 or the Environmental Assessment, to the best of Regional's knowledge, (a) none of the Hospital Facilities has Released any Hazardous Substances in a manner that has materially violated any Environmental Laws, and (b) there has been no such release by any previous owner or operator of the Premises. Except as disclosed in Schedule 4.18 or the Environmental Assessment, to the best of Regional's knowledge, none of the Hospital Facilities has (i) any underground storage tanks, as defined in 42 U.S.C. (S) 6991(1)(A)(I), whether empty, filled or partially filled with any substance, or (ii) any asbestos or any material that contains any hydrated mineral silicate, including chrysolite, amosite, crocidolite, tremolite, anthophylite and/or actitiolite, whether friable or non-friable. Except as disclosed in Schedule 4.18, Regional has not received since March 1, 1993, any written request for information, notice or order alleging that it may be a potentially responsible party under any Environmental Laws for the investigation or remediation of a Release or threatened Release of Hazardous Substances. Except as disclosed in Schedule

4.18, to the best of Regional's knowledge, there is no lien, notice, litigation or threat of litigation relating to an alleged unauthorized Release of any Hazardous Substance on, about or beneath the Premises (or any portion thereof), or the migration of any Hazardous Substance to or from property adjoining or in the vicinity of the Premises, or alleging any obligation under Environmental Laws. Prior to the Closing, Regional will promptly notify HMA should Regional receive any such request for information, notice or order, or become aware of any lien, notice, litigation or threat of litigation relating to an alleged unauthorized Release of any Hazardous Substance on, about or beneath the Premises (or any portion thereof) or any other environmental contamination or liability with respect to the Premises (or any portion thereof). The Hospital or Regional holds all Environmental Permits required in connection with the use by Regional of the Premises or the operation of the Hospital Facilities and, to the extent permitted by law, Regional will effectively transfer to Sub all such Environmental Permits, all of which to the best of Regional's knowledge, are in good standing and are not subject to meritorious challenge.

4.19 Taxes. Except as set forth in Schedule 4.19, Regional has: (a) timely

filed or extended all returns required to be filed by or in respect of the Hospital Facilities with respect to all Taxes; (b) paid all Taxes shown to have become due pursuant to such

-17-

returns; and (c) paid all other Taxes for which a notice of assessment or demand for payment has been received, if the due date therefor has occurred. Except as set forth in Schedule 4.19, Regional has not received any notice of any proposed assessments of Taxes against or in respect of the Hospital Facilities, or of any proposed adjustments to any tax returns filed by or in respect of the Hospital Facilities.

4.20 No Broker's Fees. Regional has not employed any investment banker,

broker, finder, agent or other intermediary in connection with the negotiation or consummation of this Agreement or the Lease or of any of the transactions contemplated hereby or thereby as to which Sub or HMA may have any liability for such investment banker's, broker's, finder's, agent's or other intermediary's financial advisory or similar fee.

4.21 Disclaimer. Regional shall not be deemed to have made to Sub or HMA

any representation or warranty other than as expressly made by Regional in Sections 4.1 through 4.20 hereof. Without limiting the generality of the foregoing, and notwithstanding any otherwise express representations and warranties made by Regional in Sections 4.1 through 4.20 hereof, Regional has made no representation or warranty to Sub or HMA with respect to:

- (a) any projections, estimates or budgets heretofore delivered to or made available to Sub or HMA of future revenues, expenses or expenditures, or future results of operations;
- (b) any other information or documents made available to Sub or HMA or their counsel, accountants, or advisors, except as expressly covered by a representation and warranty contained in Sections 4.1 through 4.20 hereof;
- (c) whether any contract or lease to which any entity comprising Regional is a party or by which any assets owned or used by it are bound complies with laws, rules, regulations, safe harbors, governmental alerts or pronouncements with respect to Medicare or Medicaid, fraud and abuse or similar laws (whether federal or state) or the so-called Stark legislation; or
- (d) as to the state of the status of title to real property, Sub and/or HMA and Regional have relied on the Title Commitment.

Sub and HMA, jointly and severally, represent and warrant to Regional the following:

5.1 Organization and Authorization of HMA. HMA is a corporation duly

organized, validly existing and in good standing under the laws of the State of Delaware. HMA has the requisite corporate power and authority to enter into this Agreement, the Lease and the other documents contemplated hereby and thereby, perform its obligations hereunder and thereunder, and conduct its business as now being conducted.

-18-

5.2 Organization and Authorization of Sub. Sub is a corporation duly

organized, validly existing and with active status under the laws of the State of Florida. Sub has the requisite corporate power and authority to enter into this Agreement, the Lease and the other documents contemplated hereby and thereby, perform its obligations hereunder and thereunder, and conduct the business conducted at the Hospital Facilities.

5.3 Consents; Absence of Conflicts With Other Agreements, Etc. Sub's and

HMA's respective execution, delivery and performance of this Agreement, the Lease and the other documents contemplated hereby and thereby, and the consummation by Sub and HMA of the transactions contemplated hereby and thereby: (a) are within Sub's and HMA's corporate powers, and are not in contravention of law or of the terms of their respective certificates of incorporation or bylaws, and have been duly authorized by all appropriate action; (b) will neither conflict with nor result in any material breach or contravention of, or the creation of any lien under, any indenture, agreement, lease, instrument or understanding to which either Sub or HMA is a party or by which either is bound; (c) except as otherwise expressly provided in Schedule 5.3(c), do not require any approval or consent of, or any declaration or filing with, any Governmental Entity which is required by law or the regulations of any Governmental Entity; (d) will not violate, conflict with or constitute on the part of Sub or HMA a breach of or a default under any existing statute, law, rule or regulation of any Governmental Entity or any agreement, indenture, mortgage or lease to which Sub or HMA may be subject; and (e) will not violate any order or judgment of any Governmental Entity to which Sub or HMA may be subject.

5.4 Binding Obligations. This Agreement is the valid and legally binding

obligation of Sub and of HMA, respectively, and is enforceable against Sub and HMA, respectively, in accordance with its terms, except as enforceability against Sub or HMA may be restricted or limited by any or all of the Enforceability Exceptions.

5.5 Regulatory Compliance. Each of Sub and HMA is in compliance in all

material respects with all applicable statutes, rules, regulations and requirements of all Governmental Entities having jurisdiction over Sub and HMA and their respective operations, including, without limitation, the Florida Agency for Health Care Administration, and each of Sub and HMA has timely filed all reports, data and other information required to be filed by it with such Governmental Entities where a failure to filed timely would have a material adverse effort on Sub or HMA. Neither Sub nor HMA has received written notice of a violation by it of any applicable ordinance or other law, order, regulation or requirement.

5.6 Litigation or Proceedings. No action or preceding before any

Governmental Entity has been instituted or, to the best of HMA's and Sub's knowledge, threatened against HMA or Sub which seeks to restrain or prohibit the transactions contemplated hereby or by the Lease.

5.7 No Broker's Fees. Neither Sub nor HMA has employed any investment

banker, broker, finder, agent or other intermediary in connection with the negotiation or consummation of this Agreement or the Lease or of any of the transactions contemplated hereby or thereby or as to which Regional or the County may have any liability for such investment banker's, broker's, finder's, agent's or other intermediary's financial advisory or similar fee.

ARTICLE 6. PRE-CLOSING ACTIONS BY REGIONAL

6.1 Transition; Access. Prior to Closing, each of Regional, Sub and HMA

shall have used reasonable efforts to ensure a smooth transition of operations and control of the Hospital Facilities to Sub. Without limiting the generality of the foregoing, Regional will have permitted to the officers and authorized representatives and agents of HMA (at HMA's expense) reasonable access to the medical staff, employees and other personnel of the Hospital Facilities, and to the Hospital Facilities, the Assets and the books and records of Regional and of the Hospital relating thereto, including the right to inspect the same and conduct audits and verifications thereof, provided however, that: (a) none of the foregoing violates patient or other confidentiality requirements or impairs any other privilege or requirement of confidentiality under law or contract; (b) HMA has first provided reasonable notice of such access and inspection and has conducted the same in such a manner as not to interfere unreasonably with the operation of the Hospital Facilities or the conduct of Regional's business; (c) no such inspections will have taken place, and no members of the medical staff, employees or other personnel of the Hospital will have been contacted by HMA, without HMA first having coordinated such inspection or contact with the Hospital's Chief Executive Officer or his designee; and (d) HMA and Sub, jointly and severally, will indemnify and hold Regional and the County harmless with respect to any damages or injury to the extent caused by HMA or Sub or their officers, agents, employees and contractors in conducting such inspections, audits or verifications.

6.2 Operations. Prior to Closing, Regional: (a) will have carried on the

business of the Hospital Facilities in substantially the same manner as they have, since January 1, 1997, been conducted and will have not made any material change in the operations, finance, accounting policies or real or personal property of the Hospital Facilities except in the ordinary course of Regional's business; (b) will not make offers of employment at the Hospital Facilities to any person for periods expiring subsequent to the Closing Date, except in the ordinary course of business; (c) will not enter into any contract or commitment, or incur or agree to incur any liability not in the ordinary course of business, except for those offers, contracts, commitments and liabilities currently

-20-

under negotiation and identified in Schedule 6.2(c); (d) will not increase the compensation payable or to become payable or make a bonus payment to or otherwise enter into one or more bonus agreements with any employee or agent, except in the ordinary course of business in accordance with existing personnel policies or contracts or except as identified in Schedule 6.2(d); (e) will not create or assume any mortgage, pledge or other lien or encumbrance upon any of the Assets, irrespective of when acquired, other than Permitted Encumbrances, except in the ordinary course of business; (f) will not make or authorize any purchase order in excess of \$10,000, except in the ordinary course of business;

- (g) will not make or authorize any capital expenditures, except in the ordinary course of business and except as to those capital expenditures which will continue to be funded in the ordinary course of business prior to the Closing Date identified in Schedule 6.2(g), which Schedule includes the four major capital expenditure programs currently approved for the Hospital Facilities (the "Approved Capital Expenditures"); (h) will not sell, assign or otherwise transfer or dispose of any property, plant or equipment (other than supplies), except in the ordinary course of business or as contemplated in this Agreement; (i) will not take any action outside the ordinary course of business that would materially affect the Assets or the operation of the Hospital Facilities; (j) will have kept in full force and effect its insurance policies (unless replaced with comparable coverages); and (k) will have used reasonable efforts to (i) maintain and preserve the business organization of the Hospital Facilities intact, (ii) retain employees at the Hospital Facilities (except for employment terminations in accordance with past practices), (iii) maintain relationships with physicians, consistent with the bylaws, rules and regulations of the medical staff of the Hospital, and (iv) maintain relationships with suppliers, customers and others having business relations with the Hospital Facilities consistent with the terms of such relationships.
  - 6.3 Regulatory Approvals. Prior to Closing, Regional will have fully and

reasonably cooperated with Sub and HMA in Sub's and HMA's efforts to obtain, as promptly as practicable, all approvals, authorizations and clearances of Governmental Entities required to consummate the transactions contemplated hereby and by the Lease. The County and Regional shall be apprised of Sub's and HMA's efforts in obtaining such approvals, authorizations and clearances and shall be provided with copies of correspondence pertaining to requests and applications submitted by Sub and/or HMA.

6.4 Additional Financial Information. Prior to Closing, Regional will

have delivered to HMA true and complete copies of the unaudited balance sheets and the related unaudited statements of revenues and expenses of the Hospitals (if any) that have been prepared by Regional for each month then ended subsequent to the date of the Financial Statements, which will have been prepared from and in accordance with the books and records of the Hospitals and will fairly present in all material respects the financial position and results of operations of the Hospitals as of the dates and for the periods indicated except for the absence of year end adjustments and the absence of footnote information.

-21-

- 6.5 Additional Requirements. Prior to the date of this Agreement Regional
- shall have delivered to Sub the Survey.

reasonable effort to cause the conditions specified in Articles 8 and 9 over which Regional has reasonable control (and which are not assigned to Sub or HMA as duties under this Agreement) to have been satisfied as soon as reasonably practicable.

6.6 Closing Conditions. Prior to Closing, Regional will have used

- ARTICLE 7. PRE-CLOSING ACTIONS BY SUB AND HMA
- 7.1 Regulatory Approvals. Prior to Closing, Sub and HMA will have used

all reasonable efforts to obtain the transfer or issuance, as the case may be, as promptly as practicable, of all required Licenses and Permits as set forth in Schedule 8.3(b), and all approvals, authorizations and clearances of Governmental Entities required to consummate the transactions contemplated hereby and by the Lease and to allow Sub to operate the Hospital Facilities

pursuant to the terms of the Lease. HMA shall pay the filing fee with respect to the Hart-Scott-Rodino filing.

7.2 Closing Conditions. Prior to Closing, Sub and HMA will have used

reasonable efforts to cause the conditions specified in Articles 8 and 9 over which Sub and HMA have reasonable control (and which are not assigned to Regional as duties under this Agreement) to have been satisfied as soon as reasonably practicable.

ARTICLE 8. CONDITIONS PRECEDENT TO SUB'S AND HMA'S OBLIGATIONS

This Agreement is executed by Sub and HMA subject to satisfaction, on or before the Closing Date, of all of the conditions precedent set forth in this Article 8, any of which may be waived in writing by HMA:

- 8.1 Representations, Warranties and Covenants. The representations and
- warranties of Regional contained in this Agreement will be true in all material respects on and as of the Closing Date, and the covenants under this Agreement required to have been complied with or performed by Regional and County before Closing will have been duly complied with or performed in all material respects.
- 8.2 Opinion of Regional's Counsel. Sub and HMA will have received an opinion from Carlton, Fields, Tampa, Florida, counsel to Regional, dated as of the Closing Date and addressed to Sub and HMA, in the form set forth in Schedule 8.2(a).

-22-

8.3 Pre-Closing Confirmations. Sub and HMA will have received (a) the

Bankruptcy Order from the U.S. Bankruptcy Court for the Middle District of Florida approving the transactions under this Agreement; and (b) confirmation from the appropriate Governmental Entities that the specific Licenses and Permits and Governmental Approvals described in Schedule 8.3(b) will have been or will be transferred or issued, as the case may be, in the name of Sub on the Closing Date.

- 8.4 Hart-Scott Rodino. The waiting period under the Hart-Scott Rodino Act
  -----shall have expired or been terminated.
  - 8.5 Actions or Proceedings. No order, ruling or judgment of any

Governmental Entity will remain in effect which restrains, enjoins and/or otherwise prohibits the completion of the transactions contemplated hereby or by the Lease.

8.6 Survey and Leasehold Title Insurance Policy. Contemporaneously with

the signing of this Agreement, HMA and Sub hereby acknowledge receipt of the Survey and the Title Commitment and hereby accept the same in the form as delivered and will accept title to the Assets, on the Closing Date, subject to the matters set forth in the Survey, the Title Commitment and Permitted Encumbrances, provided that there is no intervening instrument placed of record or action of Regional and/or the County that would otherwise impair delivery of Good Title to the Assets by Regional and/or County and provided that all of the instruments required to satisfy the conditions set forth in Schedule B - Section 1, of the Title Commitment have or will be delivered in a form sufficient for the title insurance company issuing the Title Commitment to delete all of the Schedule B - Section 1, requirements.

8.7 No Adverse Change. Since December 31, 1997, no event, condition, or

change will have occurred with respect to the operations, financial condition, assets, liabilities (contingent or otherwise) for which Sub or HMA will be liable, or the income or business of the Hospital Facilities, that materially, adversely impairs or will materially adversely impair the use or value to Sub of the Assets or of the Hospital Facilities, other than those events, conditions or changes resulting from any or all of the following: (a) seasonal factors; (b) general economic, industry or labor conditions; (c) operating losses incurred in the ordinary course of business; (d) changes or proposed changes to laws, regulations, rules, reimbursement rates or policies of governmental agencies or bodies; (e) requirements, reimbursement rates, policies or procedures of third party payors or accreditation commissions or organizations that are generally applicable to hospitals (or hospitals within a region of the United States) or to those hospitals that are similar to the Hospitals; (f) national initiatives (such as with respect to DRG 79 or laboratory unbundling or coding) that are generally applicable to hospitals (or hospitals within a region of the United States) or to those hospitals that are similar to the Hospitals; (g) financial difficulties of third party payors; (h) payments with respect to Regional's creditors or the compromise, forgiveness, reduction, satisfaction, or elimination of

-23-

any of Regional's debts; or (i) the payment of costs incurred in Regional's bankruptcy proceeding by certain of Regional's creditors and the County.

8.8 Instruments from County. The County shall have delivered a certified

copy of the Resolution of the Board of County Commissioners of Hernando County, Florida, authorizing the signing and delivery to Sub of the Lease. In addition, the County shall deliver the instruments required to be delivered by the County set forth in Schedule B, Section 1, of the Title Commitment; and

8.9 Other Instruments and Documents. Regional shall have delivered to Sub
-----and HMA each of the instruments and documents required to be delivered by it by

### ARTICLE 9. CONDITIONS PRECEDENT TO REGIONAL'S OBLIGATIONS

Section 3.3.

This Agreement is executed by Regional subject to satisfaction on or before the Closing Date, of all of the conditions precedent set forth in this Article 9, any of which may be waived in writing by Regional:

- 9.1 Representations, Warranties and Covenants. The representations and warranties of Sub and HMA contained in this Agreement will be true in all material respects as of the Closing Date; and the covenants in this Agreement required to be complied with or performed by Sub or HMA before Closing will have been duly complied with or performed in all material respects.
- - 9.3 Pre-Closing Confirmations. Regional and County will have received:
- (a) the Bankruptcy Order from the U.S. Bankruptcy Court for the Middle District of Florida approving the transactions under this Agreement and the Lease; and (b) confirmation from the appropriate Governmental Entities that the specific Licenses and Permits and Governmental Approvals described in Schedule 8.3(b)

will have been or will be transferred or issued, as the case may be, in the name of Sub on the Closing Date.

- 9.4 Hart-Scott Rodino. The waiting period under the Hart-Scott Rodino Act
- shall have expired or been terminated.
  - 9.5 Actions or Proceedings. No action or proceeding before any

Governmental Entity will remain in effect which restrains, enjoins and/or otherwise prohibits the transactions contemplated hereby or by the Lease.

-24-

9.6 Other Instruments and Documents. HMA and Sub will have delivered to

Regional each of the instruments and documents required to be delivered to it pursuant to Section 3.2.

9.7 Instruments From County. The County shall have delivered a certified

copy of the Resolution of the Board of County Commissioners of Hernando County, Florida, authorizing the signing and delivery to Sub of the Lease. In addition, (a) the County and Sub shall have executed and delivered the Lease, and (b) Regional and the County shall have executed and delivered an agreement with respect to how the Rent shall be applied and with respect to the payment or discharge or provision for payment or discharge of Regional's liabilities and obligations.

#### ARTICLE 10. INDEMNIFICATION

10. Indemnification by HMA. To the extent, and subject to the

limitations, provided in this Article 10, from and after the Closing Date, HMA and Sub, jointly and severally, will indemnify, protect, defend and hold harmless Regional and its directors, trustees officers and Affiliates, from and against:

- (a) any Losses suffered by any or all of them with respect to any or all of the Assumed Liabilities;
- (b) any Losses suffered by Regional arising out of the ownership, use or operation of the Purchased Assets or the use or operation of the Leased Assets on or after the Closing Date, including, without limitation, any Medicare, Medicaid or any other third party payor claim relating to any period on or after the Closing Date; and (iii) all sales taxes, documentary or other excise or transfer taxes that Sub and HMA are specifically responsible for under the terms of this Agreement, the Lease and other agreements related thereto; and
- (c) any Losses suffered by Regional arising out of or resulting from any misrepresentation, breach of warranty or breach or nonfulfillment of any covenant or agreement on the part of Sub or HMA contained in this Agreement.
- 10.2 Indemnification by Regional. To the extent, and subject to the

limitations, provided in this Article 10, from and after the Closing Date Regional will indemnify, protect, defend and hold harmless Sub and HMA, and their respective directors and officers, solely from the Indemnity Escrow Fund described in Section 10.4(b), for a period of one year from the Closing Date from and against:

(a) any Losses suffered by Sub or HMA (except with respect to any or all of the Assumed Liabilities) arising out of (i) any liability, obligation, claim against or contract of Regional incurred

-25-

- prior to the Closing Date, or (ii) any negligence or intentional misconduct of Regional prior to the Closing Date; and
- (b) any Losses suffered by Sub or HMA arising out of or resulting from any misrepresentation, breach of warranty or breach or nonfulfillment of any covenant or agreement on the part of Regional contained in this Agreement.
- 10.3 Survival. The representations and warranties made by Regional in this

Agreement shall survive the Closing for 12 months after the Closing Date and the representations and warranties of Sub and HMA in this Agreement shall survive the Closing for thirty-six (36) months after the Closing Date, and on the 12th or 36th month anniversary date of the Closing Date, as applicable, (respectively, the "Indemnification Period") shall expire, together with any right to indemnification for breach thereof. However, to the extent notice of such claim shall have been given with respect to such representation or warranty prior to the expiration of the applicable Indemnification Period in accordance with this Article 10 by the party seeking indemnification (the "Indemnitee") to the party from which indemnification is being sought (the "Indemnitor"), in which case the representation or warranty alleged in the notice to have been breached shall survive, to the extent of the claim set forth in the notice only, until such claim is resolved. The covenants and agreements contained herein (other than the covenants and agreements to indemnify against breaches of representations and warranties, which shall expire as described above) shall survive the Closing until the covenants and agreements are complied with in accordance with their terms.

### 10.4 Limitations.

-----

(a) As used in this Article 10, the term "Losses" includes only losses actually paid or incurred, and does not include: (i) any amounts recovered, or to which an Indemnitee is entitled, from any surety, insurance carrier or third party obligor, or the cost of maintaining any surety or insurance policies, or under any warranty or other claim or right, and no right of subrogation against the Indemnitor will accrue hereunder to or for the benefit of any surety, insurance company or any third party; (ii) any cost or expense previously counted in determining Losses or taken into account in any adjustments under Section 2.6. The Indemnitee agrees to submit in a timely manner to any applicable surety, insurance carrier or third party obligor, or under any warranty, claim or other right, all claims for indemnifiable Losses for which such entity may have liability.

-26-

(b) Any recovery by Indemnitees for indemnification shall be limited as follows: (i) Sub and HMA shall not be entitled to any recovery unless a claim for indemnification is made (A) in accordance with Section 10.3, and (B) within the time period of survival set forth in Section 10.3; (ii) Indemnitees shall not be entitled to recover any Losses unless and until the amount which all Indemnitees are entitled to recover in respect of such Losses exceeds, in the aggregate, \$100,000 (the "Deductible"), in which event (subject to clause (iii) below) Indemnitees shall be

entitled to recover in respect of such Losses only the amount of the Losses that exceed the Deductible; and (iii) any amounts recoverable by Sub and HMA as Indemnitees with respect to such Losses shall be payable solely from the Indemnity Escrow Fund, established under the Escrow Agreement, to be entered into at Closing between Regional, HMA, County and First Union National Bank, as Escrow Agent (a copy of which is attached as Schedule 10.4(b)), and they shall have no other recourse to Regional in respect of indemnification claims under this Agreement or otherwise. The Indemnity Escrow Fund will be in an amount of One Million Five Hundred Thousand and 00/100 Dollars (\$1,500,000.00) and be funded out of the purchase proceeds to be paid by Sub pursuant to Section 2.5 hereof. No Losses shall be included in determining whether the Deductible has been reached unless a notice seeking indemnification for such Losses has been given by the Indemnitee to the Indemnitor in accordance with Article 10. Copies of any and all written notices sent pursuant to this Section 10.4 shall be mailed to the County at the following address: c/o Hernando County Board of County Commissioners, County Courthouse, 20 North Main Street, Room 460, Brooksville, Florida 34601.

(c) Notwithstanding anything to the contrary contained in this Agreement or otherwise, the aggregate amount of liability of Regional to all Indemnitees under this Agreement (including costs, expenses and attorneys' fees paid or incurred in connection therewith, in connection with the curing of any and all misrepresentations or breaches of warranties or covenants under this Agreement) shall not exceed the amount deposited in the Indemnity Escrow Fund and earnings thereon, and HMA and Sub shall be entitled to recover their Losses solely against such Indemnity Escrow Fund and not against any other assets of Regional or the County.

-27-

- (d) Notwithstanding anything to the contrary contained in this Agreement, Regional shall not be liable under the provisions of Article 10 hereof or otherwise have any liability for any misrepresentation or breach of warranty, covenant or agreement under this Agreement or otherwise have any liability with the transactions contemplated by this Agreement to the extent that the existence of such liability, the breach of warranty or covenant or the falsity of the representation or warranty upon which such liability would be based is clearly disclosed in any of the contracts and documents referred to in this Agreement, in the Schedules attached hereto or which is disclosed in a written notice furnished to Sub or HMA prior to the Closing.
- (e) Notwithstanding anything to the contrary contained in this Agreement, Sub and HMA shall not have the right to elect not to close the transactions contemplated by this Agreement by reason of any misrepresentation or breach of warranty or covenant contained in this Agreement or otherwise if: (i) Regional would have no liability therefor to Sub or HMA by reason of the provisions contained in this Section 10.4 or elsewhere in this Agreement; or (ii) Regional undertakes, at its sole cost and expense (but subject to the limitations and other provisions contained in this Article 10 or elsewhere in this Agreement), to promptly cure such misrepresentation or breach (or defend, settle, compromise or discharge any third party claim which forms the basis thereof) as soon as practicable after the Closing.
- (f) The indemnification provided in this Article 10 shall be the sole

and exclusive remedy, after the Closing, with respect to any inaccuracy or breach by Regional and/or HMA and Sub of any representation, warranty or covenant. It is acknowledged that the application of Article 10 will not impact the rights of any party hereto to payment of any required adjustment amounts as determined pursuant to Section 2.6 hereof.

(g) Upon making any payment to an Indemnitee for any indemnification claim pursuant to this Article 10, the Indemnitor shall be subrogated, to the extent of such payment, to any rights which the Indemnitee may have against any other parties with respect to the subject matter underlying such indemnification claim.

-28-

(h) No Losses shall be recoverable to the extent that Sub or HMA had knowledge, prior to the Closing, of such Losses or the facts or circumstances that might give rise to such Losses.

### 10.5 Indemnification Procedure.

\_\_\_\_\_\_

(a) As a condition precedent to a claim under this Article 10, a party seeking indemnification hereunder (the "Indemnified Party") will promptly give to the party from whom indemnity is sought (the "Indemnifying Party") notice of any matter which the Indemnified Party has determined has given or could give rise to a right of indemnification under this Agreement, which notice will specify in reasonable detail the nature of the claim, the basis for indemnification and the amount and nature of the Losses for which indemnification is sought. If a claim by a third party is made against an Indemnified Party and the Indemnified Party intends to seek indemnity hereunder with respect thereto, the Indemnified Party will promptly so as to not prejudice the Indemnifying Party's defenses of, or other obligations with respect to, such claim (and in any case within 30 days of such claim being formally made) give the Indemnifying Party notice of such claim. The Indemnified Party will cooperate with the Indemnifying Party in such defense and make available to the Indemnifying Party all pertinent records, materials and information in its possession or under its control relating thereto as is reasonably requested by the Indemnifying Party. If and to the extent that the Indemnifying Party does not, within 20 days after receipt of such notice, object, then the Indemnifying Party will have 30 days after receipt of such notice to pay the Losses specified in the notice or to assume and control the defense of such third-party claim (including any appeals with respect thereto) at its expense and through counsel of its choice. If the Indemnifying Party elects not to defend against such claim, then it will promptly so notify the Indemnified Party and, in such event, the Indemnified Party will then be entitled, at its option, to assume and control the defense of such claim (including any appeals with respect thereto) at its expense and through counsel of its choice. In such event, if it is finally determined that the Indemnifying Party failed to assume the defense of a claim for which it is liable hereunder, then the expense of defending

-29-

such claim will be borne by the Indemnifying Party.

(b) The Indemnifying Party will be subrogated to any and all

defenses, claims and setoffs which the Indemnified Party asserted or could have asserted against the third party making a claim. The Indemnified Party shall execute and deliver to the Indemnifying Party such documents as may be reasonably necessary to establish by way of subrogation the ability and right of the Indemnifying Party to assert such defenses, claims and setoffs.

### ARTICLE 11. TERMINATION

11.1 Optional Termination. This Agreement may be terminated and the

transactions contemplated hereby abandoned at any time prior to the Closing, as follows:

- (a) By the mutual written consent of Sub and Regional; or
- (b) By Regional, by giving notice to Sub, if any of the conditions set forth in Article 9 shall not have been satisfied or waived in writing on or before May 15, 1998; provided that Regional shall not then be in material breach of this Agreement; or
- (c) By Sub, by giving notice to Regional, if any of the conditions provided in Article 8 shall not have been satisfied or waived in writing on or before May 15, 1998; provided that neither Sub nor HMA shall then be in material breach of this Agreement.

Notwithstanding Section 11.1(c), prior to terminating this Agreement, Sub and HMA shall give Regional prompt notice of any failure of any of the conditions provided in Article 8 to be satisfied and prompt notice of any breach actually known by Sub or HMA, of any representation, warranty, covenant or agreement of Regional under the Agreement, and Regional shall have until and including June 29, 1998, to cure such breach.

11.2 Actions on Termination; Effect of Termination. In the event this

Agreement is terminated as provided above, Sub or HMA shall deliver to Regional all documents (and copies thereof in its possession) concerning Regional previously delivered by Regional to Sub or HMA, and none of the parties shall have any liability to the other party for costs, expenses, loss of anticipated profits, consequential damages, or otherwise, except for any breach by such party of any of the provisions of this Agreement and except as provided in Section 11.3. Notwithstanding the foregoing, this Section 11.2, Section 11.3 and Sections 12.3, 12.4, and Article 13

-30-

shall survive any termination of this Agreement.

### 11.3 Termination Payments.

- (a) If the conditions precedent under this Agreement to the obligations of a party (i.e., Sub and HMA, on the one hand, or Regional and County, on the other hand) (the "First Party") to close the transactions under this Agreement have been fulfilled, and the other party or parties (the "Second Party") are ready, willing and able to close the transactions under this Agreement, but the First Party fails or refuses to close the transactions under the terms of this Agreement, the First Party shall promptly pay to the Second Party the sum of Five Million Dollars (\$5,000,000) by wire transfer of immediately available funds.
- (b) In the event that the transactions under this Agreement do not close because Sub or HMA fails to or is unable to obtain

approval of the transfer or issuance, as the case may be, of any or all of the Licenses and Permits described in Schedule 8.3(b) or the expiration or termination of the waiting period under the Hart-Scott-Rodino Act (collectively referred to as "Governmental Approvals") and the inability to obtain any or all of the Governmental Approvals was not caused by (a) Regional's or County's acts or omissions or (b) governmental moratorium, legislation, or other official governmental regulations impacting on others seeking similar approvals for the transfer of other hospital or healthcare related facilities which prohibit issuance of the Governmental Approvals, then HMA shall promptly pay to Regional the sum of Five Million Dollars (\$5,000,000) by wire transfer of immediately available funds.

(c) The amounts set forth in this Section 11.3 have been agreed upon by the parties as the sole liquidated damages with respect to the matters to which such amounts relate and are not a penalty. The parties have determined that actual damages with respect to such matters would be difficult to ascertain. It is recognized that as to Sub and HMA that the damage amounts described in 11.3(a) and (b) are mutually exclusive in application and that under no circumstances could HMA and Sub be liable for payments under this Section 11.3 that would exceed the aggregate sum of \$5,000,000.

-31-

### ARTICLE 12. FURTHER COVENANTS

12.1 Terminating Cost Reports. Following the Closing Date, Sub and HMA

will fully assist Regional, at Regional's request, in the preparation and timely filing all terminating cost reports, claims and other reports and documentation required by law or contract to be filed by Regional with Medicare, Medicaid or other third-party payor programs in respect of the Hospital Facilities for operations prior to the Closing Date or as a result of the consummation of the transactions contemplated hereby or by the Lease. Such assistance will include the reasonable use of the Hospital Facilities, personnel and accounting systems of the Hospital Facilities. Notwithstanding the foregoing, Regional will be solely and financially responsible for the contents of all such reports, claims and documentation. Upon completion thereof, Regional will promptly execute and file all such reports, claims and documentation. Regional shall be solely entitled to all refunds and reimbursements and be obligated for any deficiencies relating to periods prior to the Closing Date and from additional interest payments with respect to Regional's presently outstanding bonds. Sub and HMA acknowledge and agree that in connection with the transactions under this Agreement, Regional or the Hospitals will pay unpaid interest on its bonds and notes of approximately Eighteen Million One Hundred Thousand Dollars (\$18,100,000) (the "Accrued Interest") with due credit as interest for partial payments and distributions from the Debt Service Reserve Fund made to Continental Casualty Company (CNA) during Regional's bankruptcy case. After the payment of such Accrued Interest, Regional will make an adjustment to its cost reports for the applicable prior years of reimbursements made to Regional, or its Affiliates, under Medicare and Medicaid programs for services rendered in prior years. Such adjustments are anticipated to result in additional reimbursements to Regional under Medicare in the approximate amount of Five Million Dollars (\$5,000,000) and in additional reimbursements under Medicaid in the approximate amount of One Million Dollars (\$1,000,000). Sub and HMA hereby agree that Regional is entitled to all of such additional reimbursements and that Sub and HMA shall promptly pay to Regional, without deduction, offset, claim or hold-back, all of such additional reimbursements if received by Sub, HMA or any of their Affiliates.

12.2 Insurance Coverage. Following the Closing Date, Regional will

-----

maintain, at its own expense, "tail coverage" for acts, errors and omissions of professional liability incurred by Regional prior to the Closing Date with Sub listed as an additional insured.

12.3 Confidentiality. Prior to the Closing Date, Sub and HMA (treated as

one party for this purpose) and Regional (each, the "First Party") will, and will use all reasonable efforts to cause its Affiliates, employees, representatives and agents to, hold in strict confidence all Confidential Information of the other party (each, the "Other Party"), unless compelled to disclose the same by judicial or administrative process or, in the opinion of counsel, by other requirements of law; provided, however, that in either such case the First Party will provide the Other Party with prompt prior

-32-

notice thereof so that the Other Party may seek a protective order or other appropriate remedy or waive compliance with the provisions of this Section 12.3. In the event that such protective order or other remedy is not obtained, or the Other Party waives compliance with the provisions hereof, the First Party will furnish only that portion of Confidential Information which, in the written opinion of the First Party's counsel, is required, and the First Party will exercise all reasonable efforts to obtain reliable assurance that confidential treatment will be accorded such of the disclosed Confidential Information as the Other Party so designates. The First Party will not otherwise disclose Confidential Information to any person, except with the consent of the Other Party. For the purposes hereof, "Confidential Information" means all information of any kind concerning the Other Party or any of its Affiliates, obtained directly or indirectly from the Other Party or any of its Affiliates, employees, representatives or agents in connection with the transactions contemplated hereby or by the Lease, except information (a) ascertainable or obtained from public or published sources, (b) received from a third party not under an obligation to keep such information confidential, (c) which is or becomes known to the public (other than through a breach of this Agreement), or (d) which was in the First Party's possession prior to disclosure thereof to the First Party and which was not subject to any obligation to keep such information confidential. The First Party recognizes that any breach of the provisions of this Section 12.3 would result in irreparable harm to the Other Party and its Affiliates and, therefore, that the Other Party will be entitled to an injunction to prohibit any such breach or anticipated breach, without the necessity of posting a bond, cash or otherwise, in addition to all of its other legal and equitable remedies.

12.4 Costs of Transaction. Whether or not the transactions contemplated  $\_\_\_\_$ 

hereby or by the Lease are consummated, and except as otherwise expressly provided herein or in the Lease, each party will bear its own fees, expenses and disbursements, and those of its Affiliates, agents, representatives, accountants, counsel and investment bankers, incurred in connection with the subject matter hereof.

12.5 Preservation of Books and Records. Until the later to occur of (a)

the final adjudication of any dispute or investigation arising out of the business, operations or affairs of the Hospital Facilities before the Closing Date, or (b) 60 days following the running of applicable statutes of limitations, Sub will maintain all books and records of the Hospital Facilities constituting a part of the Assets which relate to the use, operations or maintenance of the Assets or Hospital Facilities prior to the Closing Date, and Regional or its assignee will maintain all such books and records not constituting a part of the Assets, in each case to the extent reasonably necessary in connection with any tax, Medicare or Medicaid or other liability or matter for any period ending before the Closing Date. After the expiration of

such period, neither Sub nor Regional will destroy any of such books or records without giving the other the opportunity, at the latter's sole expense, to take possession thereof.

12.6 Cooperation. The parties will pursue and perform with all due

diligence all acts, applications, authorizations and consents necessary or appropriate to the fulfillment of the provisions of this Agreement and the Lease, and will cooperate with each other and execute any and all documents reasonably incident thereto. In addition, for five years following the Closing Date, each party (treating Sub and HMA as one party for this purpose) will, upon reasonable notice, during normal business hours, at the expense of the requesting party, only to the extent reasonably necessary to facilitate the transactions contemplated hereby or by the Lease, audits, compliance with governmental requirements and regulations and the prosecution or defense of third-party claims, and only to the extent that it does not unreasonably interfere with its business operations: (a) afford to the representatives of the other, including its counsel and accountants, full and complete access to such records and information as may be available relating to the assets and the Hospital Facilities for periods prior to and subsequent to the Effective Time, and full and complete access to its officers and employees; and (b) cooperate with, and use all reasonable efforts to cause its officers and employees to cooperate with the other and with appropriate Governmental Entities and third parties, in furnishing information, evidence, testimony and other reasonable assistance.

12.7 Special Post-Closing Covenants. HMA shall, and shall cause Sub to, \_\_\_\_\_comply with the following covenants:

(a) Governing Board. Sub will establish and maintain, at all times

during the term of the Lease, a local governing board of the Hospitals (the "Governing Board") consisting of fifteen (15) members initially comprised of the present members of the Board of Directors of Regional and the Executive Director and Operations Vice-President of the Hospitals. HMA may increase the size of the Governing Board at any time and from time to time with an additional two (2) members selected by HMA. In the event that either or both of the two additional members are members of the medical staff of the Hospitals, the prospective candidates must be approved by the Medical Executive Committee of the candidate's Hospital prior to final selection by HMA. The size of the Governing Board, however, may neither be reduced nor increased (except for the two (2) additional representatives that may be selected by HMA, as described above) without the consent of the Governing Board. At all times during the term of the Lease, not less than the greater of ten (10) or a majority of the members of the Governing Board, will be residents of Hernando County who are not officers, employees or agents of HMA or any of its Affiliates or directors of HMA or any of its other Affiliates. The terms of each member of the Governing Board other than

-34-

those members selected by HMA shall be staggered, with a maximum term for any member of three (3) years. Each member of the Governing Board may serve for two consecutive terms, and then is required to remain off the Governing Board for one year after the completion of such successive terms. If a Governing Board

member dies, voluntarily leaves such position sooner or is removed from such position by a vote of the Governing Board or if a Governing Board member's term expires, or if the size of the Governing Board is increased beyond seventeen (17) members, the vacancies and/or additions so created shall be filled by the Governing Board recommending one or more candidates to fill each of such positions to HMA for HMA's sole approval, which approval shall not be unreasonably withheld. Notwithstanding the preceding sentences, HMA may select up to four (4) members of the Governing Board (including the Executive Director and Operations Vice President of the Hospitals), as described in this sub-section (a) above, without recommendations from the Governing Board as to such members. A quorum of the Governing Board shall consist of a majority of its members present in person or by means of communications equipment whereby all members of the Governing Board can hear each other simultaneously. Actions by the Governing Board shall be by a majority of a quorum. The Governing Board's powers and responsibilities shall include:

- (i) Adopting a hospital vision, mission and values statement, and any amendments thereto, assisting in developing policies and monitoring progress toward strategic goals;
- (ii) For the term of the Lease the Governing Board shall have the right to approve the hiring or termination of the Executive Director of the Hospitals, which approval shall not be unreasonably withheld;
- (iii) Approval of any change of the name of any of the Hospitals;

-35-

- (iv) Participating on an advisory basis in development of and reviewing all operating and capital budgets and facility plans;
- (v) Granting medical staff membership and clinical privileges and, when necessary, taking disciplinary action consistent with credentialing processes, medical staff by-laws and strategic plans;
- (vi) Monitoring medical staff compliance with JCAHO and Hospital Facilities compliance with HCFA's conditions of participation in the Medicaid and Medicare programs;
- (vii) Fostering community relationships;
- (viii) Review of operating information provided by Sub regarding performance of the Hospital Facilities for the period from the prior meeting to the then current meeting of the Governing Board. In addition to financial statements and operating statistics for the Hospital for the aforesaid period, the Governing Board shall also be provided with the results of an annual audit of the Hospitals.

determines that any material services or facilities in the Hospital Facilities should be changed, reduced or eliminated, Sub may change, reduce or eliminate such services or facilities only upon the prior approval of the Governing Board and County.

- (c) Indigent Care. HMA will cause Sub to provide the same level of
  - care to indigents currently provided by the Hospitals as currently set forth in the indigent care policy of the Hospitals as attached as Schedule 12.7(c). Any patient presented to the emergency room who has a medical emergency or who, in the judgment of a staff physician, has an immediate emergency need will be treated in accordance with such policies. No such patient will be turned away because of age, race, gender or inability to pay. HMA and Sub will ensure that the Hospitals will continue to provide medical services to patients covered by the Medicare and Medicaid programs.

-36-

providing the equipment and capital improvements that are and will be needed by the Hospital Facilities, including the Approved Capital Expenditures, to serve the medical needs of the community through state-of-the-art facilities. Sub shall fund and pay, and HMA shall cause Sub to fund and pay, for capital expenditures and improvements to the Hospital Facilities, including the Approved Capital Expenditures, a minimum sum of Twenty-Five Million Dollars (\$25,000,000.00) which sum shall be used to pay the cost determined by HMA to complete the Approved Capital Expenditures and the balance to pay for such capital expenditures and improvements to be allocated among the Hospital Facilities as HMA deems appropriate, all of which shall be funded on or before the fifth anniversary of the Closing Date. All capital expenditures shall be as defined by GAAP. At least annually and from time to time during each year of the Lease, HMA shall review and consult with the Governing Board with respect to capital expenditure programs and in the development of a strategic plan for the Hospital Facilities. HMA shall also seek the advice and input of the Governing Board regarding future capital expenditures and shall consider making changes reasonably suggested by the Governing Board to the capital expenditure program; however, HMA is under no obligation to accept such suggestions made by the Governing Board. HMA shall implement the capital expenditure programs for the Hospital Facilities that HMA approves after consultation with the Governing Board.

(e) Hospital Employees. HMA will cause Sub to offer employment on an

"at will" basis, as of the Closing Date, and at their salaries on the Closing Date, to all of the employees of Regional (including senior management) who are active employees of Regional as of the Closing Date. In addition to employees of Regional, HMA will cause Sub to offer employment on an "at will" basis at their present salaries to both Thomas Barb and Gregory Mason, currently employees of Quorum Management that are assigned to the Hospital Facilities. Sub will also assume and agrees to provide Thomas Barb with the same severance arrangement that Thomas Barb has as set forth in that certain letter agreement by and between Thomas Barb and RHI dated

October 30, 1997. All of the individuals employed by Sub will retain their current seniority, as of the Closing Date, with regard to vacation and sick time accrual under Sub's plans. Such employees will also be given credit for any years of service with Regional or Quorum Management, as the case may be, for purposes of qualified retirement plans that cover Sub's

-37-

employees with respect to eligibility, participation and vesting. With respect to health and disability insurance coverage, exclusions for pre-existing conditions and applicable waiting periods will be waived. The salaries of the employees at the Closing Date will provide the base for future merit increases. In the event of an employee reduction, HMA will cause Sub to pay severance based on position and length of service with Regional, Quorum, Sub and HMA in accordance with a severance policy that is applicable generally to all HMA employees. In addition to a severance payment, eligible employees will also have normal and customary rights to health plan continuation coverage as mandated by Section 4980B of the Code or applicable state health care continuation coverage statutes. Sub agrees that it will not effect any reductions in work force for a period of twelve (12) months after the Closing Date other than reductions through attrition, flexible staffing for seasonal adjustments, or decreases in patient census.

after the first anniversary of the Closing Date, but prior to the third anniversary of the Closing Date, to in good faith "unwind" the transactions contemplated hereby, to reacquire the Leased Assets, to terminate the Lease and to purchase the then remaining portion of the Purchased Assets and any substitutions, replacements and additions thereof, by paying an amount equal to the product of (a) EBITDA for the Hospital Facilities, multiplied by (b) seven (7). Immediately prior to the date of closing of the "unwind" ("Unwind Closing") Sub shall present to the County its calculation of EBITDA (Earnings Before Interest, Taxes, Depreciation and Amortization) determined in accordance with GAAP for the twelve (12) month period immediately preceding the date of the Unwind Closing. Within sixty (60) days after the date of the Unwind Closing, the County may have its auditors review Sub's determination of EBITDA. In the event there is any disagreement with respect to the determination of this amount, the County and Sub shall resolve their disagreement or, alternatively, the dispute procedure described in Section 2.6 hereof shall be used to resolve any differences. In order to exercise the unwind option, the County shall give Sub written notice thereof at least 180 days prior to the date of the Unwind Closing and shall tender the amount as determined by Sub above to Sub on the date of the Unwind Closing subject to adjustment as provided in this Section 12.7(f), and Sub, HMA and the County will cooperate reasonably with each other in effecting such "unwind"

-38-

(including matters of transition), execute such documentation (including conveyance and assignment documents) as is reasonably necessary to effect the "unwind" (including the transfer to the County of the Leased Assets and Purchased Assets, to the extent then available, the substitutions, replacements and additions

thereto, all assignable and transferable Licenses and Permits and other assets except those expressly excluded, all as set forth in Article III of the Lease) pursuant to this Subsection (f). Upon such an "unwind," Sub and HMA shall surrender to the State of Florida all licenses, certificates, certifications and approvals to operate the Hospitals. In addition, in the event of an "unwind" the County shall covenant not to sell or lease the Hospital Facilities to any other party for a period of three (3) years from the date of the Unwind Closing except to a qualified not-for-profit entity satisfying the following requirements: (i) the not-for-profit entity shall be a local entity totally controlled by Hernando County residents, (ii) the local not-forprofit entity shall not be an Affiliate of any other not-forprofit organization, (iii) the local not-for-profit entity shall operate and maintain the Hospital Facilities, (iv) the local not-for-profit entity shall not pay any rent or other remuneration to the County, other than nominal rent or remuneration, for the Hospital Facilities, and (v) the financing for the purchase and operation of the Hospital Facilities as described herein shall be accomplished by County using traditional governmental financing techniques.

(g) Assignment or Sublease. Sub shall have the right to assign its

interest in the Lease or to sublease all or any portion of the Premises or its real estate interest in the Premises or HMA or its Affiliates shall have the right to sell or transfer fifty percent (50%) or more of the voting stock of Sub or enter into any transaction changing the control of Sub (whether in a single transaction or a series of transactions) only with the prior written consent of County, which consent shall not be unreasonably withheld or delayed, if (a) the proposed assignee has a financial net worth of at least equal to that of HMA immediately prior to such assignment and has the experience to operate the Hospital Facilities; (b) the proposed assignee shall expressly assume and agree to perform all of Sub's obligations under the Lease, in writing, in form reasonably acceptable to County; (c) the proposed assignee is duly qualified to operate the Premises as two acute care hospital facilities, as determined by the County; and (d) the proposed assignee has experience in and a generally accepted positive reputation as an owner, operator

-39-

and/or manager of hospitals or other healthcare facilities. If the assignee meets the foregoing criteria, the assignment shall operate to release Sub and HMA from any further liability for any of the conditions, obligations, agreements, and covenants of the Lease required to be performed after the date of such assignment.

shall be deemed to have consented to such Business Combination if HMA (and all of its Affiliates) and any other company surviving such Business Combination (and all of its Affiliates) have, within twelve (12) months after the date on which such Business Combination becomes effective, divested themselves of their entire right, title and interest in the Other Hospitals, whether held directly or indirectly, and any interest, whether direct or indirect, in the management or operation of the Other Hospitals.

(i) Transfer of Control of HMA.

(1) In the event that, at any time during the Lease Term, as defined in the Lease, HMA proposes (i) a transfer or sale (whether in a single transaction or series of transactions) of fifty percent (50%) or more of the voting stock of HMA, or substantially all of the assets of HMA, to a third party which was not an Affiliate of HMA immediately prior to such transfer or sale, or (ii) the merger or consolidation of HMA with or into a third party which was not an Affiliate of HMA immediately prior to such merger or consolidation (other than a merger or consolidation in which HMA is the surviving entity thereof) (each a "Disposition"), then the County shall have the following options: (a) approve the Disposition and thereby consent to the transaction by and between the third party and HMA; or (b) disapprove the

-40-

Disposition to the third party.

- (2) In the event the County disapproves the Disposition as described in Section 12.7(i)(1)(b), the County shall elect one of the following options: (i) County may purchase the assets of Sub that would be transferred to County at the time of termination of the Lease (the "Termination Assets"), within one hundred and eighty (180) days from the date the County disapproves the Disposition, upon payment of the fair market value for the Termination Assets as hereinafter described (the "Lease Asset Value"), or (ii) the County will, within one hundred and eighty (180) days from the date the County disapproves the Disposition, find a purchaser acceptable to County that will purchase and close on the Termination Assets as provided for herein. If the County does not exercise its rights under Section 12.7(i)(2)(i) or Section 12.7(i)(2)(ii) within the time frame set forth herein, the County shall be deemed to have consented to the Disposition and to the transaction by and between the Acquiror (as herein defined) and HMA.
- (3) In the event a transaction as described in Section 12.7(i) is proposed, HMA shall give written notice to County ("Disposition Notice"), containing the name, address, qualifications, financial information and background of the third party with whom the Disposition is proposed (the "Acquiror").
- (4) The Disposition Notice shall also set forth the Lease Asset Value for the Termination Assets. County shall, within ninety (90) days after receipt of the Disposition Notice, either accept or reject the Disposition. In addition, if County disagrees with HMA's determination of the Lease Asset Value, then County and HMA shall each select an MAI

commercial real estate appraiser, with at least 10 years' experience in appraising hospital facilities, in the southeast region of the United States. Upon selection, County's and HMA's appraisers will work together in good faith to agree upon the fair market value of the Termination Assets, taking into consideration the remaining term of the Lease. If after thirty (30) days following selection of the appraisers by County and HMA a dispute continues, then in that event, both

-41-

appraisers shall select a third appraiser meeting the criteria described above. Once the third appraiser has been selected, then, as soon thereafter as practicable but within thirty (30) days, the third appraiser shall make its determination as to which of the estimates received from the original appraisers most closely reflects the fair market value of the Termination Assets. The determination by the third appraiser shall be rendered in writing to both the County and HMA and shall be final and binding upon them. The parties shall share equally in the cost of the third appraiser.

- (5) Within thirty (30) days following the final determination of the Lease Asset Value, HMA and County will enter into an agreement setting forth the terms and conditions of the sale of the Termination Assets for the Lease Asset Value. The parties' efforts in that regard will be pursued diligently and in good faith, and the agreement entered into by County and HMA will contain only those terms and conditions that would be customary to consummate the transaction in accordance with the provisions hereof, including those with respect to any required governmental approvals, transfers of Licenses and Permits and contracts, and other provisions necessary to ensure that County receives in the Termination Assets those assets required to operate the Hospital Facilities as acute care, in-patient hospital facilities.
- (6) In evaluating the third party seeking to purchase, merge or consolidate with HMA, the County shall in each and every instance determine if the third party satisfies the conditions described in Section 12.7(g) hereof, and if it does, the County shall approve the third party and the transaction so contemplated.

-42-

- (7) In the event that the County fails to close under the purchase agreement described above, then in that event HMA may complete the Disposition transaction with the Acquiror.
- (8) Sub and HMA will afford to County and its counsel, accountants and other representatives reasonable access to the books and records of Sub and HMA relevant to the operation of the Hospital Facilities, and promptly will furnish County with all information as County may reasonably request in connection therewith.

of the Governing Board.

- (k) Members of the Governing Board. For the term of the Lease, Sub
  -----and HMA, jointly and severally, shall provide directors and
  officers liability insurance for the benefit of members of the
  Governing Board in accordance with policies and procedures
  generally applicable to other hospital facilities owned by HMA.
- (1) Contribution to Foundation. During each of the ten (10)

consecutive calendar years from the Closing Date, HMA will contribute to Hernando Healthcare Foundation, Inc., a Florida not-for-profit corporation (the "Foundation"), Two Hundred Thousand Dollars (\$200,000.00) (for an aggregate of Two Million Dollars (\$2,000,000.00) during such ten calendar years) to be held, invested, applied or distributed in accordance with the corporate purposes of the Foundation as the Board of Directors of the Foundation, in its sole discretion, may determine. Sub and HMA acknowledge and agree that neither Sub nor HMA shall be entitled to control how such contributions, or any portion thereof, are held, invested, applied or distributed by the Foundation. The first installment of Two Hundred Thousand and 00/100 Dollars (\$200,000.00) shall be due at the Closing Date and each subsequent installment shall be due on the respective anniversary date of the Closing Date for each and every year thereafter as long as the Foundation is a bona fide charitable organization under the rules and regulations of the United States Internal Revenue Code, as amended from time to time.

-43-

#### ARTICLE 13. IN GENERAL

13.1 Enforcement Expenses. Except as otherwise expressly provided herein,

in the event any party elects to incur legal expenses to enforce, defend or interpret any provision of this Agreement as between it and the other party, the prevailing party will be entitled to recover from the non-prevailing party (or non-prevailing parties, jointly and severally) the amount of such legal expenses, including attorneys' fees, costs and necessary disbursements, in addition to any other relief to which such party will be entitled.

13.2 Notice. Any notice, demand or communication required, permitted or

desired to be given hereunder must be in writing and will be deemed effectively delivered when personally delivered, when actually received by recognized overnight courier, or five days after being deposited in the United States mail, with postage prepaid thereon, by certified or registered mail, return receipt requested, addressed as follows:

if to Regional: Regional Healthcare, Inc.

14540 Cortez Boulevard Brooksville, FL 34613 Attention: Chairman

with a copy to: Carlton, Fields, Ward, Emmanuel,

Smith & Cutler, PA One Harbour Place

777 South Harbour Island Boulevard

Tampa, Florida 33602

Attention: Nathaniel L. Doliner, Esq.

if to Sub or to HMA:

Health Management Associates, Inc. 5811 Pelican Bay Boulevard, Suite 500

Naples, FL 34108-2710

Attention: William J. Schoen

Chairman and Chief Executive

Officer

with a copy to:

Health Management Associates, Inc. 5811 Pelican Bay Boulevard, Suite 500

Naples, FL 34108-2710

Attention: Timothy R. Parry, Esq.

Vice President and General

Counsel

or to such other address, or to the attention of such other person, as any party may designate by notice delivered in like manner.

-44-

13.3 Schedules and Other Instruments. Each Schedule, provided hereunder

and each written disclosure required hereby is incorporated by reference into this Agreement and will be considered a part hereof as if set forth herein in full.

- 13.5 Benefit. This Agreement will inure to the benefit of and be binding ----upon the parties hereto, Hernando County, Florida, and their respective legal

representatives, successors and permitted assigns. This Agreement is not intended to benefit any other individual, group, association or entity.

13.6 Waivers and Consents. Any waiver of any provision of this Agreement

and any consent given hereunder must be in writing signed by the party sought to be bound. The waiver by any party of a breach or violation of any provision of Agreement will not operate as, or be construed to constitute, a waiver of any subsequent breach or violation of the same or any other provision hereof. No delay or failure on the part of any party in exercising or enforcing any right, power or privilege hereunder will operate as a waiver thereof nor will any single or partial exercise of any right, power or privilege preclude any other or further exercise of any other right, power or privilege.

- 13.7 Severability. In the event any provision of this Agreement is held to
- be invalid, illegal or unenforceable for any reason and in any respect, such invalidity, illegality or unenforceability will in no event affect, prejudice or disturb the validity of the remainder of this Agreement, which will be and remain in full force and effect, and enforceable in accordance with its terms.
  - 13.8 Inferences. Inasmuch as this Agreement is the result of negotiations

between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of or against any party will be drawn from the fact that any portion of this Agreement has been drafted by or on behalf of such party.

13.9 Entire Agreement. This Agreement supersedes all previous agreements

and constitutes the entire agreement of whatsoever kind or nature existing among the parties representing the within subject matter, and no party will be entitled to benefits other than those specified herein and therein. The parties specifically acknowledge that in entering into and executing this Agreement, the parties rely solely upon the representations and agreements contained herein and therein and no others. All prior representations or agreements, whether written or oral, not expressly referenced herein are superseded.

-45-

- 13.10 Amendment. This Agreement may be amended, and the terms hereof may be -----modified, only by a writing executed by each of the parties hereto, and any matter referred to herein as mutually agreed to or designated by the parties must be evidenced by such a writing.

- agree that the Bankruptcy Court shall retain exclusive jurisdiction with respect to this Agreement and shall resolve all disputes with respect to this Agreement. Any such resolution by the Bankruptcy Court shall be final and binding on the parties.

13.13 Bankruptcy Court Jurisdiction. The parties to this Agreement hereby

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their authorized officers, all as of the date and year first above written.

"REGIONAL"

Regional Healthcare, Inc.

By: /s/ Betty Escamilla

Title: Chairman

Hernando Healthcare, Inc.

By:/s/ Betty Escamilla

Title: Chairman

Spring Hill Regional Hospital, Inc.

By:/s/ Betty Escamilla

Title: Chairman

"SUB"

Hernando HMA, Inc.

By: /s/ Earl P. Holland

\_\_\_\_\_

Title: Vice Chairman

\_\_\_\_\_

"HMA"

Health Management Associates, Inc.

By: /s/ Earl P. Holland

\_\_\_\_\_

By: Vice Chairman

\_\_\_\_\_

-47-

### AMENDMENT TO DEFINITIVE AGREEMENT

That certain Definitive Agreement dated March 12, 1998, (the "Definitive Agreement") by and among REGIONAL HEALTHCARE, INC., a Florida not-for-profit corporation ("RHI"), HERNANDO HEALTHCARE, INC., a Florida not-for-profit corporation ("HHI") and SPRING HILL REGIONAL HOSPITAL, INC. a Florida not-for-profit corporation ("Spring Hill") (hereinafter RHI, HHI and Spring Hill are collectively called "Regional"), and HERNANDO HMA, INC., a Florida corporation ("Sub"), and HEALTH MANAGEMENT ASSOCIATES, INC., a Delaware corporation ("HMA") is hereby amended as follows:

The Lease Agreement attached to the Definitive Agreement as Exhibit "A" is hereby deleted in its entirety and the Lease Agreement attached hereto as Exhibit "A-1" is substituted in lieu thereof (the "Lease Agreement").

- 1. Section 2.5 is hereby deleted in its entirety and the following Section 2.5 is substituted in lieu thereof:
  - 2.5 Purchase Price; Rent. As consideration for the Purchased

Assets, Sub will pay (a) to Regional a purchase price for the Purchased Assets in the amount of SEVENTY-FIVE MILLION FIVE HUNDRED THOUSAND AND 00/100 DOLLARS (\$75,500,000.00) minus (i) an amount equal to fifty percent (50%) of the accrued sick leave, (ii) one hundred percent (100%) of the accrued vacation and holiday pay of Regional's employees that are hired by Sub as of the Closing Date, and (iii) one hundred percent (100%) of all of the assumed accounts payable and accrued expenses of Regional as of the Effective Time that exceed the sum of One Million One Hundred Thousand and 00/100 Dollars (\$1,100,000.00) (the "Purchase Price"), and (b) to County, as rent for the Leased Assets the amount of THREE HUNDRED THOUSAND AND 00/100 DOLLARS (\$300,000.00) per annum (the "Rent") commencing on the Closing Date and on each anniversary thereof during the Lease Term, in accordance with the terms and conditions set forth in the Lease Agreement. On or prior to the Closing Date, Regional will deliver to Sub Regional's estimate of the Purchase Price including, but not limited to, the items described in Section 2.5(a)(i), (ii) and (iii) in accordance with Regional's methods of determining such items (the "Estimated Purchase Price"). At the Closing, Sub will pay to Regional, and HMA will cause Sub to pay to Regional, by wire transfer of immediately

available funds, the Estimated Purchase Price and Sub shall pay to County, and HMA will cause Sub to pay to County the first annual payment of Rent. The Estimated Purchase Price will be subject to a post-Closing adjustment as provided by Section 2.6.

- 2. Sections 3.1 and 11.1 are hereby modified by deleting the references to May 15, 1998 and in lieu thereof inserting in each such instance May 31, 1998.
- 3. The caption of Article 10 is deleted in its entirety and the following substituted in lieu thereof:

Article 10. Indemnification; Cooperation.

4. Section 10.2 is deleted in its entirety and the following Section 10.2 is substituted in lieu thereof:

Section 10.2 Cooperation by Regional. From and after the Closing Date,

Regional will provide assistance and will cooperate with Sub and HMA in attempting to cure any breach of any representation, warranty, covenant or agreement of Regional in this Agreement, under the following conditions:

- (i) notice of breach of warranty is given to Regional by Sub within 12 months after the Closing Date;
- (ii) Regional shall not have any financial liability to Sub and/or HMA or be required to expend or incur any sums to cure any breach of a representation, warranty, covenant or agreement of Regional in this Agreement. Sub's and HMA's sole remedy against Regional after the Closing with respect to the breach of any representation, warranty, covenant or agreement shall be for specific performance to compel Regional to perform its obligation of assistance and cooperation described in this Section 10.2;
- (iii) Regional shall have no obligation to indemnify or hold harmless Sub or HMA under this Agreement by virtue of Regional's breach of the representations, warranties, covenants and agreements as described above; and
- (iv) Regional shall not be liable for any damages suffered by Sub or HMA with respect to any breach of representation, warranty, covenant or agreement of Regional as described above.
- 5. Section 10.4(b) is hereby deleted and the following Section 10.4(b) is substituted in lieu thereof:
  - (b) Any recovery by Regional, and its directors, trustees,

-49-

officers and Affiliates (for purposes of this Section 10.4(b), the "Indemnitees") for indemnification shall be limited as follows:

- (i) Regional shall not be entitled to any recovery unless a claim for indemnification is made (A) in accordance with Section 10.3, and (B) within the time period of survival set forth in Section 10.3,
- (ii) Indemnitees shall not be entitled to recover any Losses

unless and until the amount which all Indemnitees are entitled to recover in respect of such Losses exceeds, in the aggregate, \$100,000 (the "Deductible"), in which event, Indemnitees shall be entitled to recover in respect to such Losses only the amount of the Losses that exceed the Deductible. No Losses shall be included in determining whether the Deductible has been reached unless a notice seeking indemnification for such Losses has been given by the Indemnitee to the Indemnitor within the time frames and in accordance with the terms set forth in Article 10 hereof.

- 6. Section  $10.4\,(c)$  is hereby deleted and the following Section  $10.4\,(c)$  is substituted in lieu thereof:
  - (c) HMA and Sub shall not be entitled to any financial recovery from Regional in the event of a claim submitted pursuant to the indemnity provisions set forth in this Article 10; provided, however, as applicable, they may seek solely the enforcement of Regional's obligation under Section 10.2.
- 7. Section 12.7(a)(viii) is hereby deleted and the following Section 12.7(a)(viii) is substituted in lieu thereof:
  - (viii) Review of operating information provided by Sub regarding performance of the Hospital Facilities for the period from the prior meeting to the then current

-50-

meeting of the Governing Board. In addition to financial statements and operating statistics for the Hospital for the aforesaid period, the Governing Board shall also be provided with the results of an annual financial audit of the Hospitals as certified by an independent public accounting firm. The County and the Governing Board shall also receive the results of the aforesaid financial audit, copies of the annually released corporate information of HMA, and a copy of the asset inventory prepared in connection with the annual financial audit of the Hospitals.

- 8. Section 12.7(c) is hereby deleted and the following Section 12.7(c) is substituted in lieu thereof:
  - 12.7 (c) Indigent Care. Sub shall treat indigent patients as required

under applicable Florida law. Sub acknowledges that on the Commencement Date, it shall continue the same indigent care program that was previously adopted by Regional for the Hospital Facilities, a copy of which is attached hereto as Schedule 12.7(c) (the "Indigent Care Policy"), and that it shall in the future provide, at the minimum, indigent care to the extent as specified in the Indigent Care Policy and as may be specifically required by the provisions of Chapter 87-92, Laws of Florida, and Florida Statutes, Section 155.40, and any other applicable provision of federal or state law. Sub acknowledges that any patient presented to the emergency room who has a medical emergency or who, in the judgment of a staff physician, has an immediate emergency need will be treated in accordance with such policies. No such patient will be turned away because of age, race, gender or inability to pay. Sub will ensure that the Hospitals will continue to provide medical services to patients covered by the Medicare and Medicaid programs.

9. Section 12.7(g) is hereby deleted in its entirety and the following

12.7(g) Assignment or Sublease. Sub shall have the right to assign its

interest in the Lease or to sublease all or any portion of the Leased Premises or its real estate interest in the Leased Premises or HMA or its Affiliates shall have the right to sell or transfer fifty percent (50%) or more of the voting stock of Sub or enter into any transaction changing the control of Sub (whether in a single transaction or a series of transactions) only with the prior written consent of the County. Notwithstanding the above, Sub may sublease portions of the Leased Premises, i.e., office space, examination rooms, etc., to third parties in the normal course of business without the consent of County.

-51-

- 10. Section 12.7(h) is hereby deleted in its entirety and the following Section 12.7(h) is substituted in lieu thereof:
  - 12.7(h) Competing Properties Acquired by Business Combination. Any sale,

purchase, merger, other reorganization, partnership, affiliation, or other transaction (in any case, a "Business Combination") pursuant to which HMA (or any of its Affiliates) or any corporation or other entity surviving or resulting from such Business Combination (or any of its Affiliates) will own or operate or have any interest in the ownership or operation of any other hospital in Hernando County, Florida, and/or Bayonet Point Hospital, in Pasco County, Florida (the "Other Hospitals") must be approved by the County; provided, however, as to Bayonet Point Hospital, said approval shall not be unreasonably withheld; provided further, however, that the County shall be deemed to have consented to such Business Combination if HMA (and all of its Affiliates) and any other company surviving such Business Combination (and all of its Affiliates) have, within twelve (12) months after the date on which such Business Combination becomes effective, divested themselves of their entire right, title and interest in the Other Hospitals, whether held directly or indirectly, and any interest, whether direct or indirect, in the management or operation of the Other Hospitals.

- 11. Section 12.7(i)(2) is hereby deleted in its entirety and the following Section 12.7(i)(2) is substituted in lieu thereof:
- 12.7(i)(2) In the event the County disapproves the Disposition as described in Section 12.7(i)(1)(b), the County shall elect one of the following options: (i) the County may purchase the Leased Premises, New Improvements and all tangible and intangible personal property used therein, other than the Excluded Assets, of Sub that would be transferred to the County at the time of termination of the Lease (the "Termination Assets"), within one hundred and eighty (180) days from the date the County disapproves the Disposition, upon payment of the fair market value for the Termination Assets as hereinafter described (the "Lease Asset Value"), or (ii) the County will, within one hundred and eighty (180) days from the date the County disapproves the Disposition, find a purchaser acceptable to the County that will purchase and close on the Termination Assets as provided for herein. If the County does not exercise its rights under Section 12.7(i)(2)(i) or Section 12.7(i)(2)(ii) within the time frame set forth herein, the County shall be deemed to have consented to the Disposition and to the transaction by and between the Acquiror (as herein defined) and HMA.

- 12. Section 12.7(i)(4) is hereby deleted in its entirety and the following Section 12.7(i)(4) is substituted in lieu thereof:
  - 12.7(i)(4) The Disposition Notice shall also set forth the Lease Asset Value for the Termination Assets. The County shall, within ninety (90) days after receipt of the Disposition Notice, either accept or reject the Disposition. In addition, if the County disagrees with HMA's determination of the Lease Asset Value, then the County and HMA shall each select an MAI commercial real estate appraiser, with at least 10 years' experience in appraising hospital facilities, in the southeast region of the United States. Upon selection, the County's and HMA's appraisers will work together in good faith to agree upon the fair market value of the Termination Assets, taking into consideration the remaining term of the Lease, and the business operated therein. If after thirty (30) days of selection of the appraisers by the County and HMA a dispute continues, then in that event, both appraisers shall select a third appraiser meeting the criteria described above. Once the third appraiser has been selected, then, as soon thereafter as practicable but within thirty (30) days, the third appraiser shall make its determination as to which of the estimates received from the original appraisers most closely reflects the fair market value of the Termination Assets. The determination by the third appraiser shall be rendered in writing to both the County and HMA and shall be final and binding upon them. The parties shall share equally in the cost of the third appraiser.
- 13. Section 12.7(i)(7) is hereby deleted in its entirety and the following Section 12.7(i)(7) is substituted in lieu thereof:

-53-

- 12.7(i)(7) In the event that the County does not agree with the final determination of the Lease Asset Value or fails to close under the purchase agreement described above, then in that event HMA may complete the Disposition transaction with the Acquiror. In such event, the Acquiror shall take the Termination Assets subject to all terms set forth in this Lease Agreement.
- 14. Section 12.7(1) is hereby deleted in its entirety, and the following Section 12.7(1) is substituted in lieu thereof:
  - 12.7(1) Contribution to Foundation. On or before the Closing Date, Regional

shall establish Hernando Healthcare Foundation, Inc., a Florida notfor-profit corporation (the "Foundation"). Sub shall fund to Regional at the Closing Date a sum equal to \$1,216,624.00, (the "Discount Amount") which the parties hereto agree is the present value of \$1,800,000.00 payable over 10 years in equal annual payments applying a 10% discount rate. The Discount Amount may be used by Regional, in its sole discretion, for the purpose of paying or discharging or making adequate provisions for the payment or discharge of Regional's liabilities and obligations, and satisfying costs, expenses and claims associated with the Closing and in the processing of reimbursement claims pursuant to Medicaid and Medicare guidelines. The Discount Amount not funded by Regional for such purposes shall be disbursed to the Foundation. Upon receipt by Regional of Medicare or Medicaid reimbursement sums, pursuant to Section 12.1 of this Agreement (the "Reimbursements"), Regional shall contribute such portion of the Reimbursements after Regional

has paid or discharged or made adequate provisions for the payment and discharge of Regional's liabilities and obligations, until such time as the Foundation is in receipt of the aggregate sum of the Discount Amount (including any prior cash contributions of Regional to Foundation). The sums distributed to the Foundation shall be held, invested, applied and distributed in accordance with the corporate purposes of the Foundation, as such purposes as may exist from time to time, as the Board of Directors of the Foundation, in its sole discretion, may determine. Sub and HMA acknowledge and agree that neither Sub nor HMA shall be entitled to control how such contributions, or any portion thereof, are held, invested, applied or distributed by the Foundation.

- 15. The Table of Exhibits and Schedules attached to the Definitive Agreement is hereby deleted in its entirety and the Revised Table of Exhibits and Schedules attached hereto is substituted in lieu thereof.
  - 16. In the event of any inconsistencies between any specific

-54-

provision in the Definitive Agreement and any specific provision in the Lease Agreement, the provisions of the Lease Agreement shall control.

17. All other terms and conditions of the Definitive Agreement not modified hereby remain in full force and effect and are hereby ratified and confirmed.

IN WITNESS WHEREOF, the parties have caused this Amendment to the Definitive Agreement to be executed by their authorized officers, all as of the date and year first above written.

"REGIONAL"

REGIONAL HEALTHCARE, INC., a Florida not-for profit corporation  $% \left( 1\right) =\left( 1\right) +\left( 1\right) +\left($ 

By:			 	
Title:				

-55-

HERNANDO HEALTHCARE, INC., a Florida not-for profit corporation

By: /s/ Betty Escamilla
----Title: Chairman

SPRING HILL REGIONAL HOSPITAL, INC., a Florida notfor profit corporation  $% \left( 1\right) =\left( 1\right) +\left( 1\right) +\left$ 

By: /s/ Betty Escamilla
Title: Chairman

"SUB"

HERNANDO HMA, INC., a Florida corporation

By: /s/ Earl P. Holland

\_\_\_\_\_\_

Title: Vice Chairman

\_\_\_\_\_

"HMA"

By:/s/ Earl P. Holland

\_\_\_\_\_

Title: Vice Chairman

-----

-56-

### LEASE AGREEMENT

### by and between

# HERNANDO COUNTY, FLORIDA a political subdivision of the State of Florida ("County")

and

HERNANDO HMA, INC.

# LEASE AGREEMENT

### TABLE OF CONTENTS

<table> <caption></caption></table>			
			PAGE
<s></s>	<c></c>		
ARTICLE I			<c></c>
		RECITALS, DEFINITIONS AND DEMISE	. 2
	1.1	RECITALS	
	1.2	DEFINITIONS	. 2
	1.3	DEMISE OF THE LEASED PREMISES	. 12
ARTICLE II			
		REPRESENTATIONS	-
	2.1	REPRESENTATIONS OF LESSOR	. 12
	2.2	REPRESENTATIONS OF LESSEE	. 14
ARTICLE III			
		LEASE TERM AND RENTAL PROVISIONS	. 15
	3.1	LEASE TERM	. 15
	3.2	EXPIRATION OR TERMINATION OF LEASE	. 15
	3.3	RENT	. 22
ARTICLE IV			
		OPERATION OF HOSPITAL FACILITIES	. 22
	4.1	OPERATION OF HOSPITAL FACILITIES	. 22

		TAXES AND UTILITIES 30	
	5.1	TAXES 30	
	5.2	UTILITIES	
ARTICLE VI		INSURANCE	
	6.1	INSURANCE	
	6.2	BLANKET POLICIES	
	6.3	INCREASE IN COVERAGE	
	6.4	QUALIFIED INSURER AND STANDARD PROVISIONS 35	
	6.5	~	
	6.6	INSURANCE PROCEEDS	
ARTICLE VII			
		MAINTENANCE AND REPAIR	
		ALTERATIONS AND IMPROVEMENTS 36	
	7.1	CONDITION OF HOSPITAL FACILITIES 36	
	7.2	REPAIRS	
	7.3	ALTERATIONS AND IMPROVEMENTS	
	7.4 7.5	REMOVAL OR DISPOSITION OF PROPERTY	
	7.5 7.6		

 7.0 | CONSTRUCTION LIENS |  || (/ 17101111/ |  |  |  |
		-i-	
ARTICLE VIII			
	0 1	ASSIGNMENTS, SUBLEASES AND ENCUMBRANCES	40
	8.1		
		COMBINATIONS, AND RIGHT OF FIRST REFUSAL TO LESSOR	40
	8.2		40
	0.2	ENCUMBRANCES OF LEASED PREMISES BY LESSEE	44
ARTICLE IX			
		ADDITIONAL COVENANTS OF LESSEE	
	9.1	PROTECTION OF THE LEASED PREMISES	45
ADMICIE V			
ARTICLE X		EVENTS OF DEFAULT AND REMEDIES	45
	10 1	EVENTS OF DEFAULT DEFINED	
		REMEDIES ON DEFAULT	
		ADVANCES BY LESSOR	
	AGREEMENT TO PAY ATTORNEYS' FEES AND EXPENSES		
ARTICLE V

	10.5 WAIVER 4	łδ
ARTICLE XI		
ANTICLE AT	11.2 COMPLETION OF REMEDIAL WORK	49 50 51
ARTICLE XII	CONDEMNATION	- 1
	12.1 TAKING	51 51 52
ARTICLE XIII		
	ASSUMPTION OF RISK	
ARTICLE XIV		
	INDEMNITY	
ARTICLE XV		
	15.2 COUNTERPARTS.  15.3 BINDING EFFECT.  15.4 SEVERABILITY.  15.5 ARTICLE AND SECTION CAPTIONS.  15.6 ENTIRE AGREEMENT.  55.7 GUARANTY AGREEMENT.  55.7 SUARANTY AGREEMENT.  55.7 SUARANTY AGREEMENT.  56.7 SUARANTY AGREEMENT.  57.7 SUARANTY AGREEMENT.	55 55 56 56 57

			-ii-	
<\$>	15.10 LESSOR'S ACCESS TO LEASED PREMISES	2> 58 58 51 51		

	15.18 INDUCEMENT	62 62 63 63

Exhibit "A"	Composite Legal Description - Leased Premises			
Exhibit "B"	Indigent Care Policy			
Exhibit "C"	Asbestos			
Exhibit "D"

Exhibit "E"

Guaranty

Memorandum of Lease

-iii-

## LEASE AGREEMENT

THIS LEASE AGREEMENT is made as of the 1st day of June, 1998, by and between HERNANDO COUNTY, FLORIDA, a political subdivision of the State of Florida ("Lessor"), acting through its Board of County Commissioners, and HERNANDO HMA, INC., a Florida for profit corporation ("Lessee").

# RECITALS

WHEREAS, Lessor has previously leased certain property in Hernando County, Florida, to Regional Healthcare, Inc., a Florida not-for-profit corporation ("RHI"), and its predecessors for operation of Hospital Facilities (as hereinafter defined) to benefit the residents of Hernando County and adjacent areas;

WHEREAS, on or about February 26, 1993, RHI and its affiliates, Spring Hill Regional Hospital, Inc. ("Spring Hill"), Hernando Healthcare Inc., ("HHI") and Healthcare Realty Corporation ("HRC") filed Chapter 11 petitions with the United States Bankruptcy Court, Middle District of Florida, Tampa Division (the "Bankruptcy Court");

WHEREAS, the Debtors filed a Plan of Arrangement in the Bankruptcy Court which contemplated the sale and/or lease of the Hospital Facilities to a third party and subsequently entered into negotiations with the Lessee pertaining to such transaction;

WHEREAS, pursuant to the provisions of that certain Definitive Agreement dated as of the 12th day of March, 1998, by and among RHI, Spring Hill, HHI, and Lessee, as amended by Amendment to Definitive Agreement dated May 18, 1998

(together the "Agreement"), Lessee has agreed to lease the Leased Premises (as hereinafter defined)

- 1 -

subsequent to the approval by the Bankruptcy Court of the Agreement and this Lease Agreement.

WHEREAS, Lessor has agreed to lease the Leased Premises owned by Lessor to Lessee upon, and subject to, the terms set forth herein, subsequent to the approval by the Bankruptcy Court of the Agreement and this Lease Agreement and at such time as delivery of the Lease Agreement is required under the terms of the Agreement;

WHEREAS, Lessee has agreed to continue to operate the Leased Premises as Hospital Facilities and to establish a Governing Board (as herein defined) that would have certain advisory as well as specific responsibilities, all as hereinafter set forth, in the oversight of the operation of the Hospital Facilities; and

WHEREAS, pursuant to the provisions of Florida Statutes, Section 155.40, Lessor has the authority to enter into this Lease Agreement.

### AGREEMENTS

\_\_\_\_\_

NOW, THEREFORE, in consideration of the foregoing, and the mutual covenants and agreements hereinafter contained, Lessor and Lessee hereby agree as follows:

### ARTICLE I

### RECITALS, DEFINITIONS AND DEMISE

### SECTION 1.1 RECITALS

The recitals stated above are true and correct and are incorporated into this Lease by this reference.

## SECTION 1.2 DEFINITIONS

The following terms, when used in this Lease, shall have the following meaning:

- 2 -

A. "Affiliate" means an entity which, directly or indirectly, controls, is controlled by, or is under common control with, the referenced party.

- B. "Brooksville Regional Hospital" means the hospital facilities known as Brooksville Regional Hospital located on the land described for such facility in Composite Exhibit "A" attached hereto and incorporated herein, together with all improvements, additions and fixtures installed, constructed or placed thereon, and all appurtenances and hereditaments pertaining thereto, whether now existing or whether installed, placed or constructed thereon after the date of this Lease, less any parts thereof which may be taken by exercise of the power of eminent domain.
  - C. "Commencement Date" means as of June 1, 1998.
  - D. "Debtors" mean RHI, Spring Hill and HHI.
- E. "Environmental Condition" means any noncompliance on or about the Leased Premises with any Environmental Law caused by Lessee's Operations on the Leased Premises.
- F. "Environmental Law" means any and all federal, state, local and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or other governmental restrictions relating to the environment or to the Handling (as hereinafter defined), emissions, discharges, releases or threatened releases of pollutants, contaminants, chemicals, or industrial, toxic or hazardous substances, materials or wastes, including without limitation petroleum products, into the environment including, without limitation, ambient air, surface water, ground water, or land.

- 3 -

- G. "Environmental Notice" means any written notice or report of any the following:
  - 1) Any suit, proceeding, investigation, order, consent order, injunction, writ, award, or action related to or affecting the Handling of any Waste (as hereinafter defined) on or about the Leased Premises relating to Lessee's Operations (as hereinafter defined) on the Leased Premises;
  - 2) Any Spill or Environmental Condition on or about the Leased Premises relating to Lessee's Operations on the Leased Premises;
  - 3) Any dispute relating to the Handling of any Waste, Spill or Environmental Condition on or about the Leased Premises relating to Lessee's Operations on the Leased Premises;
  - 4) Any claims by or against any insurer related to or arising out of any Waste, Spill or Environmental Condition on or about the Leased Premises relating to Lessee's Operations on the Leased Premises;
  - 5) Any recommendations or requirements of any governmental or regulatory authority, or insurer relating to any Handling of Waste, Spill,

or Environmental Condition on or about the Leased Premises, relating to Lessee's Operations on the Leased Premises;

6) Any legal requirement or deficiency related to the Handling of Waste, Spill or Environmental Condition on or about the Leased Premises relating to Lessee's Operations on the Leased Premises; or

- 4 -

- 7) Any notice given to a tenant, concessionaire, manager, or other party or entity occupying or using the Leased Premises, or any part thereof which has engaged in or engages in the Handling of any Waste on or about the Leased Premises during the period of Lessee's Operations.
- H. "Excluded Assets" means, collectively:
- 1) Cash, cash deposits and escrows, and all other cash equivalent items of Lessee, including all investments of Lessee;
- 2) Lessee's corporate and fiscal records and other records pertaining to the operation of the Hospital Facilities by Lessee which Lessee is required by law to retain in its possession to the extent that such records are not necessary for the operation of the Hospital Facilities by Lessor; and
- 3) All refunds and reimbursements for periods within the Lease Term, even if payable after the expiration of the Lease Term, available from insurors, third party payors, Medicaid and Medicare under applicable rules and regulations and other comparable programs;
- 4) All accounts receivable that exist and are available to the Lessee for periods on or prior to the expiration or termination of the Lease; and
- 5) All notes payable held by Lessee as of the date of the expiration or termination of the Lease, including notes signed by physicians.
- I. "Governing Board" means the local governing board of the Hospital Facilities, initially consisting of the present members of the Board of Directors of Debtors and the Chief Executive Officer and Operations Vice President of Lessee.

- 5 -

- J. "Guarantor" means Health Management Associates, Inc. a Delaware corporation.
- K. "Handling" means the use, treatment, storage, manufacture, processing, distribution, transport, placement, handling, discharge, generation, production or disposal.

- L. "Hospital Facilities" means (i) Brooksville Regional Hospital and Spring Hill Regional Hospital and any new hospital developed on the Leased Premises by Lessee that replaces either Hospital (singularly, hereinafter "Hospital"), (ii) medical offices and administrative facilities appurtenant to Brooksville Regional Hospital and Spring Hill Regional Hospital, (iii) all ancillary facilities pertaining to the operation of each of the Hospitals, and (iv) all fixtures in the Hospital and the ancillary, office and administrative facilities, all of which are located on the Leased Premises that were transferred by the Debtors to Lessor immediately prior to the Commencement Date of this Lease, whether now existing or whether installed, constructed or placed on the Leased Premises after the date of this Lease, less any portions thereof and interests therein released in writing by Lessor and Lessee from this Lease or taken by the exercise of the power of eminent domain.
- M. "Indemnified Parties" means Lessor and Lessor's successor and assigns, including all employees, commissioners and administrators of the Lessor.
- N. "Indemnity" means the indemnity provisions contained in this Lease in favor of Lessor. Wherever this Lease requires Lessee to indemnify Lessor, such indemnity shall extend to all claims arising in connection with the indemnified matter and shall expressly include all of Lessor's attorneys' fees and costs and consultants' fees.

- 6 -

Upon written request of Lessor, Lessee shall defend and hold Lessor harmless and shall undertake the defense of Lessor, at Lessee's sole expense, in connection with any indemnity set forth in this Lease. Lessee shall have the right to settle and/or compromise any claim that is the subject of Lessee's Indemnity obligations provided for in this Lease. In no event shall Lessor be required or obligated to advance any attorneys' fees and costs or consultant's fees as a condition to enforcement of any indemnity of Lessee under this Lease. In the event Lessee fails or refuses to undertake the defense of Lessor at Lessee's expense or fails to pay, when incurred, the reasonable attorneys' fees and costs and reasonable consultants' fees of Lessor after receiving such request, or if Lessee fails to diligently and continuously conduct such defense after receiving such request as determined by Lessor in its sole discretion, then if Lessor so elects, Lessor may undertake such defense without reducing Lessee's obligation to protect, indemnify and hold harmless Lessor as provided in this Lease, and Lessee shall pay the costs incurred by Lessor in undertaking its own defense, including but not limited to, reasonable attorneys' fees and costs and reasonable consultants' fees. All such fees and costs incurred by Lessor shall constitute a portion of the indemnification duties set forth in this Lease. It is the intent of the parties that Lessee shall pay, address, and satisfy in full all such fees and costs incurred by Lessor from the date of the assertion of any claim through the appellate process at no expense to Lessor.

- 7 -

O. "Judicial Determination" means the final decision of a court of

competent jurisdiction subsequent to expiration of all appeals and/or appeal periods and the posting of a supersedeas bond, if required, in the appeal process.

- P. "Lease" and "Lease Agreement" mean this Lease Agreement and all duly executed amendments thereto.
- Q. "Lease Term" means the period that Lessee has the right to possession under this Lease which shall begin on the Commencement Date and shall expire on June 1, 2028, unless sooner terminated pursuant to the provisions of this Lease.

### R. "Leased Premises" means:

- 1) The real property and facilities pertaining to Brooksville Regional Hospital, Spring Hill Regional Hospital, Brooksville Service Center, Brooksville Enrichment Center, Pinebrook Medical Center and parking lot, all as specifically described in Composite Exhibit "A" attached hereto and incorporated herein, including all buildings, improvements, additions, appurtenances and hereditaments thereto, all of which are used as or constitute a part of the Hospital Facilities.
- 2) A total of one hundred sixty-seven (167) licensed hospital beds comprised of ninety-two (92) beds at Brooksville Regional Hospital and seventy-five (75) beds at Spring Hill Regional Hospital (all of which are appurtenances to Brooksville Regional Hospital) and any expansion or additions thereto together with all improvements, fixtures and appurtenances; and

- 8 -

- 3) All other assets which make up and comprise the Hospital Facilities, including medical offices and administrative facilities appurtenant to Brooksville Regional Hospital and Spring Hill Regional Hospital which were owned or operated by any of the Debtors and transferred to Lessor immediately prior to the Commencement Date of this Lease, (including, without limitation, Brooksville Enrichment Center, Brooksville Service Center, Pinebrook Medical Center, and the Desoto Avenue Parking Lot, all as more particularly described in Composite Exhibit "A") and all fixtures and appurtenances thereto whether now existing or whether installed, constructed or placed on the Leased Premises after the date of this Lease, less any parts thereof and interests therein released in writing from this Lease by Lessor and Lessee or taken by the exercise of the power of eminent domain.
- S. "Lessee" means Hernando HMA, Inc., a Florida for-profit corporation, and its successors and assigns as specifically provided for in this Lease Agreement, including any surviving, resulting or transferee corporation that may be permitted under Section 8.1 of this Lease.
  - T. "Lessor" means Hernando County, a political subdivision of the State

of Florida, and its successors and assigns.

U. "Operations" means Lessee's or any other party's use or occupancy of the Leased Premises from the date Lessee first enters the Leased Premises through the expiration or termination of this Lease (or the date Lessee vacates the Leased Premises, whichever is later).

- 9 -

- V. "Permitted Encumbrances" means those certain title exceptions set forth in Schedule B, Section 2, subsections 6 through 17 of the Title Commitment, and matters shown on the Survey.
- W. "Rental Payment" means all payments due from Lessee to Lessor or otherwise required to be paid by Lessee pursuant to the terms of this Lease.
- X. "Spill" means any major spill, contamination, discharge, leakage, release or escape of any Waste in or affecting the Leased Premises as a result of Lessee's Operations, whether sudden or gradual, accidental or anticipated, or of any other nature or manner that has previously occurred as a result of Lessee's Operations or which may occur as a result of Lessee's Operations.
- Y. "Spring Hill Regional Hospital" means the real property described for such facility in Composite Exhibit "A" attached hereto and incorporated herein, together with all improvements, additions, and fixtures installed, constructed or placed thereon, and all appurtenances and hereditaments thereto, whether now existing or whether installed, placed or constructed thereon after the date of this Lease, less any parts thereof which may be taken by exercise of the power of eminent domain.
- Z. "Survey" means the following ALTA surveys of the Leased Premises prepared by Heidt and Associates, Inc.: Drawing No. 222219-1 dated February 11, 1998, Drawing No. 342218-1 dated February 23, 1998, Drawing No. 342318-1 dated February 23, 1998, and Drawing No. 182220-1 dated February 23, 1998, certified in accordance with generally accepted professional standards, and describing the boundaries,

- 10 -

improvements, setbacks and all matters of record set forth in the Title Commitment.

AA. "Taxes" means all real estate and personal property taxes, including general and special assessments, and all other charges, taxes and assessments, ordinary and extraordinary, foreseen and unforeseen, which are levied, imposed or assessed upon or against this Lease, against the leasehold estate, against the Leased Premises and against all leasehold improvements and all fixtures that may be constructed or installed on the Leased Premises, as well as all sales, use, excise, and all other taxes of any nature whatsoever now or hereafter imposed by any lawful authority on all Rental and/or payments due or required

under this Lease. The term "Taxes" shall also include all fees for licenses and permits, and all fees, charges, taxes and assessments, now or hereafter imposed, foreseen and unforeseen, that may be due, levied or assessed against Lessee or any business conducted on or affiliated with the Leased Premises during the Lease Term.

BB. "Title Commitment" means the Leasehold Title Insurance Commitment dated February 9, 1998, Case No. 9801043, issued by Lawyers Title Insurance Corporation, that will insure as of the commencement of the Lease that the Lessor and Lessee, as their respective interests may appear, have good and marketable title to the Leased Premises, and that the Leased Premises are free and clear of all liens, encumbrances, charges, assessments, taxes, easements, restrictions and stipulations except for items shown on Schedule B - Section 1 of the Title Commitment that will be removed or satisfied at or before Closing and the Permitted Encumbrances.

- 11 -

- CC. "Unwind" means the unwind transaction of this Lease as set forth in Section 3.2(C).
- DD. "Waste" means any contaminant, chemical, petroleum product, waste, waste product, radioactive waste, poly-chlorinated biphenyls, asbestos, hazardous or toxic substance, contaminant, pollutant, substance, or waste of any kind, and any other substance which is regulated by any Environmental Law which is caused by Lessee's Operations on the Leased Premises.

SECTION 1.3 DEMISE OF THE LEASED PREMISES

For and in consideration of the Lessee's timely payment of the Rental Payment and the Lessee's timely performance of the other covenants and agreements of Lessee required under this Lease, Lessor does hereby demise and lease to Lessee, and Lessee does hereby lease, take, and hire from Lessor, the Leased Premises, and all appurtenances and hereditaments relating thereto.

#### ARTICLE II

#### REPRESENTATIONS

SECTION 2.1 REPRESENTATIONS OF LESSOR

Lessor makes the following representations to Lessee:

- A. Lessor is a political subdivision of the State of Florida duly created and existing pursuant to the laws and Constitution of the State of Florida, and has the power to enter into this Lease.
  - B. Lessor is duly authorized to execute and deliver this Lease and

contemporaneous with the signing of this Lease, shall deliver to Lessee a copy of the Certified Resolution of the County Commission authorizing the appropriate representative of the Lessor to sign the Lease and all other documents required by the Lessor to effect the lease of the Leased Premises as provided herein.

- 12 -

- C. This Lease is the legally binding obligation of Lessor and is enforceable against Lessor in accordance with its terms, except as enforceability against Lessor may be affected by its bankruptcy, insolvency, other laws governing creditors' rights generally, and equitable principles.
- D. The leasing of the Leased Premises to Lessee will enable Lessee to provide needed hospital and health care facilities in Hernando County and will promote the health and welfare of the people of Hernando County.
- E. Lessor has good title to the Leased Premises and shall deliver to Lessee such affidavits and certificates as may be reasonably necessary to delete the standard exceptions to the title insurance policy to be issued pursuant to the Title Commitment and shall deliver such instruments to Lessee as are required to be delivered by the Lessor as set forth in Schedule B, Section 1, of the Title Commitment.
- F. Lessor acknowledges that the Leased Premises are in compliance with and in concurrence with all applicable land use and zoning regulations of Hernando County, Florida, that pertain to those portions of the Leased Premises located in the unincorporated areas of Hernando County, Florida, and represents as to those portions of the Leased Premises that are located within any municipality within Hernando County, Florida, that the Lessor is not aware of nor has Lessor received any written notice of noncompliance of the Leased Premises with respect to any zoning code of the applicable municipality.

- 13 -

G. Lessor, without independent investigation, is not aware of any litigation or claims made against the Leased Premises other than claims that were made through the Bankruptcy Court.

### SECTION 2.2 REPRESENTATIONS OF LESSEE

Lessee makes the following representations to Lessor:

- A. Lessee is a duly incorporated for-profit corporation, in good standing under the laws of the State of Florida, and empowered to operate hospital and health care facilities under the laws of the State of Florida.
- B. Lessee is duly authorized to enter into this Lease and to perform its obligations under this Lease.

- C. This Lease is the legally binding obligation of Lessee and is enforceable against Lessee in accordance with its terms, except as enforceability against Lessee may be affected by its bankruptcy, insolvency, other laws governing creditors' rights generally, and equitable principles.
- D. Lessee is in compliance in all material respects with all applicable statutes, rules, regulations and requirements of all applicable governmental entities having jurisdiction over Lessee and its operations, including, without limitation, the Florida Agency for Health Care Administration, and Lessee has timely filed all reports, data and other information required to be filed by it with such governmental entities.

- 14 -

E. Lessee and Guarantor expressly represent that they have the requisite skills, financial ability and management expertise to operate the Hospital Facilities.

#### ARTICLE III

#### LEASE TERM AND RENTAL PROVISIONS

## SECTION 3.1 LEASE TERM

The Lease Term shall begin on the Commencement Date and, subject to the provisions of this Lease, shall continue for a period of thirty (30) years thereafter.

## SECTION 3.2 EXPIRATION OR TERMINATION OF LEASE

Upon the expiration of the Lease Term or earlier termination of this Lease by a Judicial Determination in the event of a default by Lessee hereunder, other than the early Unwind termination as set forth in Section 3.2(C) hereof, Lessee shall for no consideration relinquish and surrender to Lessor possession of the Leased Premises and Lessee shall also convey to Lessor by (i) quit claim deed all of its interest in the Leased Premises, including, without limitation, the Brooksville Regional Hospital, Brooksville Enrichment Center, Brooksville Service Center, Pinebrook Medical Center, and the Desoto Avenue Parking Lot, and any additional structures developed by Lessee on the Leased Premises, and (ii) bill of sale without warranties of all hospital licenses, certificates and permits for the Hospital Facilities, including, but not limited to, the certificates of need and rights to the hospital beds so licensed, and all tangible and intangible personal property including furniture, fixtures, equipment, inventories, and medical records necessary for the Operations that are located on the Leased Premises, other than tangible and intangible property described herein as Excluded Assets, free and clear of any

Taxes (except as to ad valorem taxes, if any, for the balance of the year of closing hereunder), indebtedness, liens or encumbrances, but such conveyances and transfers shall be subject to (a) the Permitted Encumbrances, (b) all restrictions, covenants, reservations, and easements placed of record by Lessee that were approved and joined in by Lessor, (c) ad valorem taxes, if any, for the balance of the year of closing hereunder, and (d) all applicable zoning and land use restrictions. It is the intention of the parties to this Lease Agreement that in the event of the expiration of the Lease Term or earlier termination of this Lease by a Judicial Determination in the event of a default by Lessee under the Lease Agreement (other than the Excluded Assets), that the Leased Premises and all tangible and intangible property necessary for the immediate and continued operation of the Hospital Facilities by Lessor as acute care hospital facilities be conveyed, transferred or assigned, or to the extent not transferable or assignable made available to the Lessor including, but not limited to, all licenses and permits held by the Lessee for the operation of the Hospital Facilities.

B. Upon expiration of the Lease Term or early termination of this Lease by a Judicial Determination in the event of a default by Lessee hereunder or in the event of the "unwind" transaction described in Section 3.2(C), the Lessor shall have the option of requiring Lessee to transfer to Lessor for the consideration herein described all right, title and interest in and to any and all other health care facilities that Lessee may have developed in Hernando County, subsequent to the Commencement Date, for new medical technology or new medical related services or for certain material hospital services transferred from the Hospitals in accordance with Section 4.1(c) that

- 16 -

were not developed on the Leased Premises (the "New Improvements"). The definition of New Improvements shall include all licenses, certificates and permits for the New Improvements, including, but not limited to, certificates of need and all tangible property including the then remaining furniture, fixtures, equipment and other tangible and intangible personal property, including medical records necessary for the operation that is conducted at the New Improvements other than those specific items of tangible and intangible property described herein as Excluded Assets. In consideration for the transfer of the New Improvements to Lessor, the Lessor agrees to pay to Lessee in immediately available funds on the date of transfer of the New Improvements to Lessor an amount equal to the fair market value of the New Improvements and the business operated therein. Prior to the date of transfer of the New Improvements to Lessor, the Lessee shall deliver to Lessor a written notice setting forth the fair market value for the New Improvement. Lessor shall, within ninety (90) days after receipt of the notice, either accept or reject the fair market value determination of Lessee. If Lessor objects, then Lessor and Lessee shall each select an MAI commercial real estate appraiser, with at least ten (10) years experience in appraising hospital facilities, in the State of Florida. Upon selection, Lessor's and Lessee's appraiser shall work together in good faith to agree upon the fair market value. If after thirty (30) days of selection of the

appraisers by Lessor and Lessee a dispute continues, then in that event both appraisers shall select a third appraiser meeting the criteria described above. Once the third appraiser has been selected, then, as soon thereafter as practical but within forty-five (45) days, the third appraiser shall make its determination as

- 17 -

to which of the estimates received from the original appraisers most closely reflects the fair market value of the New Improvements. The determination by the third appraiser shall be rendered in writing to both the Lessor and Lessee and shall be final and binding upon them. The parties shall share equally in the cost of the third appraiser. Upon the final determination of value of the New Improvements, the Lessor shall pay such amount as provided for herein or reject the sum so determined and not purchase the New Improvements.

In the event that Lessee has, in accordance with the terms of this Lease Agreement, transferred to the New Improvements certain material hospital services from the Hospital Facilities pursuant to Section 4.1(C) and the Lessor determines that it will not purchase the New Improvements the Lessee shall prior to the closing contemplated under this Article III restore to the Hospital Facilities, to the greatest extent possible and with all reasonable effort, the material hospital services transferred or relocated to the New Improvements. In addition, if the Lessee has reduced, liquidated or eliminated any hospital services or facilities in accordance with Section 4.1(C) hereof from the Hospital Facilities that are rebuilt or reconstituted on a facility not located on the Leased Premises or a New Improvement, then in that event, the Lessee shall restore such facility to the Hospital Facilities prior to the time of the transfers as contemplated herein.

- 18 -

C. The Lessor shall have the right at any time after the first anniversary of the Commencement Date, but prior to the third anniversary of the Commencement Date, to in good faith "unwind" the lease transaction contemplated hereby, to reacquire Lessee's interest in the Leased Premises, to terminate the Lease and to purchase the New Improvements, all hospital licenses, certificates and permits for the Hospital Facilities and New Improvements, including, but not limited to, the certificates of need and the rights to hospital beds so licensed, and all tangible and intangible property including the then remaining furniture, fixtures, equipment and medical records necessary for Operations that are located on the Leased Premises and New Improvements, other than those specific items of tangible and intangible property described herein as Excluded Assets, by paying to Lessee an amount equal to the product of (i) EBITDA for the Hospital Facilities and New Improvements, multiplied by (ii) seven (7). Sixty (60) days prior to the date of closing of the "unwind" ("Unwind Closing"), Lessee shall present to the Lessor its calculation of EBITDA (EBITDA as used herein is defined as earnings before interest, taxes, depreciation and amortization) accomplished in accordance with generally acceptable accounting principles for the twelve month period immediately preceding the notice of the

value of EBITDA provided Lessor by Lessee sixty (60) days prior to the date of the Unwind Closing. Upon receipt of the notice of value but prior to the date of the Unwind Closing, the Lessor may have its auditors

- 19 -

review the calculation of EBITDA. In the event there is any discrepancy of the calculation of this amount, the Lessor and Lessee shall resolve their differences or, alternatively, Lessee and Lessor jointly will immediately select an independent public accounting firm to resolve such dispute prior to the date of the Unwind Closing, which resolution will be final and binding. In the event Lessor still disagrees with the EBITDA calculation subsequent to such final resolution, the Lessor may not close and in such event the Unwind Closing and the rights provided to Lessor pursuant to this Section 3.2(c) shall be terminated and of no further force and effect. The fees and expenses of such accounting firm will be shared by Lessor and Lessee in proportion to the relative amounts of the dispute and the amount resolved to be determined for the account of Lessor and Lessee, respectively. In order to exercise the unwind option, the Lessor shall give Lessee written notice thereof at least 180 days prior to the date of the Unwind Closing and shall tender the required funds as calculated above to Lessee on the date of the Unwind Closing, and Lessee and Lessor will cooperate reasonably with each other in effecting such "unwind" (including matters of transition), execute such documentation (including conveyance and assignment documents) as is reasonably necessary to effect the "unwind" (including the transfer to the Lessor of the Leased Premises, New Improvements and tangible and intangible personal property, other than the Excluded Assets, necessary for operations that are located on the Leased Premises, New Improvements) and all licenses and permits pertaining to the Leased Premises and New Improvements, free and clear of any indebtedness, liens or encumbrances but subject to the provisions described in Section 3.2(A)(a)-(d). In addition, the parties hereto

- 20 -

agree to make appropriate adjustments post Unwind Closing to reflect a purchase price using an EBITDA for the twelve (12) month period immediately preceding the Unwind Closing.

D. In the event of an "unwind" the Lessor shall and does hereby covenant not to sell or lease the Hospital Facilities to any other party for a period of three (3) years from the date of the Unwind Closing except to a qualified not-for-profit entity satisfying the following requirements: (i) the not-for-profit entity shall be a local entity managed and totally controlled by Hernando County residents, (ii) the local not-for-profit entity shall not be an Affiliate of any other not-for-profit organization, (iii) the local not-for-profit entity shall operate and maintain the Hospital Facilities, (iv) the local not-for-profit entity shall not pay any rent or other remuneration to the County, other than nominal rent or remuneration, for the Hospital Facilities, and (v) the financing for the purchase and operation of the Hospital Facilities as described herein shall be accomplished by County using traditional governmental financing

techniques.

E. Upon expiration of the Lease Term or early termination of the Lease by a Judicial Determination by virtue of a default by Lessee under the Lease, the Lessee hereby agrees that for a period of three (3) years following such termination of the Lease, that Lessee shall not operate any competing hospital facilities (other than New Improvements not purchased by Lessor or other healthcare previously consented to by Lessor) within Hernando County except with the express written consent of the Lessor. This covenant not to compete is specifically not applicable to the termination or expiration of the

- 21 -

Lease Term by virtue of an Unwind Closing that occurs pursuant to the provisions of Section 3.2(B).

F. Contemporaneous with the expiration of the Lease Term or early termination as described above, the Lessee shall provide Lessor with evidence of tail insurance evidencing that Lessee has maintained professional negligence liability insurance with minimum limits of liability of \$1,000,000 for each occurrence and \$3,000,000 in the aggregate, which insurance shall be provided to Lessor to protect Lessor from any claims or liabilities arising from Lessee's operation of the Hospital Facilities.

SECTION 3.3 RENT

The annual rental payment for the Leased Premises for each year of the Lease Term (the "Rental Payment") shall be in the amount of Three Hundred Thousand and 00/100 Dollars (\$300,000.00). This Rental Payment shall be paid to Lessor by Lessee on the Commencement Date and on each anniversary date of the Commencement Date during the Lease Term.

#### ARTICLE IV

#### OPERATION OF HOSPITAL FACILITIES

#### SECTION 4.1 OPERATION OF HOSPITAL FACILITIES

A. Lessee covenants and agrees that during the entire Lease Term, Lessee will continuously operate as going concerns the Hospital Facilities as (i) two general acute care hospitals with programs for the diagnosis, treatment, and care of sick and injured persons, without discrimination on account of race, creed, color, national origin, or sex, consistent with accepted principles of hospital financial management and (ii) ancillary services necessary and proper for the Hospitals, including, without limitation, radiology,

pathology, laboratory, diagnostic imaging services, chemotherapy, inhalation, pulmonary and physical therapy, practice of medicine and health care services, and (iii) administrative offices related thereto as may be customary and proper for a general acute care hospital. Any change in such operation shall be determined by the Governing Board and Lessor as described in Section 4.1(C). Lessee further covenants and agrees that it will not deny emergency hospital care to any person based on ability to pay; provided, however, Lessee shall have the right to refuse to admit patients because of lack of facilities or appropriate personnel or to protect the welfare of patients already admitted and Lessee may adopt and amend from time to time appropriate rules respecting the admission of patients provided, however, that such rules are consistent with the terms of this Lease. Notwithstanding anything to the contrary contained in this Subsection, Lessee shall treat indigent patients as required under applicable Florida law. Lessee acknowledges that on the Commencement Date, it shall continue the same indigent care program that was previously adopted by Debtors for the Hospital Facilities, a copy of which is attached hereto as Exhibit "B" (the "Indigent Care Policy"), and that it shall in the future provide, at the minimum, indigent care to the extent as specified in the Indigent Care Policy and as may be specifically required by the provisions of Chapter 87-92, Laws of Florida, and Florida Statutes, Section 155.40, and any other applicable provision of federal or state law.

- 23 -

Lessee shall establish and maintain, at all times during the Lease Term, a Governing Board for the Hospitals. The Governing Board shall consist of fifteen (15) members initially comprised of the present members of the Board of Directors of the Debtors and the Executive Director and Operations Vice President of the Hospitals. Guarantor may increase the size of the Governing Board at any time and from time to time with an additional two (2) representatives selected by Guarantor. In the event that either or both of the two additional members are members of the medical staff of the Hospitals, the prospective candidates must be approved by the Medical Executive Committee of the candidate's Hospital prior to final selection by Guarantor. The size of the Governing Board, however, may neither be reduced nor increased (except for the two (2) additional representatives that may be selected by Guarantor as described above) without the consent of the Governing Board. during the term of the Lease, not less than the greater of ten (10) or a majority of the members of the Governing Board, will be residents of Hernando County, Florida, who are not officers, employees or agents of Guarantor, or any of its Affiliates or directors of Guarantor or any of its Affiliates. The terms of each member of the Governing Board other than those members selected by Guarantor shall be staggered, with a maximum term for any member of three (3) Each member of the Governing Board may serve for two consecutive terms, and then is required to remain off the Governing Board for one year after the completion of such successive terms. If a Governing Board member dies, voluntarily leaves such position sooner or is removed from such position by a vote of the Governing Board or if a Governing Board member's term expires, or if the size of the Governing Board is

increased beyond seventeen (17) members by the Governing Board, the vacancies and/or additions so created shall be filled by the Governing Board recommending one or more candidates to fill each of such positions to Guarantor for Guarantor's sole approval, which approval shall not be unreasonably withheld. Notwithstanding the preceding sentences, Guarantor may select up to four (4) members of the Governing Board (including the Executive Director and Operation's Vice President of the Hospitals) as described herein, without recommendations from the Governing Board as to such members. A quorum of the Governing Board shall consist of a majority of its members. Members of the Governing Board may be present in person or by means of communications equipment whereby all members of the Governing Board can hear each other simultaneously. Actions by the Governing Board shall be by a majority of a quorum. Governing Board's powers and responsibilities shall include:

- 1) Adopting a hospital vision, mission and value statement, and any amendments thereto, assisting in developing policies and monitoring progress toward strategic goals;
- 2) For the term of the Lease the Governing Board shall have the right to approve the hiring or termination of the Executive Director of the Hospitals, which approval shall not be unreasonably withheld;
  - 3) Approval of any change of the name of any of the Hospitals;

- 25 -

- 4) Participating on an advisory basis in development of and reviewing all operating and capital budgets and facility plans;
- 5) Granting medical staff membership and clinical privileges and, when necessary, taking disciplinary action consistent with credentialing processes, medical staff by-laws and strategic plans;
- 6) Monitoring medical staff and Hospital compliance with the Joint Commission for Accreditation of Healthcare Organizations ("JCAHO") and Health Care Financing Administration's ("HCFA") conditions of participation in the Medicaid and Medicare programs;
  - 7) Fostering community relationships; and
- 8) Review of operating information provided by Lessee regarding performance of the Hospital Facilities for the period from the prior meeting to the then current meeting of the Governing Board. In addition to financial statements and operating statistics for the Hospital for the aforesaid period, the Governing Board shall also be provided with the results of an annual financial audit of the Hospitals as certified by an independent public accounting firm. The Lessor shall also receive the

results of the aforesaid financial audit, copies of the annually released corporate information of Guarantor, and a copy of the asset inventory prepared in connection with the annual financial audit of the Hospitals.

C. During the term of the Lease, Lessee shall operate the Hospital Facilities as full service hospitals substantially in the same manner as currently operated; provided, however that if Lessee

- 26 -

reasonably determines that any material hospital services or facilities should be changed, reduced, liquidated or eliminated, Lessee may change, reduce, liquidate or eliminate such services or facilities only upon the prior approval of the Governing Board and Lessor.

- D. Lessee agrees to provide the same level of care to indigents as provided by the Hospitals immediately prior to the Commencement Date as described in the Agreement. Lessee acknowledges that any patient presented to the emergency room who has a medical emergency or who, in the judgment of a staff physician, has an immediate emergency need will be treated in accordance with such policies. No such patient will be turned away because of age, race, gender or inability to pay. Lessee will ensure that the Hospitals will continue to provide medical services to patients covered by the Medicare and Medicaid programs.
- E. In insuring a modern full-service hospital, Lessee is committed to providing the equipment and capital improvements that are and will be needed by the Hospital Facilities to serve the medical needs of the community through state of the art facilities. Lessee shall fund and pay, and Guarantor shall cause Lessee to fund and pay, for capital expenditures and improvements to the Hospital Facilities to the extent as provided for in Section 12.7(d) of the Agreement, the terms of which are hereby incorporated herein by reference, and, thereafter during the Lease Term, to the extent necessary to maintain the Hospital Facilities as two full-service acute care facilities, to maintain the licenses for the Hospital

- 27 -

Facilities issued by the State of Florida, and to maintain the accreditation of the Hospital Facilities by JCAHO and HCFA.

F. Lessee will offer employment on an "at will" basis, as of the Commencement Date, and at their salaries on the Commencement Date, to all of the employees of Debtors (including senior management) who are active employees of Debtors as of the Commencement Date performing services at the Hospital Facilities. The employees will retain their current seniority, as of the Commencement Date, with regard to vacation and sick time accrual under Lessee's plans. Such employees shall continue to vest in their accrued benefit of Debtor's Pension Plan, as hereinafter defined, determined as of the date of closing based on service with Lessee (the full vesting under the Pension Plan is

five (5) years). In addition, such employees shall also be given credit for any years of service with Debtors for purposes of Lessee's qualified retirement plans with respect to eligibility, participation and vesting. Lessee's retirement plan requires one full year of service to be eligible to participate and seven (7) years to be fully vested. With respect to health insurance coverage, exclusions for pre-existing conditions and applicable waiting periods will be waived. The salaries of the employees at the Commencement Date will provide the base for future merit increases. In the event of an employee reduction, Lessee will pay severance based on position and length of service with Debtors and Lessee in accordance with a severance policy that is applicable generally to all Lessee employees. In addition to a severance payment, eligible employees will also have normal and customary rights to health plan continuation coverage as mandated by Section 4980B of the Code or applicable state health care continuation coverage statutes. Lessee agrees that it will not effect

- 28 -

any reductions in work force for a period of twelve (12) months after the Commencement Date other than reductions through attrition, flexible staffing for seasonal adjustments, or decreases in patient census.

- G. Lessee acknowledges that all of the Debtors obligations and liabilities with respect to the pension plan for the current employees of the Debtors are evidenced by the plan described as Regional Healthcare, Inc. Pension Plan and Trust Agreement as amended and restated, effective as of January 1, 1989 (the "Pension Plan"). Lessee agrees to be the plan sponsor and assume and be responsible for all obligations and liabilities set forth in the Pension Plan. It is understood that the Lessee shall freeze the Pension Plan pursuant to applicable law immediately subsequent to the Commencement Date. Lessee shall service such Pension Plan in order to insure that the benefits are distributed to qualified beneficiaries of the Pension Plan. It is further understood that Lessee shall maintain the Pension Plan in accordance with all applicable laws and shall under no circumstances borrow from such funds. All Pension Plan funds shall be used solely for the benefit of the participants of the Pension Plan and to pay for custodial and other administrative expenses associated with the Pension Plan.
- H. Lessee covenants and agrees that during the entire Lease Term, the Hospital Facilities shall be operated as two full service hospitals providing all essential services necessary to maintain their hospital licenses issued by the Agency for Healthcare Administration, or its successor, and full accreditation by JCAHO, HCFA and any comparable successor accreditation body, as a general acute care hospitals, unless otherwise prescribed by applicable health planning

- 29 -

regulations approved by Lessor. It is, however, understood that on the Commencement Date the Hospital Facilities are subject to conditional accreditation issued by JCAHO. Lessee shall undertake to review, cure and

correct the deficiencies that exist in the Hospital Facilities as to eliminate the conditional accreditation as soon as reasonably possible.

I. Lessee shall comply with all laws, rules, regulations, and requirements of all federal, state, and local governments and agencies and departments thereof which are applicable to it or to licensed hospitals in the State of Florida, including, but not limited to, Florida Statutes, Sections 155.40 and 155.41 and shall at all times have in effect all licenses necessary for the operation of the Hospital Facilities as contemplated by this Lease; provided, however, that nothing herein shall preclude Lessee from challenging in good faith any of such laws, rules, regulations, or requirements.

ARTICLE V

#### TAXES AND UTILITIES

SECTION 5.1 TAXES

Lessee shall pay all Taxes, if any, prior to delinquency. If Lessee desires to contest the validity of any Taxes for which Lessee is responsible, Lessee may do so without being in default under its obligation to pay Taxes, provided Lessee institutes appropriate legal proceedings to contest the validity of the tax or assessment. If assessed, Lessee shall comply with Florida Statutes, Sections 194.171(3) (or shall obtain an injunction, if necessary, or such other provision as may be specified by law) to prevent the sale of any tax certificate or the sale of any property subject to the tax lien by reason of non-payment of the tax or assessment being contested in

- 30 -

such legal proceedings. If Lessee fails to comply with Florida Statutes, Sections 194.171(3) (or to obtain such injunction, or if any injunction so obtained is dissolved), Lessee shall deposit with Lessor at least thirty (30) days before the contested tax or assessment would become delinquent for non-payment (or within ten (10) days of dissolution of the injunction), an amount which is sufficient to pay in full the contested tax or assessment, including, without limitation, all penalties and attorneys' fees and costs if the adjudication in such proceedings should be adverse to Lessee. Lessor shall have the right to make any such payment; provided, however, Lessor shall refund to Lessee any portion of the deposit retained by Lessor which shall be determined by the court not to be due to the taxing authorities on account of such taxes, penalties or attorneys' fees and costs.

### SECTION 5.2 UTILITIES

Lessee shall pay all charges for gas, electric power, water, sewer, and all other utilities and services necessary or desirable for the operation, maintenance, use, and upkeep of the Hospital Facilities and the Leased Premises.

#### ARTICLE VI

#### INSURANCE

## SECTION 6.1 INSURANCE

Lessee covenants that it will obtain and maintain at all times during the Lease Term the insurance for the Hospital Facilities and all tangible personal property used in connection therewith as described in this Article and all other insurance for the Hospital Facilities that Lessee and Lessor reasonably deem necessary or appropriate. Lessee will cause the Hospital Facilities and all tangible personal property used in connection therewith at all times to be reasonably insured against all such risks as are customarily insured against in connection with the operation of hospital facilities of type and size comparable to the Hospital Facilities within the State of Florida in areas comparable to Hernando County. In connection therewith, Lessee will carry and maintain, or cause to be carried and maintained, as minimum requirements, and pay, or cause to be paid, timely the premiums for at least the following insurance when and as such insurance is commercially available:

A. With respect to the Leased Premises including the Hospital Facilities and all tangible personal property used in connection therewith, fire insurance, with uniform standard extended coverage endorsements, and vandalism and malicious mischief insurance with the broadest coverage approved for issuance in the State of Florida, including insurance against loss or damage from lightning, windstorm, hurricane, tornado, hail, explosion, riot, riot attending a strike, civil commotion, aircraft, vehicles, and smoke, in an amount equal to the replacement cost of the Hospital Facilities and all tangible

- 32 -

personal property used in connection therewith, excluding land and foundations;

- B. Commercial general liability, property damage and bodily injury insurance, insuring Lessor and Lessee against any liability whatsoever occasioned by any accident, injury or damage suffered or occurring on or about the Leased Premises or any appurtenance thereto. Lessee shall also procure and maintain during the Lease Term contractual liability insurance coverage for the performance of Lessee's indemnity obligations under this Lease. Each liability policy shall be for the minimum amount of \$1,000,000.00 combined single limit coverage per accident.
- C. Worker's compensation insurance in the minimum amount required by the laws of the State of Florida;
  - D. Comprehensive automobile liability insurance with limits of not less

than \$1,000,000.00 per person and \$1,000,000.00 per occurrence for bodily injury and not less than \$1,000,000.00 per occurrence for property damage;

E. So long as the Medical Malpractice Liability Reform Act, Section 766.105 et seq., Florida Statutes, or any similar law or laws providing insurance to health care providers against liability for death, injury, loss or damage occurring in the examination, diagnosis, treatment, or care of any patient is in effect in the State of Florida (or in the United States if applicable within the State of Florida), a policy or policies of medical liability and malpractice insurance with limits not less than the basic coverage, if any, required under such law or laws, which policy or policies shall be in such form as may be required under such law or laws

- 33 -

(provided that, in lieu of maintaining such policy or policies, the Lessee may comply with any self-insurance requirements respecting qualifications for such insurance if the laws of the State of Florida provide for such qualification by self-insuring in lieu of maintaining such policy or policies) and the Lessee will take any and all other actions required for the Lessee to qualify and to remain qualified for such insurance.

F. Directors and Officers liability insurance for the benefit of Lessee, Guarantor and the members of the Governing Board shall be obtained in accordance with policies, procedures and amounts generally applicable to other hospital facilities owned or operated by Guarantor.

### SECTION 6.2 BLANKET POLICIES

Any property insurance required to be carried under this Article may be included as part of any blanket or other policy or policies of insurance as long as the minimum coverage standards required by this Lease are met or exceeded for the Hospital Facilities independently of coverage for other properties or facilities covered by such policies.

## SECTION 6.3 INCREASE IN COVERAGE

The minimum limits and coverage for liability policies shall be increased from time to time, as reasonably determined by Lessee and Lessor, to assure full protection to Lessor against claims for injury (including death) and damage to property.

- 34 -

## SECTION 6.4 QUALIFIED INSURER AND STANDARD PROVISIONS

Lessee covenants that each insurance policy required by this Article: (i)

shall be issued by such insurer or insurers as are financially responsible, are capable and qualified to write the respective insurance in the State of Florida and of recognized standing; and (ii) shall be in such form and with such provisions (including, without limitation, the loss payable clause, the waiver of subrogation or right of recovery claim clause, deductible clause and the designation of the named insured parties) as are generally considered standard provisions for the type of insurance involved and are reasonably acceptable to Lessor. Lessor agrees that in the event of destruction of all or any part of the Leased Premises, all insurance proceeds paid as a result of such casualty shall be made available to Lessee for payment for the repair, restoration or replacement of the portion of the Leased Premises damaged or destroyed. All insurance required under this Lease shall be written with insurance companies authorized to do business in the state of Florida and reasonably acceptable to Lessor. The cost of all premiums on the policies shall be paid by Lessee. form and substance of all policies shall be subject to the approval of Lessor. The policies shall contain a clause that the insurer will not cancel or change the insurance coverage without first giving Lessor thirty (30) days' prior written notice. The original policies or binders shall be delivered to Lessor within ten (10) days of the commencement of this Lease. Lessee shall also furnish Lessor with original or binders of renewal policies for each such policy at least ten (10) days prior to the expiration date of the expiring policy. All policies

- 35 -

(exclusive of worker's compensation) required under this Section shall name Lessor as an additional named insured.

### SECTION 6.5 LIABILITY PROTECTION

Lessee will provide for and insure in the commercial general liability policies required above, not only its own liability in respect of the matters there mentioned, but also the interest of Lessor as an additional insured party. Lessee will not, without the prior written consent of Lessor, settle or consent to the settlement of any pending or threatened litigation or claims for which Lessee is obligated under the provisions of this Section to insure the interest of Lessor.

## SECTION 6.6 INSURANCE PROCEEDS

All insurance proceeds paid as compensation for loss or damage to the Leased Premises as a result of any casualty shall be the property of Lessee to be used in accordance with Section 7.5 and, if not used by Lessee as provided in Section 7.5, shall be provided to Lessor.

ARTICLE VII

MAINTENANCE AND REPAIR

#### ALTERATIONS AND IMPROVEMENTS

#### SECTION 7.1 CONDITION OF HOSPITAL FACILITIES

\_\_\_\_\_

Lessee accepts the Hospital Facilities in the condition existing on the Commencement Date, "as is, where is, with all faults." Lessee shall keep and maintain the Hospital Facilities and tangible personal property used therein in good repair and operating condition, reasonable wear, tear, and depreciation excepted, and subject to

- 36 -

replacement in the normal course of business at Lessee's own expense during the Lease Term.

#### SECTION 7.2 REPAIRS

\_\_\_\_\_

Lessee shall make all repairs (including replacements) to the Hospital Facilities, and tangible personal property used therein, foreseen and unforeseen, structural or otherwise, whether or not caused by Lessee's act or omission, that may be necessary to keep and maintain the Hospital Facilities and tangible personal property used therein in good repair and operating condition and in compliance with all applicable rules, codes, laws, regulations, and ordinances of all governmental and quasi governmental agencies or boards having jurisdiction over the Hospital Facilities during the Lease Term.

### SECTION 7.3 ALTERATIONS AND IMPROVEMENTS

Lessee shall have the right, at its own expense, to make such additions, alterations, improvements in or to the Hospital Facilities as it deems necessary or desirable, provided that neither the value of the Hospital Facilities nor their utility for the purposes described in Section 4.1 is thereby materially impaired, and provided, further, that all permanent or structural additions, alterations, improvements, or replacements shall be deemed a part of the Hospital Facilities and shall become the property of Lessor without further cost to Lessor.

## SECTION 7.4 REMOVAL OR DISPOSITION OF PROPERTY

If Lessee is not in default under this Lease, and provided such change is not governed by the approval requirements set forth in Section 4.1(C) hereof, Lessee shall have the right from time to time to remove any or all non-structural improvements, furniture, equipment and fixtures in the Hospital Facilities, provided Lessee repairs any

damage to the Hospital Facilities caused by such removal, and provided Lessee restores at the time of termination of the Lease as provided herein that portion of the Hospital Facilities affected by such removal to the condition existing prior to the installation of the item so removed (and provides a substitution or replacement of any item so removed if such item constitutes a part of the Leased Premises necessary or desirable for continued use of the Hospital Facilities as full service general acute care hospitals).

If any property or structure, or any furnishings, machinery, equipment, or other improvement constituting a part of or used in the Leased Premises or the Operations shall become inadequate, obsolete, worn out, unsuitable, undesirable, or unnecessary and if its demolition, disposal or removal will not impair the structural soundness, efficiency, or the economic value of the Leased Premises, Lessee may demolish, remove, or dispose of such property and may, to the extent permitted by law, sell, trade in, exchange, replace or otherwise dispose of same, in whole or in part, as long as the Leased Premises and the Operations remain functional and operational as contemplated in Section 4.1 hereof.

#### SECTION 7.5 DAMAGED OR DESTROYED PROPERTY

If any part of the Hospital Facilities shall be destroyed or damaged by fire or other casualty Lessee shall promptly replace, repair, rebuild, and restore the property damaged or destroyed to substantially the same condition as existed prior to such damage or destruction, with such alterations and additions as will not impair the capacity or character of the applicable Hospital Facilities for the purposes stated in Section 4.1, applying so much as may be necessary of the net proceeds of insurance received by Lessee by

- 38 -

virtue of any such damage or destruction and paying from Lessee's own funds for the cost of such replacement, repair, rebuilding, or restoration, either on completion thereof or as the work progresses, if such insurance proceeds shall not be sufficient to pay for all such work.

#### SECTION 7.6 CONSTRUCTION LIENS

Lessor's interest in the Leased Premises shall not be subject to liens for improvements made by Lessee and Lessee shall have no power or authority to create any lien or permit any lien to attach to Lessee's leasehold or to the estate, reversion or other estate of Lessor in the Leased Premises or any improvements of which the Leased Premises are a part. All contractors, artisans, mechanics and laborers and other persons supplying design services, materials or labor or contracting with Lessee with respect to the Leased Premises or any part thereof, or any party entitled to claim a construction lien

under the laws of Florida (whether same shall proceed in law or in equity) are hereby charged with notice that they shall look solely to Lessee to secure payment of any amounts due for work done or material furnished to Lessee relating to the Leased Premises, or for any other purpose during the Lease Term.

Lessee covenants and agrees that Lessee shall indemnify the Indemnified Parties against all such claims and Lessee covenants and agrees to transfer any claimed or asserted lien to a bond or such other security as may be permitted by law within thirty (30) days of the assertion of any such lien or claim of lien. In the event Lessee fails to transfer such lien to bond or other security within such thirty (30) day period then, in addition to its other remedies specified in this Lease, Lessor shall have the right to discharge

- 39 -

the lien or to transfer the lien claimed to bond or other security permitted by law and in any such event Lessee shall pay all costs so incurred by Lessor immediately upon demand therefor.

#### ARTICLE VIII

#### ASSIGNMENTS, SUBLEASES AND ENCUMBRANCES

SECTION 8.1 ASSIGNMENTS AND TRANSFERS, BUSINESS COMBINATIONS, AND
RIGHT OF FIRST REFUSAL TO LESSOR

A. Assignment and Transfers. Lessee shall have the right to assign

its interest in this Lease or to sublease all or any portion of the Leased Premises or its real estate interest in the Leased Premises or Guarantor or its Affiliates shall have the right to sell or transfer fifty percent (50%) or more of the voting stock of Lessee or enter into any transaction changing the control of Lessee (whether in a single transaction or a series of transactions) only with the prior written consent of Lessor.

B. Competing Properties Acquired by Business Combination. Any sale,

purchase, merger, other reorganization, partnership, affiliation, or other transaction (in any case, a "Business Combination") pursuant to which Guarantor (or any of its Affiliates) or any corporation or other entity surviving or resulting from such Business Combination (or any of its Affiliates) will own or operate or have any interest in the ownership or operation of any other hospital in Hernando County, Florida, and/or Bayonet Point Hospital, in Pasco County, Florida (the "Other Hospitals") must be approved by the Lessor; provided, however, as to Bayonet Point Hospital, said approval shall not be unreasonably withheld; provided further, however, that the Lessor shall be deemed to have consented to such Business Combination if Guarantor (and all of its Affiliates) and

any other company surviving such Business Combination (and all of its Affiliates) have, within twelve (12) months after the date on which such Business Combination becomes effective, divested themselves of their entire right, title and interest in the Other Hospitals, whether held directly or indirectly, and any interest, whether direct or indirect, in the management or operation of the Other Hospitals.

- C. Transfer of Control of Guarantor.
- 1) In the event that, at any time during the Lease Term, Guarantor proposes (i) a transfer or sale (whether in a single transaction or series of transactions) of fifty percent (50%) or more of the voting stock of Guarantor, or substantially all of the assets of Guarantor, to a third party which was not an Affiliate of Guarantor immediately prior to such transfer or sale, or (ii) the merger or consolidation of Guarantor with or into a third party which was not an Affiliate of Guarantor immediately prior to such merger or consolidation (other than a merger or consolidation in which Guarantor is the surviving entity thereof) (each a "Disposition"), then the Lessor shall have the following options:
- (a) Approve the Disposition and thereby consent to the transaction by and between the third party and Lessee; or
  - (b) Disapprove the Disposition to the third party.

- 41 -

- 2) In the event the Lessor disapproves the Disposition as described in Section 8.1 C.1) (b), the Lessor shall elect one of the following options: (i) Lessor may purchase the Leased Premises, New Improvements and all tangible and intangible personal property used therein, other than the Excluded Assets, of Lessee that would be transferred to Lessor at the time of termination of the Lease (the "Termination Assets"), within one hundred and eighty (180) days from the date the Lessor disapproves the Disposition, upon payment of the fair market value for the Termination Assets as hereinafter described (the "Lease Asset Value"), or (ii) the Lessor will, within one hundred and eighty (180) days from the date the Lessor disapproves the Disposition, find a purchaser acceptable to Lessor that will purchase and close on the Termination Assets as provided for herein. If the Lessor does not exercise its rights under Section 8.1 C.1) (a) or Section 8.1 C.1) (b) within the time frame set forth herein, the Lessor shall be deemed to have consented to the Disposition and to the transaction by and between the Acquiror (as herein defined) and Guarantor.
- 3) In the event a Disposition transaction as described in Section 8.1 C.1) is proposed, Guarantor shall give written notice to Lessor ("Disposition Notice"), containing the name, address, qualifications, financial information

and background of the third party with whom the Disposition is proposed (the "Acquiror").

4) The Disposition Notice shall also set forth the Lease Asset Value for the Termination Assets. Lessor shall, within ninety (90) days after receipt of the Disposition Notice, either accept or reject the Disposition. In addition, if Lessor disagrees with Guarantor's determination of the Lease Asset Value, then Lessor and Guarantor

- 42 -

shall each select an MAI commercial real estate appraiser, with at least 10 years' experience in appraising hospital facilities, in the southeast region of the United States. Upon selection, Lessor's and Guarantor's appraisers will work together in good faith to agree upon the fair market value of the Termination Assets, taking into consideration the remaining term of the Lease, and the business operated therein. If after thirty (30) days of selection of the appraisers by Lessor and Guarantor a dispute continues, then in that event, both appraisers shall select a third appraiser meeting the criteria described above. Once the third appraiser has been selected, then, as soon thereafter as practicable but within thirty (30) days, the third appraiser shall make its determination as to which of the estimates received from the original appraisers most closely reflects the fair market value of the Termination Assets. The determination by the third appraiser shall be rendered in writing to both the Lessor and Guarantor and shall be final and binding upon them. The parties shall share equally in the cost of the third appraiser.

5) If the final determination of the Lease Asset Value is acceptable to Lessor, then, within thirty (30) days following the final determination of the Lease Asset Value, Guarantor and Lessor will enter into an agreement setting forth the terms and conditions of the sale of the Termination Assets for the Lease Asset Value. The parties' efforts in that regard will be pursued diligently and in good faith, and the agreement entered into by Lessor and Guarantor will contain only those terms and conditions that would be customary to consummate the transaction in accordance with the provisions hereof, including those with respect to any required governmental approvals, transfers of Licenses and Permits and contracts, and other

- 43 -

provisions necessary to ensure that Lessor receives in the Termination Asset those assets required to operate the Hospital Facilities as an acute care, inpatient hospital facility.

6) In evaluating the third party seeking to purchase, merge or consolidate with Lessee and/or Guarantor, the Lessor shall in each and every instance determine if the third party satisfies the conditions described in Section 8.1 A. hereof, and if it does, the Lessor shall approve the third party and the transaction so contemplated.

- 7) In the event that the Lessor does not agree with the final determination of the Lease Asset Value or fails to close under the purchase agreement described above, then in that event Guarantor may complete the Disposition transaction with the Acquiror. In such event, the Acquiror shall take the Termination Assets subject to all terms set forth in this Lease Agreement.
- 8) Lessee and Guarantor will afford to Lessor and its counsel, accountants and other representatives reasonable access to the books and records of Lessee and Guarantor relevant to the operation of the Hospital Facilities, and promptly will furnish Lessor with all information as Lessor may reasonably request in connection therewith.

SECTION 8.2 RESTRICTIONS ON MORTGAGE OR OTHER ENCUMBRANCES
OF LEASED PREMISES BY LESSEE

\_\_\_\_\_

Except as otherwise expressly provided in this Lease, Lessee shall have no power or authority to mortgage, pledge, sell, assign, transfer, convey, or make any other disposition or encumbrance of the Leased Premises or Lessee's leasehold interest in the Lease Agreement without the prior written consent of Lessor. It shall be an express condition of any such mortgage, sale, assignment, transfer, conveyance or other disposition or encumbrance, that it shall be

- 44 -

subject to all of the terms and conditions of this Lease, and that any transfer as a result of the exercise of any remedies in favor of a mortgagee or lien holder shall be subject to the provisions of Section 8.1, above, requiring Lessor's (additional) consent thereto.

ARTICLE IX

ADDITIONAL COVENANTS OF LESSEE

SECTION 9.1 PROTECTION OF THE LEASED PREMISES

Lessee shall use reasonable care in performing its duties to protect the Leased Premises. Lessee covenants that it will not knowingly permit anything to be done on or about the Leased Premises that will adversely affect, impair, or contravene any policies of insurance that may be carried on the Leased Premises or any part thereof against loss or damage by fire, casualty, or otherwise.

ARTICLE X

EVENTS OF DEFAULT AND REMEDIES

SECTION 10.1 EVENTS OF DEFAULT DEFINED

Each of the following shall be an event of default under this Lease and the terms "event of default" or "default" shall mean, whenever they are used in this Lease, any one or more of the following events:

- 45 -

- A. Failure by Lessee to observe and perform any covenant, condition, or agreement on its part to be observed or performed which can be cured by the payment of money, for a period of thirty (30) days after written notice is given by Lessor to Lessee, specifying such failure; or
- B. Failure by Lessee to observe and perform any covenant, condition, or agreement on its part to be observed or performed other than the payment of money, for a period of sixty (60) days after written notice is given by Lessor to Lessee, specifying such failure; provided, however, if the failure stated in the notice cannot be corrected within such sixty (60) day period, Lessee shall not be deemed to be in default if corrective action is instituted by Lessee within such sixty (60) day period and is diligently pursued until the corrective action is completed; or
- C. The dissolution or liquidation of Lessee or the filing by Lessee of a voluntary petition in bankruptcy, or the filing of an involuntary petition in bankruptcy against Lessee (unless such petition is dismissed within one hundred twenty (120) days after it was filed); failure by Lessee promptly to lift any execution, garnishment, or attachment of such consequence as will impair its ability to carry on its operation of the Hospital Facilities; Lessee's seeking of or consenting to or acquiescing in the appointment of a receiver of the Leased Premises; the commission by Lessee of an act of bankruptcy, or adjudication of Lessee as bankrupt; or any assignment by Lessee for the benefit of its creditors, or the entry by Lessee into an agreement of composition with its creditors. The term "dissolution or liquidation of Lessee" as used in this subsection, shall not be construed to include (i) the cessation of

- 46 -

the corporate existence of Lessee resulting either from a merger or consolidation of Lessee into or with another corporation, (ii) an administrative dissolution of Lessee by the Florida Department of State, because of the failure of Lessee to file its annual report with the Department of State provided that Lessee is reinstated as a corporation within ninety (90) days after such administrative dissolution, or (iii) a transfer of all or substantially all of Lessee's assets as permitted under this Lease.

### SECTION 10.2 REMEDIES ON DEFAULT

If any event of default described in Section 10.1 shall occur, Lessor, without limiting its other remedies under the law or pursuant to the provisions of this Lease, may elect to terminate this Lease, and exclude Lessee from

possession of the Leased Premises after giving Lessee written notice of its election to cancel and terminate this Lease and in any such event Lessee shall execute and deliver to Lessor, if necessary, all documents and items required pursuant to the provisions of Section 3.2. No termination shall be implied or effective unless Lessor elects in writing to terminate this Lease. In addition, the Lessor shall have the right to seek payment of all amounts due hereunder to be paid by Lessee or loaned to Lessee by Lessor pursuant to Section 10.3 hereof.

SECTION 10.3 ADVANCES BY LESSOR

In addition to the foregoing remedy, if Lessee fails to pay the premiums on policies to provide the full insurance coverage required by this Lease, fails to pay any Taxes at or prior to the time they are required to be paid, or fails to keep the Hospital Facilities in good order and repair or fails to perform any other covenants or agreements on Lessee's part to be performed hereunder, Lessor, after

- 47 -

first giving Lessee reasonable prior notice of any such failure on its part, may (but shall not be obligated to) pay the premiums on such insurance, pay such taxes or other charges, or make such repairs, renewals, and replacements as may be necessary to maintain the Hospital Facilities properly and in good order and repair and in a reasonably safe condition, or perform such acts which Lessee has failed to perform and all amounts advanced or paid therefor by Lessor shall become an additional obligation of Lessee to Lessor. Lessee hereby agrees to pay the same upon demand to Lessor together with interest thereon at the maximum rate then permitted to be charged by private parties in contracts governed by Florida law.

SECTION 10.4 AGREEMENT TO PAY ATTORNEYS' FEES AND EXPENSES

In the event Lessor or Lessee is required to employ an attorney or incur other expenses for the enforcement or interpretation of this Lease, the non-prevailing party will, on demand therefor, pay to the other party its attorneys' fees (including fees for paralegals and legal assistants) and costs and expenses incurred out of court, in the trial court, on appeal or in bankruptcy, mediation, arbitration or any administrative proceedings.

SECTION 10.5 WAIVER

Any of the terms or conditions of this Lease may be waived at any time and from time to time, in writing, by both parties; provided, however, that except as otherwise specifically provided in this Lease, no failure or delay on the part of either party in exercising any of its respective rights hereunder upon any failure by the other party to perform or observe any condition, covenant or provision herein contained shall operate as a waiver thereof, nor shall any

further exercise thereof or the exercise of any other right hereunder. No waiver or release of any of the terms, conditions, or provisions of this Lease shall be valid or asserted or relied upon by either party hereto or offered in any judicial proceeding or otherwise, unless the same is in writing, and duly executed by the party granting such waiver or release.

#### ARTICLE XI

#### ENVIRONMENTAL PROVISIONS

SECTION 11.1 COMPLIANCE

Lessee covenants and warrants that Lessee's Operations on the Leased Premises will at all times substantially comply with and conform to all Environmental Laws, including without limitation, those Environmental Laws which relate to the Handling of any Waste on or about the Leased Premises. Upon receipt of any material Environmental Notice whatsoever, Lessee shall deliver to Lessor a true, correct and complete copy of any written Environmental Notice. In the event (i) of any Environmental Notice; or (ii) if Lessee has caused, suffered or permitted, directly or indirectly, any Spill or Environmental Condition on or about the Leased Premises, or (iii) if any Spill or Environmental Condition has occurred on or about the Leased Premises or otherwise affecting the Leased Premises, then Lessee shall immediately take all of the following actions:

A. Notify Lessor;

- 49 -

- B. Promptly commence and diligently pursue all steps necessary to clean up any such Spill and any contamination related to the Spill or to remediate or abate such Environmental Condition and Environmental Notice;
- C. Promptly provide Lessor with copies of all reports, data, proposals, test results or analyses, assessment or remediation plans relating to such incidents;
- D. Abate the Environmental Condition and the Environmental Notice and otherwise restore the Leased Premises or affected property to the condition required by all applicable laws and regulations; provided, however, if any laws or regulations hereafter enacted shall prescribe more stringent standards than those in effect at the time any remediation is effected, Lessee shall perform any work required to bring the Leased Premises into compliance with such stricter standards at Lessee's sole expense; and

E. Fully cooperate with Lessor with respect to any such incident, including permitting Lessor to monitor and inspect all activities; provided, however, Lessor shall not interfere with any remediation work that may be in progress.

## SECTION 11.2 COMPLETION OF REMEDIAL WORK

In the event Lessee shall fail to commence the remedial work in timely fashion or fail to prosecute the remedial work to completion within a reasonable time after commencement of such work, Lessor may, but shall not be required to, cause the remedial work to be performed, subject fully to the Environmental Indemnification provisions of this Lease, and Lessee shall pay for all such work as the costs are incurred.

- 50 -

### SECTION 11.3 CLAIMS

Lessee shall immediately notify Lessor of any material claims for damages, penalties or otherwise against Lessor or Lessee described above.

## SECTION 11.4 ASBESTOS

Without representation or warranty by Lessor as to its content or to the existence or non-existence of asbestos in the Leased Premises, attached hereto as Exhibit "C" is a description of certain asbestos that is present in the Leased Premises.

ARTICLE XII

CONDEMNATION

#### SECTION 12.1 TAKING

If any competent authority, for any public or quasi-public purposes, takes or condemns, or if Lessor makes a conveyance in lieu of a taking or condemnation of, either (i) the whole of the Leased Premises, or (ii) such part of the Leased Premises as hereafter defined so as to prevent or substantially impair the use of the Leased Premises for the purposes stated in Section 4, then the Lease Term shall automatically cease and terminate on the date possession of the Leased Premises is required by, or surrendered to, the condemning authority.

## SECTION 12.2 LAST YEAR OF LEASE TERM

If any competent authority, for any public or quasi- public purpose, takes

or condemns, or if Lessor makes a conveyance in lieu of a taking or condemnation of, any part of the Leased Premises in the last year of the Lease Term such that the use of the premises for the purposes stated in Section 4 is prohibited or substantially impaired, either party shall have the right to terminate this Lease

- 51 -

by written notice to the other party tendered no later than thirty (30) days following the date possession is required by the condemning authority. If either party elects so to terminate, the Lease Term shall cease and terminate as of the date when possession of the area so taken is required by, or surrendered to, the condemning authority.

## SECTION 12.3 COLLECTION OF AWARDS

In the event of a complete or partial taking of the Leased Premises, the parties agree that Lessor shall be entitled to collect the portion of the award paid by the condemning authority attributable to the land, permanent improvements and all estates or interests therein that is equal to the expired portion of the Lease Term expressed as a percentage and Lessee shall collect the balance of the award. Lessee, however, may also claim business relocation damages described in Section 12.5 herein.

## SECTION 12.4 NON-TERMINATION OF LEASE

In the event of a partial taking of the Leased Premises which does not result in termination of this Lease then, Lessee shall, at Lessee's expense, promptly repair, restore and reconstruct the Leased Premises and Lessor shall make available to Lessee such portion of the condemnation awards as may be required to pay for such repairs and restoration. The restoration shall be made as nearly as possible to the condition existing immediately prior to such taking of the Leased Premises and improvements thereon.

## SECTION 12.5 BUSINESS LOCATION EXPENSES

Nothing in this Section shall be deemed to prohibit Lessee from claiming compensation from the condemning authority (but not from Lessor) for business relocation expenses in connection with such taking.

- 52 -

ARTICLE XIII

ASSUMPTION OF RISK

SECTION 13.1 USE AND OCCUPATION

\_\_\_\_\_\_

To the maximum extent permitted by law, Lessee agrees to use and occupy the Leased Premises at its own risk, and hereby (for itself and all persons claiming under, by or through Lessee) releases Lessor and its servants, contractors, and employees, from all claims and demands of every kind resulting from any accident, damage, injury, or death occurring thereon, unless solely due to such party's gross negligence or willful misconduct. Lessee expressly covenants and agrees that Lessor shall not be liable to Lessee or Lessee's contractors, licensees, agents, guests, invitees, or employees for any injury, damage, or loss to its, his or their persons or property by any cause whatsoever, including, without limitation, construction defects, water, rain, sleet, fire, storms, negligence and accidents, breakage, stoppage, or leaks of gas, water, heating, sewer pipes, boilers, wiring or plumbing or any other defect in, on or about the Leased Premises.

SECTION 13.2 LIABILITY

Lessee expressly assumes all liability for or on account of all claims, suits, liability and expense arising from claims for injury, loss or damage, on or about or relating to the Leased Premises except those caused by the acts of Lessor or Lessor's agents, contractors or representatives.

- 53 -

#### ARTICLE XIV

INDEMNITY

### SECTION 14.1 GENERAL INDEMNITY

Lessee agrees to indemnify Lessor and to defend and hold the Indemnified Parties harmless from and against all liability and damage caused by or growing out of any injury, death, loss or damage to persons or property upon the Leased Premises or arising from Lessee's Operations upon the Leased Premises during the Lease Term (except to the extent caused by Lessor's acts), including, without limitation, the obligation to pay all attorneys' fees and costs, and consultants' fees incurred by Lessor as a result thereof. Lessee and Lessor acknowledge that the Indemnity obligation of Lessee as to notice, settlement and attorneys' fees are as set forth in Section 1.2(N) herein. Lessee shall also indemnify, defend and hold the Indemnified Parties harmless from and against any losses, liabilities, damages, interests, fines, penalties, expenses, and attorneys' fees and costs based upon, arising out of or otherwise in respect to, noncompliance with any governmental law, statute, regulation, ordinance, or administrative or judicial judgment or order relating to Lessee's Operations on the Leased Premises except as to those matters of non-compliance that existed as of the Commencement Date for the Hospital Facilities and the operation thereof.

As a material inducement to Lessor to enter into this Lease, Lessee shall hold the Indemnified Parties harmless from all judgments or claims, and shall assume the burden and expense of defending all suits, administrative proceedings, and negotiations of any description with any and all persons, political subdivisions or government

- 54 -

agencies arising out of any of the occurrences of any Spill or Environmental Conditions on the Leased Premises which may occur as a result of Lessee's Operations on the Leased Premises.

By its execution of this Lease, Lessee acknowledges the receipt and sufficiency of Fifty Dollars (\$50.00) paid by Lessor as separate consideration for all indemnities provided and given by Lessee in this Lease, which sum also is separate (and sufficient) consideration for all other indemnities provided by Lessee to Lessor under this Lease, including, without limitation, Lessee's indemnification of Lessor described in other Sections of this Lease.

ARTICLE XV

#### MISCELLANEOUS

## SECTION 15.1 COVENANT OF QUIET ENJOYMENT

So long as Lessee performs and observes all of the covenants and agreements on its part herein contained, it shall peaceably and quietly have, hold, and enjoy the Leased Premises during the Lease Term subject to all terms and provisions hereof, against the lawful claims of all persons or entities claiming by, through or under Lessor.

### SECTION 15.2 COUNTERPARTS

This Lease may be executed in counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument.

- 55 -

## SECTION 15.3 BINDING EFFECT

This Lease shall inure to the benefit of, and shall be binding upon, Lessor, Lessee, Guarantor and their respective authorized successors and assigns as provided for herein.

In the event any provision of this Lease shall be held invalid or unenforceable by any court of competent jurisdiction, such holding shall not invalidate or render unenforceable any other provision hereof.

### SECTION 15.5 ARTICLE AND SECTION CAPTIONS

The Article and Section headings and captions contained herein are included for convenience only and shall not be considered a part hereof or affect in any manner the construction or interpretation hereof.

## SECTION 15.6 ENTIRE AGREEMENT

This Lease may not be modified, amended or otherwise changed orally, but may only be modified, amended or otherwise changed by an agreement in writing signed by both parties. This Lease Agreement and its accompanying guaranty constitute the entire agreement between the parties and there are no oral or other agreements between the parties affecting this Lease. This Lease Agreement supersedes and cancels any and all previous negotiations, arrangements, agreements, and understandings between the parties hereto with respect to the subject matter thereof, and no such outside or prior agreements shall be used to interpret or to construe this Lease. There are no promises, covenants, representations or inducements in addition to, or at variance with any of the terms of this Lease Agreement except the Guaranty.

- 56 -

## SECTION 15.7 GUARANTY AGREEMENT

That certain guaranty of even date signed by Health Management Associates, Inc., a true and complete copy of which is attached hereto as Exhibit "D", is given by Guarantor to induce Lessor to execute this Lease.

## SECTION 15.8 PLACE OF DELIVERY OF RENT AND NOTICES

Rent accruing hereunder, as well as all notices, shall be paid or delivered to Lessor at the address indicated below until Lessee is notified otherwise.

All notices given to Lessee hereunder shall be delivered to Lessee at the address indicated below until Lessor is notified otherwise.

All notices, to be effective, shall be in writing and shall be delivered by registered or certified mail, return receipt requested and postage prepaid or by personal delivery or courier service by which the sending party obtains a

receipt confirming delivery or inability to deliver the notice to the party to whom the notice was addressed. Any notice required to be given or that may be given under this Lease shall be deemed to be given upon the date of receipt thereof, or if delivery is refused, on the date of first attempted delivery thereof.

Any party designated to receive notice may change its address to any other place in the United States of America by giving notice of such change of address to the other party.

- 57 -

If to Lessor: Hernando County Board of

County Commissioners County Courthouse

20 North Main Street, Room 460 Brooksville, Florida 34601

Attention: Chairman

with copy to: R. Bruce Snow, Esquire

112 North Orange Avenue Brooksville, Florida 34601

If to Lessee: Hernando HMA, Inc.

5811 Pelican Bay Boulevard, Suite 500

Naples, Florida 33963-2710

Attention: William T. Schoen, Chairman and Chief Executive Officer

with copy to: 5811 Pelican Bay Boulevard, Suite 500

Naples, Florida 33963-2710

Attention: Timothy Parry, Vice President

and General Counsel

Lessor and Lessee may, by notice given hereunder, designate any further or different addresses to which subsequent notices or other communications shall be sent.

SECTION 15.9 RADON GAS

The following disclosure is hereby made by Lessor to Lessee as required by Florida law:

Radon is a naturally occurring radioactive gas that, when it has accumulated in a building in sufficient quantities, may present health risks to persons who are exposed to it over time. Levels of radon that exceed Federal and State guidelines have been found in buildings in Florida. Additional information regarding radon and radon testing may be obtained from your County Public Health Unit.

\_\_\_\_\_\_

Lessor shall have the right to enter the Leased Premises at reasonable business hours for the purposes of inspecting the Leased

- 58 -

Premises, and Lessor shall also have the right to make access available at all reasonable business hours to agents and consultants (or to prospective tenants of the Leased Premises during the last twelve (12) months of the Lease Term). If Lessee fails to make necessary repairs to the Leased Premises within applicable grace periods after notice of the need therefor, Lessor shall also have the right to enter the Leased Premises for the purpose of making necessary repairs thereto. In addition, during the last year of the Lease Term the Lessee shall and does hereby authorize representatives of the Lessor access to the Leased Premises acting in the capacity of a transition team for the purpose of familiarizing themselves with the operation, organization and management of the Hospital Facilities and all furniture, fixtures and equipment utilized in the operation of the Hospital Facilities. The activities of the transition team shall be conducted in such a way so as to not interfere with the operation of the Hospital Facilities.

### SECTION 15.11 SURRENDER

SECTION

All alterations, additions and improvements made during the Lease Term on the Leased Premises and all additions and/or replacement of tangible personal property used in the Operations shall remain upon the Leased Premises in the absence of any agreement to the contrary, and Lessor will accept the Leased Premises and such tangible personal property in such condition as may be subsequently improved or, as to personal property, replaced by Lessee after the commencement of Lessee's Operations on the Leased Premises. Except as expressly provided in this Lease to the contrary, Lessee shall deliver and surrender to Lessor possession of the Leased Premises and all tangible and intangible personal property other than the Excluded Assets at

- 59 -

the expiration of this Lease, or its earlier termination as herein provided, broom clean, and in as reasonably good condition and repair as the same shall be on the date of commencement of Lessee's Operations (or, if applicable, in such better condition and repair as the Leased Premises may have been put by Lessee during the continuance of Lessee's Operations), ordinary wear and tear excepted, together with all tangible personal property and replacement thereof, including furniture, fixtures, equipment and inventory located thereon. Any and all tangible personal property that becomes obsolete during the Lease Term may be discarded without replacement provided that such discarded tangible personal property is replaced with such other tangible personal property as is necessary

for the continuation of state of the art services being provided at the Hospital Facilities.

Notwithstanding anything herein to the contrary, at the expiration or termination of this Lease, the Leased Premises shall be returned to Lessor in conformity and compliance with all applicable laws, regulations and ordinances; provided, however, if any laws, regulations or ordinances hereafter enacted shall prescribe more stringent standards for remediation that those in effect at the time remediation of an Environmental Condition was effected, Lessee shall perform all work required to bring the Leased Premises into compliance with such stricter standards at Lessee's sole cost and expense. If such additional work is required after the expiration or termination of this Lease, Lessee shall commence the work within thirty (30) days of Lessor's notice of the need therefor, and shall complete the work with reasonable diligence thereafter.

- 60 -

#### SECTION 15.12 RELATIONSHIP OF PARTIES

Nothing contained herein shall be deemed or construed by the parties hereto, nor by any third party, as creating the relationship of principal and agent or of partnership or of joint venture between the parties hereto, it being understood and agreed that neither the method of computation of Rent, nor any other provision contained herein, nor any acts of the parties herein, shall be deemed to create any relationship between the parties hereto other than the relationship of landlord and tenant. Whenever the singular number is used in this Lease, the same shall include the plural, and the neuter gender shall include the feminine and masculine genders.

## SECTION 15.13 BROKER'S COMMISSIONERS

Each of the parties represents and warrants that it has dealt with no broker in the negotiation and consummation of this Lease. Each of the parties agrees to indemnify the other against all liabilities from any claims for brokers' commissions or finder's fees asserted against the indemnified party.

## SECTION 15.14 INTERPRETATION AND VENUE

This Lease shall be construed and interpreted according to the laws of the State of Florida and venue for enforcement shall be in Hernando County, Florida.

# SECTION 15.15 TIME OF THE ESSENCE

Time is of the essence in the performance of Lessee's obligations under this Lease.

### SECTION 15.16 SOVEREIGN IMMUNITY

The parties hereto acknowledge that by virtue of Lessor signing this Lease Agreement which contains specific indemnity rights from Lessee that Lessor has not waived its right of sovereign immunity as to any tort claim that may or may not exist against Lessor by virtue of any act by or failure to act of Lessor.

## SECTION 15.17 CERTIFICATION

On each and every Anniversary Date of the Lease, the Lessee shall deliver to Lessor a Certificate ratifying to Lessor the compliance (the "Certificate of Compliance") by Lessee of all of the material covenants, representations and warranties of Lessee as set forth in this Lease. The Certification of Compliance shall be delivered on or before thirty (30) days after the expiration of each anniversary date of the Lease.

The Lessee shall also provide to the Lessor, ninety (90) days after the close of each fiscal year of the Lessee, a report setting forth the status of all material covenants, representations, terms, conditions, and warranties under this Lease, which report shall contain such information as may be reasonable and necessary in order to enable Lessor to confirm Lessee's compliance with all of the material terms, conditions, covenants, representations, and warranties set forth in this Lease, including but not limited to indigent care, equipment inventory, financial reports of the Hospital Facilities, insurance coverage, and compliance with licensure and accreditation provisions.

- 62 -

#### SECTION 15.18 INDUCEMENT

The parties acknowledge and agree that Lessor would not enter into this Lease if the Rent described in this Lease were not absolutely net to Lessor or if full service acute hospital care will not be available in Hernando County or if Lessor were to incur any liability whatsoever, foreseen or unforeseen, as a result of Lessee's exercise of any of its rights under this Lease. Accordingly, anything herein to the contrary notwithstanding, Lessee expressly covenants and agrees to pay all expenses, costs, fees and charges of any nature whatsoever arising in connection with or attributable to the Leased Premises or this Lease, and, to assure Lessee's performance, Lessee's parent, Health Management Associates, has delivered its guaranty of payment and performance of Lessee's obligations simultaneously with Lessee's execution of this Lease.

SECTION 15.19 MEMORANDUM OF LEASE

The	parties	hereto	agre	e that	а	Memo	orandum	of	Lea	se	in	the	form	atta	ached
hereto a	s Exhibit	t "E" s	hall k	oe pla	ced	of	record	amo	ng	the	рu	ablic	reco	ords	of
Hernando	County,	Florid	a.												

SECTION 15.20 FURTHER ASSURANCE

The parties hereto shall perform with all due diligence all acts, applications, authorizations, assignments, transfers, and consents as necessary or appropriate to the fulfillment of the provisions of this Lease Agreement, and will cooperate with each other and execute any and all documents required under this Lease Agreement or that are incident thereto.

- 63 -

IN WITNESS WHEREOF, Lessor and Lessee have caused this Lease to be duly executed as of the day and year first above written.

HERNANDO COUNTY, FLORIDA, a political subdivision of the State of Florida

By:/s/ Hannah M. Robinson
----Name: Hannah M. Robinson

Chairman of the Board of County Commissioners

HERNANDO HMA, INC., a Florida corporation

By:/s/ Earl P. Holland

Name: Earl P. Holland

Title: Vice Chairman

STATE OF FLORIDA
COUNTY OF \_\_\_\_\_

The foregoing instrument was acknowledged before me , 1997, by	this day of
and	as Chairman and Clerk,
respectively of the Board of County Commissioners of HER	NANDO COUNTY, FLORIDA, a
political subdivision of the State of Florida, on behalf	of the

	me or has produced (state) driver's license nidentification.	10.
My Commission Expires:	Notary Public (Signature)	
(AFFIX NOTARY SEAL)	(Printed Name)	
	(Title or Rank)	
	(Serial Number, if any)	

- 64 -

#### <ARTICLE> 5

#### <LEGEND>

THIS SCHEDULE CONTAINS SUMMARY FINANCIAL INFORMATION EXTRACTED FROM THE FINANCIAL STATEMENTS WITHIN THE COMPANY'S FORM 10-Q AND IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO SUCH FINANCIAL STATEMENTS.

</LEGEND>

<\$>	<c></c>	
<period-type></period-type>	9-MOS	
<fiscal-year-end></fiscal-year-end>		SEP-30-1998
<period-start></period-start>		OCT-01-1997
<period-end></period-end>		JUN-30-1998
<cash></cash>		15675000
<securities></securities>		0
<receivables></receivables>		296728000
<allowances></allowances>		82726000
NVENTORY		22407000
<current-assets></current-assets>		277868000
<pp&e></pp&e>		901062000
<pre><depreciation></depreciation></pre>		173678000
<total-assets></total-assets>		1071170000
<current-liabilities></current-liabilities>		132207000
<bonds></bonds>		150724000
<preferred-mandatory></preferred-mandatory>		0
<preferred></preferred>		0
<common></common>		2510000
<other-se></other-se>		749312000
<total-liability-and-equity></total-liability-and-equity>		1071170000
<sales></sales>		0
<total-revenues></total-revenues>		839642000
<cgs></cgs>		0
<total-costs></total-costs>		558955000
<other-expenses></other-expenses>		35816000
<loss-provision></loss-provision>		70380000
<interest-expense></interest-expense>		3315000
<income-pretax></income-pretax>		171176000
<income-tax></income-tax>		67186000
<pre><income-continuing></income-continuing></pre>		103990000
<pre><discontinued></discontinued></pre>		0
<extraordinary></extraordinary>		0
<changes></changes>		0
<net-income></net-income>		103990000
<eps-primary></eps-primary>		.42
<eps-diluted></eps-diluted>		.41

</TABLE>